UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

FORM 10-Q

QUARTERLY REPORT PURSUANT TO SECTION 13 OR 15 (d) OF THE SECURITIES EXCHANGE ACT OF 1934

For the Quarter Ended March 31, 2003

Commission File No. 1-10403

TEPPCO Partners, L.P.

(Exact name of Registrant as specified in its charter)

Delaware (State of Incorporation or Organization) 76-0291058 (I.R.S. Employer Identification Number)

2929 Allen Parkway P.O. Box 2521 Houston, Texas 77252-2521 (Address of principal executive offices, including zip code)

(713) 759-3636 (Registrant's telephone number, including area code)

Indicate by check mark whether the registrant (1) has filed all reports required to be filed by Section 13 or 15 (d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days. Yes [X] No []

Indicate by check mark whether the registrant is an accelerated filer (as defined in Rule 12b-2 of the Exchange Act). Yes [X] No []

Indicate the number of shares outstanding of each of the issuer's classes of common stock, as of the latest practicable date.

Limited Partner Units outstanding as of April 30, 2003: 57,751,447

TABLE OF CONTENTS

PART I. FINANCIAL INFORMATION Item 1. Financial Statements **CONSOLIDATED BALANCE SHEETS CONSOLIDATED STATEMENTS OF INCOME** CONSOLIDATED STATEMENTS OF CASH FLOWS NOTES TO CONSOLIDATED FINANCIAL STATEMENTS Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations Forward-Looking Statements Item 3. Quantitative and Qualitative Disclosures About Market Risk Item 4. Controls and Procedures PART II. OTHER INFORMATION Item 6. Exhibits and Reports on Form 8-K. **SIGNATURES** Joint Development Agreement Computation of Ratio of Earnings to Fixed Charges Certification of CEO Pursuant to Section 906 Certification of CFO Pursuant to Section 906

TABLE OF CONTENTS

	Page
PART I. FINANCIAL INFORMATION	
Item 1. Financial Statements	
Consolidated Balance Sheets as of March 31, 2003 (unaudited) and December 31, 2002	1
Consolidated Statements of Income for the three months ended March 31, 2003 and 2002 (unaudited)	2
Consolidated Statements of Cash Flows for the three months ended March 31, 2003 and 2002 (unaudited)	3
Notes to the Consolidated Financial Statements (unaudited)	4
Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations	24
Forward-Looking Statements	37
Item 3. Quantitative and Qualitative Disclosures About Market Risk	38
Item 4. Controls and Procedures	39
PART II. OTHER INFORMATION	
Item 6. Exhibits and Reports on Form 8-K	40

PART I. FINANCIAL INFORMATION

Item 1. Financial Statements

TEPPCO PARTNERS, L.P.

CONSOLIDATED BALANCE SHEETS (in thousands)

	March 31, 2003	December 31, 2002
	(Unaudited)	
ASSETS	()	
Current assets:		
Cash and cash equivalents	\$ 13,939	\$ 30,968
Accounts receivable, trade	363,532	276,450
Accounts receivable, related party	3,459	4,313
Inventories	23,562	17,166
Other	25,854	31,670
Total current assets	430,346	360,567
Property, plant and equipment, at cost (net of accumulated depreciation and amortization of		
\$335,441 and \$338,746)	1,511,905	1,587,824
Equity investments	377,809	284,705
intangible assets	453,897	465,374
Goodwill	16,944	16,944
Dther assets	54,544	55,228
Total assets	\$2,845,445	\$2,770,642
Total assets	\$2,043,445	\$2,770,042
LIABILITIES AND PARTNERS' CAPITAL		
Current liabilities:		
Accounts payable and accrued liabilities	\$ 348,719	\$ 261,080
Accounts payable, related parties	10,085	6,619
Accrued interest	16,402	29,726
Other accrued taxes	8,930	11,260
Other	50,168	58,098
Total current liabilities	434,304	366,783
Senior Notes	1,142,163	945,692
Other long-term debt	265,000	432,000
Other liabilities and deferred credits	15,900	30,962
Redeemable Class B Units held by related party	102,693	103,363
Commitments and contingencies	- ,	,
Partners' capital:		
Accumulated other comprehensive loss	(14,586)	(20,055)
General partner's interest	10,032	12,770
Limited partners' interests	889,939	899,127
Emilia paraleto militoto		055,127
Total partners' capital	885,385	891,842
rotai partifers capitai		031,042
Total liskilizion and posterow? analtal	¢> 04E 44E	¢0.770.040
Total liabilities and partners' capital	\$2,845,445	\$2,770,642

See accompanying Notes to Consolidated Financial Statements.

CONSOLIDATED STATEMENTS OF INCOME (Unaudited) (in thousands, except per Unit amounts)

		Three Months Ended March 31,	
	2003	2002	
Operating revenues:			
Sales of petroleum products	\$ 975,960	\$545,208	
Transportation – Refined products	26,894	25,144	
Transportation – LPGs	30,821	23,360	
Transportation – Crude oil	6,905	6,128	
Transportation – NGLs	9,899	6,306	
Gathering – Natural gas	34,282	9,520	
Mont Belvieu operations		4,506	
Other	14,478	10,965	
outer		10,000	
Total operating revenues	1,099,239	631,137	
Total operating revenues	1,039,239		
Costs and expenses:			
Purchases of petroleum products	962,844	532,971	
Operating, general and administrative	42,635	31,445	
Operating fuel and power	10,177	8,589	
Depreciation and amortization	27,313	16,041	
Taxes – other than income taxes	4,928	4,505	
Total costs and expenses	1,047,897	593,551	
Operating income	51,342	37,586	
Interest expense	(21,905)	(16,787)	
Interest capitalized	596	2,109	
Equity earnings	3,710	3,572	
Other income – net	182	328	
oucl monte net			
Net income	\$ 33,925	\$ 26,808	
Net Income Allocation:			
Limited Partner Unitholders	\$ 23,087	\$ 18,594	
Class B Unitholder	1,680	1,793	
General Partner	9,158	6,421	
Total net income allocated	\$ 33,925	\$ 26,808	
Basic and diluted net income per Limited			
Partner and Class B Unit	\$ 0.43	\$ 0.46	
Weighted average Limited Partner and Class B			
Units outstanding	57,728	44,559	
Ŭ			

See accompanying Notes to Consolidated Financial Statements.

CONSOLIDATED STATEMENTS OF CASH FLOWS (Unaudited) (in thousands)

	Three Months Ended March 31,	
	2003	2002
Cash flows from operating activities:		
Net income	\$ 33,925	\$ 26,808
Adjustments to reconcile net income to cash provided by operating activities:		
Depreciation and amortization	27,313	16,041
Earnings in equity investments, net of distributions	(3,779)	3,402
Non-cash portion of interest expense	1,031	1,587
Decrease (increase) in accounts receivable	(87,082)	8,170
Increase in inventories	(6,396)	(4,667)
Decrease (increase) in other current assets	5,816	(4,955)
Increase (decrease) in accounts payable and accrued expenses	63,250	(13,327)
Other	1,576	(9,458)
Net cash provided by operating activities	35,654	23,601
Cash flows from investing activities:		
Purchase of Chaparral NGL System	_	(132,000)
Purchase of Jonah Gas Gathering Company		(7,315)
Acquisition of additional interest in Centennial Pipeline LLC	(20,000)	(7,515)
Investment in Centennial Pipeline LLC	(1,000)	(3,334)
Capital expenditures	(15,434)	(33,001)
Net cash used in investing activities	(36,434)	(175,650)
Cash flows from financing activities:	10.000	
Proceeds from term and revolving credit facilities	40,000	172,000
Repayments on term and revolving credit facilities	(207,000)	(540,658)
Issuance of Senior Notes	198,570	497,805
Debt issuance costs	(1,300)	(4,126)
Issuance of Limited Partner Units, net	12	56,839
General Partner's contributions	2	1,172
Distributions paid	(46,533)	(33,545)
Net cash provided by (used in) financing activities	(16,249)	149,487
Net decrease in cash and cash equivalents	(17,029)	(2,562)
Cash and cash equivalents at beginning of period	30,968	25,479
cash and cash equivalents at beginning of period		
Cash and cash equivalents at end of period	\$ 13,939	\$ 22,917
Non-cash investing activities:		
Net assets transferred to Mont Belvieu partnership	\$ 69,459	\$ —
····· r	,	
Supplemental disclosure of cash flows:		
Interest paid during the period (net of capitalized interest)	\$ 32,491	\$ 16,799

See accompanying Notes to Consolidated Financial Statements.

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS (Unaudited)

NOTE 1. ORGANIZATION AND BASIS OF PRESENTATION

TEPPCO Partners, L.P. (the "Partnership"), a Delaware limited partnership, is a master limited partnership formed in March 1990. We operate through TE Products Pipeline Company, Limited Partnership ("TE Products"), TCTM, L.P. ("TCTM") and TEPPCO Midstream Companies, L.P. ("TEPPCO Midstream"). Collectively, TE Products, TCTM and TEPPCO Midstream are referred to as the "Operating Partnerships." Texas Eastern Products Pipeline Company, LLC (the "Company" or "General Partner"), a Delaware limited liability company, serves as our general partner and owns a 2% general partner interest in us. The General Partner is a wholly owned subsidiary of Duke Energy Field Services, LLC ("DEFS"), a joint venture between Duke Energy Corporation ("Duke Energy") and ConocoPhillips. Duke Energy holds an approximate 70% interest in DEFS, and ConocoPhillips holds the remaining 30%. The Company, as general partner, performs all management and operating functions required for us, except for the management and operations of certain of the TEPPCO Midstream assets that are managed by DEFS on our behalf. We reimburse the General Partner for all reasonable direct and indirect expenses incurred in managing us.

As used in this Report, "we," "us," "our," and the "Partnership" means TEPPCO Partners, L.P. and, where the context requires, includes our subsidiaries.

The accompanying unaudited consolidated financial statements reflect all adjustments that are, in the opinion of the management of the Company, of a normal and recurring nature and necessary for a fair statement of our financial position as of March 31, 2003, and the results of our operations and cash flows for the periods presented. The results of operations for the three months ended March 31, 2003, are not necessarily indicative of results of our operations for the full year 2003. You should read the interim financial statements in conjunction with our consolidated financial statements and notes thereto presented in the TEPPCO Partners, L.P. Annual Report on Form 10-K for the year ended December 31, 2002. We have reclassified certain amounts from prior periods to conform with the current presentation.

We operate and report in three business segments: transportation and storage of refined products, liquefied petroleum gases ("LPGs") and petrochemicals ("Downstream Segment"); gathering, transportation, marketing and storage of crude oil and distribution of lubrication oils and specialty chemicals ("Upstream Segment"); and gathering of natural gas, fractionation of natural gas liquids ("NGLs") and transportation of NGLs ("Midstream Segment"). Our reportable segments offer different products and services and are managed separately because each requires different business strategies.

Our interstate transportation operations, including rates charged to customers, are subject to regulations prescribed by the Federal Energy Regulatory Commission ("FERC"). We refer to refined products, LPGs, petrochemicals, crude oil, NGLs and natural gas in this Report, collectively, as "petroleum products" or "products."

Basic net income per Limited Partner and Class B Unit (collectively, "Units") is computed by dividing net income, after deduction of the General Partner's interest, by the weighted average number of Units outstanding (a total of 57.7 million Units for the three months ended March 31, 2003, and 44.6 million Units for the three months ended March 31, 2002). The General Partner's percentage interest in net income is based on its percentage of cash distributions from Available Cash for each period (see Note 10. Quarterly Distributions of Available Cash). The General Partner was allocated \$9.2 million (representing 26.99%) and \$6.4 million (representing 23.95%) of net income for the three months ended March 31, 2003, and 2002, respectively. The General Partner's percentage interest in our net income increases as cash distributions paid per Unit increases.

Diluted net income per Unit is similar to the computation of basic net income per Unit above, except that the denominator is increased to include the dilutive effect of outstanding Unit options by application of the treasury

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS – (Continued) (Unaudited)

stock method. For the three months ended March 31, 2003, and 2002, the denominator was increased by 18,313 Units and 40,620 Units, respectively.

NOTE 2. NEW ACCOUNTING PRONOUNCEMENTS

In December 2002, the Financial Accounting Standards Board ("FASB") issued Statement of Financial Accounting Standards ("SFAS") No. 148, Accounting for Stock-Based Compensation – Transition and Disclosure. SFAS 148 amends SFAS No. 123, Accounting for Stock-Based Compensation, to provide alternative methods of transition for a voluntary change to the fair value based method of accounting for stock-based employee compensation. In addition, SFAS 148 amends the disclosure requirements of SFAS 123 to require prominent disclosure in both annual and interim financial statements about the method of accounting for stock-based employee compensation and the effect of the method used on the reported results. The provisions of SFAS 148 are effective for financial statements for fiscal years ending after December 15, 2002. The adoption of SFAS 148 did not affect our financial position, results of operations or cash flows.

In January 2003, the FASB issued FASB Interpretation No. 46, *Consolidation of Variable Interest Entities, an interpretation of ARB No. 51* ("FIN 46"). FIN 46 requires certain variable interest entities to be consolidated by the primary beneficiary of the entity if the equity investors in the entity do not have the characteristics of a controlling financial interest or do not have sufficient equity at risk for the entity to finance its activities without additional subordinated financial support from other parties. We are required to apply FIN 46 to all new variable interest entities created or acquired after January 31, 2003. For variable interest entities created or acquired prior to February 1, 2003, we are required to apply FIN 46 on July 1, 2003. In connection with the adoption of FIN 46, we evaluated our investments in Centennial Pipeline LLC, Seaway Crude Pipeline Company and Mont Belvieu Storage Partners, L.P. and determined that these entities are not variable interest entities as defined by FIN 46, and thus we have accounted for them as equity method investments (see Note 8. Equity Investments). The adoption of FIN 46 did not have an effect on our financial position, results of operations or cash flows.

NOTE 3. ASSET RETIREMENT OBLIGATIONS

In June 2001, the FASB issued SFAS No. 143, *Accounting for Asset Retirement Obligations*. SFAS 143 requires us to record the fair value of an asset retirement obligation as a liability in the period in which we incur a legal obligation for the retirement of tangible long-lived assets. A corresponding asset is also recorded and depreciated over the life of the asset. After the initial measurement of the asset retirement obligation, the liability will be adjusted at the end of each reporting period to reflect changes in the estimated future cash flows underlying the obligation. Determination of any amounts recognized upon adoption is based upon numerous estimates and assumptions, including future retirement costs, future inflation rates and the credit-adjusted risk-free interest rates.

The Downstream Segment assets consist primarily of a pipeline system and a series of storage facilities that originate along the upper Texas Gulf Coast and extend through the Midwest and northeastern United States. We transport refined products, LPGs and petrochemicals through the pipeline system. These products are primarily received in the south end of the system and stored and/or transported to various points along the system per customer nominations. The Upstream Segment's operations include purchasing crude oil from producers at the wellhead and providing delivery, storage and other services to its customers. The properties in the Upstream Segment consist of interstate trunk pipelines, pump stations, trucking facilities, storage tanks and various gathering systems primarily in Texas and Oklahoma. The Midstream Segment gathers natural gas from wells owned by producers and transports natural gas and NGLs on its pipeline systems, primarily in Texas, Wyoming, New Mexico and Colorado. The Midstream Segment also owns and operates two NGL fractionator facilities in Colorado.

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS – (Continued) (Unaudited)

We have completed our assessment of SFAS 143, and we have determined that we are obligated by contractual or regulatory requirements to remove facilities or perform other remediation upon retirement of our assets. However, we are not able to determine the fair value of the asset retirement obligations for our trunk, interstate and gathering pipelines and our surface facilities as they cannot be reasonably estimated, since future dismantlement and removal dates are indeterminate.

In order to determine a removal date for our gathering lines and related surface assets, reserve information regarding the life of the specific field is required. As a transporter of crude oil and natural gas, we are not a producer of the field reserves and as such, we do not have access to reserve data on those fields in which we gather crude oil and natural gas. The lack of adequate reserve data does not allow us to reasonably estimate cash flows from the fields for future periods, and we are not able to make a reasonable estimate when future dismantlement and removal dates of our gathering assets will occur. With regard to our trunk and interstate pipelines and their related surface assets, it is impossible to predict when demand for transportation of the related products will cease. Our rights-of-way agreements allow us to renew the rights-of-way rather than remove the pipe while we evaluate our trunk pipelines for alternative uses, which can be and have been found, should the need arise.

We will record such asset retirement obligations in the period in which more information becomes available for us to reasonably estimate the settlement dates of the retirement obligations. The adoption of SFAS 143 did not have an impact on our financial position, results of operations or cash flows.

NOTE 4. GOODWILL AND OTHER INTANGIBLE ASSETS

Goodwill represents the excess of purchase price over fair value of net assets acquired and is presented on the consolidated balance sheets net of accumulated amortization. We account for goodwill under SFAS No. 142, *Goodwill and Other Intangible Assets*, which was issued by the FASB in July 2001. SFAS 142 prohibits amortization of goodwill and intangible assets with indefinite useful lives, but instead should be tested for impairment at least annually. SFAS 142 requires that intangible assets with definite useful lives be amortized over their respective estimated useful lives. We will assess the useful lives and residual values of all intangible assets on an annual basis to determine if adjustments are required.

To perform an impairment test of goodwill, we have identified our reporting units and have determined the carrying value of each reporting unit by assigning the assets and liabilities, including the existing goodwill and intangible assets, to those reporting units. We would then determine the fair value of each reporting unit and compare it to the carrying value of the reporting unit. We will continue to compare the fair value of each reporting unit to its carrying value on an annual basis to determine if an impairment loss has occurred.

At March 31, 2003, we had \$16.9 million of unamortized goodwill and \$25.5 million of excess investment in our equity investment in Seaway Crude Pipeline Company (equity method goodwill). The excess investment is included in our equity investments account at March 31, 2003. The following table presents the carrying amount of goodwill and equity method goodwill at March 31, 2003, by business segment (in thousands):

Downstream Segment	Midstream Segment	Upstream Segment	Segments Total
\$ —	\$2,777	\$14,167	\$16,944
—	—	25,502	25,502
	Segment \$	Segment Segment \$ \$2,777	Segment Segment Segment \$ \$2,777 \$14,167

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS – (Continued)

(Unaudited)

The following table reflects the components of amortized intangible assets, excluding goodwill (in thousands):

	March 31, 2003		December 31, 2002	
	Gross Carrying Amount	Accumulated Amortization	Gross Carrying Amount	Accumulated Amortization
mortized intangible assets:				
Fractionation agreement	\$ 38,000	\$ (9,500)	\$ 38,000	\$ (9,025)
Natural gas gathering contracts	462,449	(39,541)	462,449	(28,710)
Other	3,745	(1,256)	3,745	(1,085)
Total	\$504,194	\$(50,297)	\$504,194	\$(38,820)

Excluding goodwill, amortization expense on intangible assets was \$11.5 million and \$4.6 million for the three months ended March 31, 2003 and 2002, respectively.

The following table sets forth the estimated amortization expense on intangible assets for the years ending December 31 (in thousands):

2003	\$45,824	
2004	47,653	
2005	51,359	
2006	48,619	
2007	44,646	

NOTE 5. DERIVATIVE FINANCIAL INSTRUMENTS

We have entered into an interest rate swap agreement to hedge our exposure to increases in the benchmark interest rate underlying our variable rate revolving credit facility. The term of the interest rate swap matches the maturity of the credit facility. We designated this swap agreement, which hedges exposure to variability in expected future cash flows attributed to changes in interest rates, as a cash flow hedge. The swap agreement is based on a notional amount of \$250.0 million. Under the swap agreement, we pay a fixed rate of interest of 6.955% and receive a floating rate based on a three month U.S. Dollar LIBOR rate. Since this swap is designated as a cash flow hedge, the changes in fair value, to the extent the swap is effective, are recognized in other comprehensive income until the hedged interest costs are recognized in earnings. During the three months ended March 31, 2003, and 2002, we recognized increases in interest expense of \$3.4 million and \$3.2 million, respectively, related to the difference between the fixed rate and the floating rate of interest on the interest rate swap. During the three months ended March 31, 2003, we measured to the interest rate swap and noted that no gain or loss from ineffectiveness was required to be recognized. The fair value of the interest rate swap agreement was a loss of approximately \$14.6 million and \$20.1 million at March 31, 2003, and December 31, 2002, respectively. We anticipate that approximately \$14.3 million of the fair value will be transferred into earnings over the next twelve months.

On October 4, 2001, our TE Products subsidiary entered into an interest rate swap agreement to hedge its exposure to changes in the fair value of its fixed rate 7.51% Senior Notes due 2028. We designated this swap agreement as a fair value hedge. The swap agreement has a notional amount of \$210.0 million and matures in January 2028 to match the principal and maturity of the TE Products Senior Notes. Under the swap agreement, TE Products pays a floating rate based on a three month U.S. Dollar LIBOR rate, plus a spread, and receives a fixed rate of interest of 7.51%. During the three months ended March 31, 2003, and 2002, we recognized reductions in interest

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS – (Continued) (Unaudited)

expense of \$2.4 million and \$1.7 million, respectively, related to the difference between the fixed rate and the floating rate of interest on the interest rate swap. During the three months ended March 31, 2003, we measured the hedge effectiveness of this interest rate swap and noted that no gain or loss from ineffectiveness was required to be recognized. The fair value of this interest rate swap agreement was a gain of approximately \$12.3 million and \$13.6 million at March 31, 2003, and December 31, 2002, respectively.

On February 20, 2002, we entered into interest rate swap agreements, designated as fair value hedges, to hedge our exposure to changes in the fair value of our fixed rate 7.625% Senior Notes due 2012. The swap agreements had a combined notional amount of \$500.0 million and matured in 2012 to match the principal and maturity of the Senior Notes. Under the swap agreements, we paid a floating rate based on a U.S. Dollar LIBOR rate, plus a spread, and received a fixed rate of interest of 7.625%. On July 16, 2002, the swap agreements were terminated resulting in a gain of approximately \$18.0 million. Concurrent with the swap terminations, we entered into new interest rate swap agreements, with identical terms as the previous swap agreements; however, the floating rate was based upon a spread of an additional 50 basis points. In December 2002, the swap agreements entered into on July 16, 2002, were terminated, resulting in a gain of approximately \$26.9 million. The gains realized from the July 2002 and December 2002 swap terminations have been deferred as adjustments to the carrying value of the Senior Notes and are being amortized using the effective interest method as reductions to future interest expense over the remaining term of the Senior Notes. In the event of early extinguishment of the Senior Notes, any remaining unamortized gains would be recognized in the consolidated statement of income at the time of extinguishment.

NOTE 6. ACQUISITIONS

On March 1, 2002, we completed the purchase of the Chaparral NGL system ("Chaparral") for \$132.4 million from Diamond-Koch II, L.P. and Diamond-Koch III, L.P., including acquisition related costs of approximately \$0.4 million. We funded the purchase by a borrowing under our \$500.0 million revolving credit facility (see Note 9. Debt). Chaparral is an NGL pipeline system that extends from West Texas and New Mexico to Mont Belvieu. The pipeline delivers NGLs to fractionators and to our existing storage in Mont Belvieu. Under a contractual agreement, DEFS manages and operates Chaparral on our behalf. We accounted for the acquisition of these assets under the purchase method of accounting. We allocated the purchase price to property, plant and equipment. Accordingly, the results of the acquisition are included in the consolidated financial statements from March 1, 2002.

On June 30, 2002, we completed the purchase of Val Verde Gas Gathering Company ("Val Verde") for \$444.2 million from Burlington Resources Gathering Inc., a subsidiary of Burlington Resources Inc., including acquisition related costs of approximately \$1.2 million. We funded the purchase by borrowings of \$168.0 million under our \$500.0 million revolving credit facility, \$72.0 million under our 364-day revolving credit facility and \$200.0 million under a six-month term loan with SunTrust Bank (see Note 9. Debt). The remaining purchase price was funded through working capital sources of cash. The Val Verde system gathers coal bed methane ("CBM") from the Fruitland Coal Formation of the San Juan Basin in New Mexico and Colorado. The system is one of the largest CBM gathering and treating facilities in the United States. Under a contractual agreement, DEFS manages and operates Val Verde on our behalf. We accounted for the acquisition of these assets under the purchase method of accounting. Accordingly, the results of the acquisition are included in the consolidated financial statements from June 30, 2002.

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS – (Continued) (Unaudited)

The following table allocates the estimated fair value of the Val Verde assets acquired on June 30, 2002 (in thousands):

Property, plant and equipment	\$205,146
Intangible assets (primarily gas gathering contracts)	239,649
Total assets	444,795
Total liabilities assumed	(645)
Net assets acquired	\$444,150

The value assigned to intangible assets relates to fixed-term contracts with customers. We are amortizing the value assigned to intangible assets on a unit of production basis, based upon the actual throughput of the system over the expected total throughput for the contracts (averaging approximately 20 years).

The following table presents our unaudited pro forma results as though the acquisition of Val Verde occurred at the beginning of 2002 (in thousands, except per Unit amounts). The unaudited pro forma results give effect to certain pro forma adjustments including depreciation and amortization expense adjustments of property, plant and equipment and intangible assets based upon the purchase price allocations, interest expense related to financing the acquisition, amortization of debt issue costs and the removal of income tax effects in historical results of operations. The pro forma results do not include operating efficiencies or revenue growth from historical results.

	Three Months Ended March 31, 2002
Revenues	\$649,241
Operating income	42,505
Net income	30,677
Basic and diluted net income per Limited Partner and Class B Unit	\$ 0.42

The summarized pro forma information has been prepared for comparative purposes only. It is not intended to be indicative of the actual operating results that would have occurred had the acquisition been consummated at the beginning of 2002, or the results which may be attained in the future.

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS – (Continued) (Unaudited)

NOTE 7. INVENTORIES

Inventories are carried at the lower of cost (based on weighted average cost method) or market. The major components of inventories were as follows (in thousands):

	March 31, 2003	December 31, 2002
Crude oil	\$ 4,377	\$ —
Gasolines	4,580	4,700
Butanes	3,415	1,991
Transmix	1,934	2,526
Other products	4,419	3,836
Materials and supplies	4,837	4,113
Total	\$23,562	\$17,166

The costs of inventories did not exceed market values at March 31, 2003, and December 31, 2002.

NOTE 8. EQUITY INVESTMENTS

Through one of our indirect wholly owned subsidiaries, we own a 50% ownership interest in Seaway Crude Pipeline Company ("Seaway"). The remaining 50% interest is owned by ConocoPhillips. Seaway owns a pipeline that carries mostly imported crude oil from a marine terminal at Freeport, Texas, to Cushing, Oklahoma, and from a marine terminal at Texas City, Texas, to refineries in the Texas City and Houston areas. The Seaway Crude Pipeline Company Partnership Agreement provides for varying participation ratios throughout the life of the Seaway partnership. From July 20, 2000, through May 2002, we received 80% of revenue and expense of Seaway. From June 2002 through May 2006, we receive 60% of revenue and expense of Seaway. Thereafter, the sharing ratio becomes 40% of revenue and expense to us. For the year ended December 31, 2002, our portion of equity earnings on a pro-rated basis averaged approximately 67%.

In August 2000, TE Products entered into agreements with Panhandle Eastern Pipeline Company ("PEPL"), a subsidiary of CMS Energy Corporation, and Marathon Ashland Petroleum LLC ("Marathon") to form Centennial Pipeline LLC ("Centennial"). Centennial owns an interstate refined petroleum products pipeline extending from the upper Texas Gulf Coast to Illinois. Through February 9, 2003, each participant owned a one-third interest in Centennial. On February 10, 2003, TE Products and Marathon each acquired an additional interest in Centennial from PEPL for \$20.0 million each, increasing their percentage ownerships in Centennial to 50% each. During the three months ended March 31, 2003, excluding the amount paid for the acquisition of the additional ownership interest, TE Products contributed approximately \$1.0 million for its investment in Centennial, which is included in the equity investment balance at March 31, 2003.

As of January 1, 2003, TE Products and Louis Dreyfus Energy Services, L.P. ("Louis Dreyfus") effectively formed Mont Belvieu Storage Partners, L.P. ("MB Storage"). TE Products and Louis Dreyfus each own a 50% ownership interest in MB Storage. The purpose of MB Storage is to expand services to the upper Texas Gulf Coast energy marketplace by increasing pipeline throughput and the mix of products handled through the existing system and establishing new receipt and delivery connections. MB Storage is a service-oriented, fee-based venture with no commodity trading activity. TE Products continues to operate the facilities for MB Storage. Effective January 1, 2003, TE Products contributed property and equipment with a net book value of \$75.5 million to MB Storage. Additionally, as of the contribution date, Louis Dreyfus had invested \$6.1 million for expansion projects for MB

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS – (Continued) (Unaudited)

Storage that TE Products was required to reimburse if the original joint development and marketing agreement was terminated by either party. This deferred liability was also contributed and converted to the capital of Louis Dreyfus in MB Storage.

We use the equity method of accounting to account for our investments in Seaway, Centennial and MB Storage. Summarized combined financial information for Seaway and Centennial as of and for the three months ended March 31, 2003 and 2002, and MB Storage for the three months ended March 31, 2003, is presented below (in thousands):

		Three Months Ended March 31,	
	2003	2002	
Revenues	\$24,322	\$14,796	
Net income	5,286	3,455	

Summarized combined balance sheet data for Seaway and Centennial as of March 31, 2003, and December 31, 2002, and for MB Storage as of March 31, 2003, is presented below (in thousands):

	March 31, 2003	December 31, 2002
Current assets	\$ 48,465	\$ 32,528
Noncurrent assets	622,254	551,324
Current liabilities	34,029	28,681
Long-term debt	140,000	140,000
Noncurrent liabilities	13,758	14,875
Partners' capital	482,932	400,296

Our investments in Seaway and Centennial at March 31, 2003, and December 31, 2002, include excess net investment amounts of \$25.5 million and \$33.1 million, respectively. Excess investment is the amount by which our investment balance exceeds our proportionate share of the net assets of the investment. Prior to January 1, 2002, and the adoption of SFAS 142, we were amortizing the excess investment in Seaway using the straight-line method over 20 years.

NOTE 9. DEBT

Senior Notes

On January 27, 1998, TE Products completed the issuance of \$180.0 million principal amount of 6.45% Senior Notes due 2008, and \$210.0 million principal amount of 7.51% Senior Notes due 2028 (collectively the "TE Products Senior Notes"). The 6.45% TE Products Senior Notes were issued at a discount of \$0.3 million and are being accreted to their face value over the term of the notes. The 6.45% TE Products Senior Notes due 2008 are not subject to redemption prior to January 15, 2008. The 7.51% TE Products Senior Notes due 2028, issued at par, may be redeemed at any time after January 15, 2008, at the option of TE Products, in whole or in part, at a premium.

The TE Products Senior Notes do not have sinking fund requirements. Interest on the TE Products Senior Notes is payable semiannually in arrears on January 15 and July 15 of each year. The TE Products Senior Notes are unsecured obligations of TE Products and rank on a parity with all other unsecured and unsubordinated indebtedness of TE Products. The indenture governing the TE Products Senior Notes contains covenants, including, but not

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS – (Continued) (Unaudited)

limited to, covenants limiting the creation of liens securing indebtedness and sale and leaseback transactions. However, the indenture does not limit our ability to incur additional indebtedness. As of March 31, 2003, TE Products was in compliance with the covenants of the TE Products Senior Notes.

On February 20, 2002, we completed the issuance of \$500.0 million principal amount of 7.625% Senior Notes due 2012. The 7.625% Senior Notes were issued at a discount of \$2.2 million and are being accreted to their face value over the term of the notes. We used the proceeds from the offering to reduce a portion of the outstanding balances of our credit facilities, including those issued in connection with the acquisition of Jonah Gas Gathering Company ("Jonah"). The Senior Notes may be redeemed at any time at our option with the payment of accrued interest and a make-whole premium determined by discounting remaining interest and principal payments using a discount rate equal to the rate of the United States Treasury securities of comparable remaining maturity plus 35 basis points. The indenture governing our 7.625% Senior Notes contains covenants, including, but not limited to, covenants limiting the creation of liens securing indebtedness and sale and leaseback transactions. However, the indenture does not limit our ability to incur additional indebtedness. As of March 31, 2003, we were in compliance with the covenants of these Senior Notes.

On January 30, 2003, we completed the issuance of \$200.0 million principal amount of 6.125% Senior Notes due 2013. The 6.125% Senior Notes were issued at a discount of \$1.4 million and are being accreted to their face value over the term of the notes. We used \$182.0 million of the proceeds from the offering to reduce the outstanding principal on our \$500.0 million revolving credit facility to \$250.0 million. The balance of the net proceeds received was used for general purposes. The Senior Notes may be redeemed at any time at our option with the payment of accrued interest and a make-whole premium determined by discounting remaining interest and principal payments using a discount rate equal to the rate of the United States Treasury securities of comparable remaining maturity plus 35 basis points. The indenture governing our 6.125% Senior Notes contains covenants, including, but not limited to, covenants limiting the creation of liens securing indebtedness and sale and leaseback transactions. However, the indenture does not limit our ability to incur additional indebtedness. As of March 31, 2003, we were in compliance with the covenants of these Senior Notes.

We have entered into interest rate swap agreements to hedge our exposure to changes in the fair value on a portion of the Senior Notes discussed above. See Note 5. Derivative Financial Instruments.

Other Long Term Debt and Credit Facilities

On April 6, 2001, we entered into a \$500.0 million revolving credit facility including the issuance of letters of credit of up to \$20.0 million ("Three Year Facility"). The interest rate is based, at our option, on either the lender's base rate plus a spread, or LIBOR plus a spread in effect at the time of the borrowings. The credit agreement for the Three Year Facility contains certain restrictive financial covenant ratios. During 2002, borrowings under the Three Year Facility were used to finance the acquisitions of the Chaparral NGL system on March 1, 2002, and Val Verde on June 30, 2002. During 2002, repayments were made on the Three Year Facility with proceeds from the issuance of our 7.625% Senior Notes, proceeds from the issuance of additional Units and proceeds from the termination of interest rate swaps (see Note 5. Derivative Financial Instruments). During the first quarter of 2003, we repaid \$182.0 million of the outstanding balance of the Three Year Facility with proceeds from the issuance of our 6.125% Senior Notes on January 30, 2003. At March 31, 2003, \$265.0 million was outstanding under the Three Year Facility at a weighted average interest rate of 2.3%. As of March 31, 2003, we were in compliance with the covenants contained in this credit agreement.

We have entered into an interest rate swap agreement to hedge our exposure to increases in interest rates on a portion of the Three Year Facility discussed above. See Note 5. Derivative Financial Instruments.

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS – (Continued) (Unaudited)

Short Term Credit Facilities

On April 6, 2001, we entered into a 364-day, \$200.0 million revolving credit agreement ("Short-term Revolver"). The interest rate was based, at our option, on either the lender's base rate plus a spread, or LIBOR plus a spread in effect at the time of the borrowings. The credit agreement contained certain restrictive financial covenant ratios. On March 28, 2002, the Short-term Revolver was extended for an additional period of 364 days, ending in March 2003. During 2002, borrowings under the Short-term Revolver were used to finance the acquisition of the Val Verde assets and for other purposes. During 2002, we repaid the existing amounts outstanding under the Short-term Revolver with proceeds we received from the issuance of Units in 2002. The Short-term Revolver expired on March 27, 2003.

On June 27, 2002, we entered into a \$200.0 million six-month term loan with SunTrust Bank ("Six-Month Term Loan") payable in December 2002. We borrowed \$200.0 million under the Six-Month Term Loan to acquire the Val Verde assets (see Note 6. Acquisitions). The interest rate was based, at our option, on either the lender's base rate plus a spread, or LIBOR plus a spread in effect at the time of the borrowings. The credit agreement contained certain restrictive financial covenant ratios. On July 11, 2002, we repaid \$90.0 million of the outstanding principal from proceeds primarily received from the issuance of Units in July 2002. On September 10, 2002, we repaid the remaining outstanding balance of \$110.0 million with proceeds received from the issuance of Units in September 2002, and canceled the facility.

The following table summarizes the principal outstanding under our credit facilities as of March 31, 2003, and December 31, 2002 (in thousands):

	March 31, 2003	December 31, 2002
Long Term Credit Facilities:		
Three Year Facility, due April 2004	\$ 265,000	\$ 432,000
6.45% TE Products Senior Notes, due January 2008	179,853	179,845
7.51% TE Products Senior Notes, due January 2028	210,000	210,000
7.625% Senior Notes, due February 2012	498,051	497,995
6.125% Senior Notes, due February 2013	198,594	
Total borrowings	1,351,498	1,319,840
Adjustment to carrying value associated with hedges of fair value	55,665	57,852
Total Long Term Credit Facilities	\$1,407,163	\$1,377,692

NOTE 10. QUARTERLY DISTRIBUTIONS OF AVAILABLE CASH

We make quarterly cash distributions of all of our Available Cash, generally defined as consolidated cash receipts less consolidated cash disbursements and cash reserves established by the General Partner in its sole discretion. Pursuant to the Partnership Agreement, the Company receives incremental incentive cash distributions when cash distributions exceed certain target thresholds as follows:

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS – (Continued) (Unaudited)

	Unitholders	General Partner
Quarterly Cash Distribution per Unit:		
Up to Minimum Quarterly Distribution (\$0.275 per Unit)	98%	2%
First Target - \$0.276 per Unit up to \$0.325 per Unit	85%	15%
Second Target - \$0.326 per Unit up to \$0.45 per Unit	75%	25%
Over Second Target - Cash distributions greater than \$0.45 per Unit	50%	50%

The following table reflects the allocation of total distributions paid during the three months ended March 31, 2003 and 2002 (in thousands, except per Unit amounts).

	Three Months E	nded March 31,
	2003	2002
Limited Partner Units	\$32,286	\$23,259
General Partner Ownership Interest	707	521
General Partner Incentive	11,191	7,514
Total Partners' Capital Cash Distributions	44,184	31,294
Class B Units	2,349	2,251
Total Cash Distributions Paid	\$46,533	\$33,545
Total Cash Distributions Paid Per Unit	\$ 0.600	\$ 0.575

On May 9, 2003, we will pay a cash distribution of \$0.625 per Unit for the quarter ended March 31, 2003. The first quarter 2003 cash distribution will total \$49.4 million.

NOTE 11. SEGMENT DATA

We have three reporting segments: transportation and storage of refined products, LPGs and petrochemicals, which operates as the Downstream Segment; gathering, transportation, marketing and storage of crude oil and distribution of lubrication oils and specialty chemicals, which operates as the Upstream Segment; and gathering of natural gas, fractionation of NGLs and transportation of NGLs, which operates as the Midstream Segment. The amounts indicated below as "Partnership and Other" relate primarily to intercompany eliminations and assets that we hold that have not been allocated to any of our reporting segments.

Our Downstream Segment revenues are earned from transportation and storage of refined products and LPGs, intrastate transportation of petrochemicals, sale of product inventory and other ancillary services. The two largest operating expense items of the Downstream Segment are labor and electric power. We generally realize higher revenues during the first and fourth quarters of each year since our operations are somewhat seasonal. Refined products volumes are generally higher during the second and third quarters because of greater demand for gasolines during the spring and summer driving seasons. LPGs volumes are generally higher from November through March due to higher demand in the Northeast for propane, a major fuel for residential heating. Our Downstream Segment also includes the results of operations of the northern portion of the Dean pipeline. Beginning in January 2003, the northern portion of the Dean pipeline was converted to transport refinery grade propylene ("RGPs") from Mont Belvieu to Point Comfort, Texas. As a result, the revenues and expenses of the northern portion of the Dean pipeline are included in the Downstream Segment. Our Downstream Segment also includes our equity investments in Centennial and MB Storage (see Note 8. Equity Investments).

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS – (Continued) (Unaudited)

Our Upstream Segment revenues are earned from gathering, transportation, marketing and storage of crude oil and distribution of lubrication oils and specialty chemicals, principally in Oklahoma, Texas and the Rocky Mountain region. Marketing operations consist primarily of aggregating purchased crude oil along our pipeline systems, or from third party pipeline systems, and arranging the necessary logistics for the ultimate sale of the crude oil to local refineries, marketers or other end users. Our Upstream Segment also includes the equity earnings from our investment in Seaway. Seaway consists of large diameter pipelines that transport crude oil from Seaway's marine terminals on the U.S. Gulf Coast to Cushing, Oklahoma, a central crude oil distribution point for the Central United States, and to refineries in the Texas City and Houston areas.

Our Midstream Segment revenues are earned from the fractionation of NGLs in Colorado, transportation of NGLs from two trunkline NGL pipelines in South Texas, two NGL pipelines in East Texas and a pipeline system ("Chaparral") from West Texas and New Mexico to Mont Belvieu; the gathering of natural gas in the Green River Basin in southwestern Wyoming, through Jonah and the gathering of CBM from the Fruitland Coal Formation of the San Juan Basin in New Mexico and Colorado, through Val Verde. Chaparral was acquired on March 1, 2002, for \$132.4 million from Diamond-Koch II, L.P. and Diamond-Koch III, L.P., including acquisition related costs of approximately \$0.4 million. The Val Verde assets were acquired on June 30, 2002, for \$444.2 million from a subsidiary of Burlington Resources Inc., including acquisition related costs of approximately \$1.2 million. The Val Verde, Jonah and Chaparral assets are managed and operated by DEFS under contractual agreements. The results of operations of the Chaparral and Val Verde acquisitions are included in periods subsequent to their respective acquisition dates (see Note 6. Acquisitions).

The table below includes interim financial information by reporting segment for the interim periods ended March 31, 2003 and 2002 (in thousands):

	Three Months Ended March 31, 2003					
	Downstream Segment	Upstream Segment	Midstream Segment	Segments Total	Partnership and Other	Consolidated
Revenues	\$67,965	\$985,380	\$46,904	\$1,100,249	\$(1,010)	\$1,099,239
Purchases of petroleum products	_	963,854	_	963,854	(1,010)	962,844
Operating expenses, including power	31,549	14,837	11,354	57,740		57,740
Depreciation and amortization						
expense	7,147	3,065	17,101	27,313	_	27,313
Operating income	29,269	3,624	18,449	51,342	_	51,342
Equity earnings	(1,259)	4,969	_	3,710	_	3,710
Other income, net	(6)	189	37	220	(38)	182
Earnings before interest	\$28,004	\$ 8,782	\$18,486	\$ 55,272	\$ (38)	\$ 55,234
-						

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS – (Continued) (Unaudited)

	Three Months Ended March 31, 2002						
	Downstream Segment	Upstream Segment	Midstream Segment	Segments Total	Partnership and Other	Consolidated	
Revenues	\$59,586	\$553,888	\$18,370	\$631,844	\$(707)	\$631,137	
Purchases of petroleum products	_	533,678	_	533,678	(707)	532,971	
Operating expenses, including power	29,107	11,939	3,493	44,539	_	44,539	
Depreciation and amortization expense	6,832	2,064	7,145	16,041	_	16,041	
Operating income	23,647	6,207	7,732	37,586	_	37,586	
Equity earnings	(796)	4,368	_	3,572	_	3,572	
Other income, net	124	185	19	328	_	328	
Earnings before interest	\$22,975	\$ 10,760	\$ 7,751	\$ 41,486	\$	\$ 41,486	

The following table provides the total assets, capital expenditures and significant non-cash investing activities for each segment as of March 31, 2003, and December 31, 2002 (in thousands):

	Downstream Segment	Upstream Segment	Midstream Segment	Segments Total	Partnership and Other	Consolidated
March 31, 2003:						
Total assets	\$911,820	\$813,416	\$1,156,816	\$2,882,052	\$(36,607)	\$2,845,445
Capital expenditures	9,625	2,291	3,518	15,434	_	15,434
Non-cash investing activities	69,459		_	69,459		69,459
December 31, 2002:						
Total assets	883,450	724,860	1,174,010	2,782,320	(11,678)	2,770,642
Capital expenditures	60,900	10,212	62,260	133,372	_	133,372

The following table reconciles the segments total earnings before interest to consolidated net income (in thousands):

	Three Months 1	Three Months Ended March 31,		
	2003	2002		
Earnings before interest	\$ 55,234	\$ 41,486		
Interest expense	(21,905)	(16,787)		
Interest capitalized	596	2,109		
Net income	\$ 33,925	\$ 26,808		

NOTE 12. COMMITMENTS AND CONTINGENCIES

In the fall of 1999 and on December 1, 2000, the General Partner and the Partnership were named as defendants in two separate lawsuits in Jackson County Circuit Court, Jackson County, Indiana, styled *Ryan E. McCleery and Marcia S. McCleery, et. al. v. Texas Eastern Corporation, et. al.* (including the General Partner and Partnership) and *Gilbert Richards and Jean Richards v. Texas Eastern Corporation, et. al.* (including the General Partner and Partnership). In both cases, the plaintiffs contend, among other things, that we and other defendants

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS – (Continued) (Unaudited)

stored and disposed of toxic and hazardous substances and hazardous wastes in a manner that caused the materials to be released into the air, soil and water. They further contend that the release caused damages to the plaintiffs. In their complaints, the plaintiffs allege strict liability for both personal injury and property damage together with gross negligence, continuing nuisance, trespass, criminal mischief and loss of consortium. The plaintiffs are seeking compensatory, punitive and treble damages. We have filed an answer to both complaints, denying the allegations, as well as various other motions. These cases are not covered by insurance. Discovery is ongoing, and we are defending ourselves vigorously against the lawsuits. The plaintiffs have not stipulated the amount of damages that they are seeking in the suit. We cannot estimate the loss, if any, associated with these pending lawsuits.

On December 21, 2001, TE Products was named as a defendant in a lawsuit in the 10th Judicial District, Natchitoches Parish, Louisiana, styled *Rebecca L. Grisham et. al. v. TE Products Pipeline Company, Limited Partnership.* In this case, the plaintiffs contend that our pipeline, which crosses the plaintiff's property, leaked toxic products onto the plaintiff's property. The plaintiffs further contend that this leak caused damages to the plaintiffs. We have filed an answer to the plaintiff's petition denying the allegations. The plaintiffs have not stipulated the amount of damages they are seeking in the suit. We are defending ourselves vigorously against the lawsuit. We cannot estimate the damages, if any, associated with this pending lawsuit; however, this case is covered by insurance.

On April 19, 2002, we, through our subsidiary TEPPCO Crude Oil, L.P., filed a declaratory judgment action in the U.S. District Court for the Western District of Oklahoma against D.R.D. Environmental Services, Inc. ("D.R.D.") seeking resolution of billing and other contractual disputes regarding potential overcharges for environmental remediation services provided by D.R.D. On May 28, 2002, D.R.D. filed a counterclaim for alleged breach of contract in the amount of \$2,243,525, and for unspecified damages for alleged tortious interference with D.R.D.'s contractual relations with DEFS. We have denied the counterclaims. Discovery is ongoing. If D.R.D. should be successful, management believes that a substantial portion of the \$2,243,525 breach of contract claim will be covered under an indemnity from DEFS. We cannot predict the outcome of the litigation against us; however, we are defending ourselves vigorously against the counterclaim. We do not believe that the outcome of this lawsuit will have a material adverse effect on our financial position, results of operations or cash flows.

In addition to the litigation discussed above, we have been, in the ordinary course of business, a defendant in various lawsuits and a party to various other legal proceedings, some of which are covered in whole or in part by insurance. We believe that the outcome of these lawsuits and other proceedings will not individually or in the aggregate have a material adverse effect on our consolidated financial position, results of operations or cash flows.

In February 2002, a producer on the Jonah system sent a letter to Alberta Energy Company implying that as a result of our acquisition of the Jonah system, it may have a right to acquire all or a portion of the assets comprising the Jonah system pursuant to an alleged right of first refusal in a gas gathering agreement between the producer and Jonah. Subsidiaries of Alberta Energy Company have agreed to indemnify us against losses resulting from any breach of representations concerning the absence of third party rights in connection with our acquisition of the entity that owns the Jonah system. We believe that we have adequate legal defenses if the producer should assert a claim, and we also believe that no right of first refusal on any of the underlying Jonah system assets has been triggered.

Our operations are subject to federal, state and local laws and regulations governing the discharge of materials into the environment. Failure to comply with these laws and regulations may result in the assessment of administrative, civil and criminal penalties, imposition of injunctions delaying or prohibiting certain activities and the need to perform investigatory and remedial activities. Although we believe our operations are in material compliance with applicable environmental laws and regulations, risks of significant costs and liabilities are inherent in pipeline operations, and we cannot assure you that significant costs and liabilities will not be incurred. Moreover, it is possible that other developments, such as increasingly strict environmental laws and regulations and enforcement policies thereunder, and claims for damages to property or persons resulting from our operations, could

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS – (Continued) (Unaudited)

result in substantial costs and liabilities to us. We believe that changes in environmental laws and regulations will not have a material adverse effect on our financial position, results of operations or cash flows in the near term.

In 1994, we entered into an Agreed Order with the Indiana Department of Environmental Management ("IDEM") that resulted in the implementation of a remediation program for groundwater contamination attributable to our operations at the Seymour, Indiana, terminal. In 1999, the IDEM approved a Feasibility Study, which includes our proposed remediation program. In March 2003, the IDEM issued a Record of Decision formally approving the remediation program. As the Record of Decision has been issued, we will enter into an Agreed Order for the continued operation and maintenance of the remediation program. We have an accrued liability of \$0.3 million at March 31, 2003, for future remediation costs at the Seymour terminal. We do not expect that the completion of the remediation program will have a future material adverse effect on our financial position, results of operations or cash flows.

In 1994, the Louisiana Department of Environmental Quality ("LDEQ") issued a compliance order for environmental contamination at our Arcadia, Louisiana, facility. This contamination may be attributable to our operations, as well as adjacent petroleum terminals operated by other companies. In 1999, our Arcadia facility and adjacent terminals were directed by the Remediation Services Division of the LDEQ to pursue remediation of this containment phase. At March 31, 2003, we have an accrued liability of \$0.2 million for remediation costs at our Arcadia facility. We do not expect that the completion of the remediation program that we have proposed will have a future material adverse effect on our financial position, results of operations or cash flows.

On March 17, 2003, we experienced a release of 511 barrels of jet fuel from a tank at our Blue Island terminal located in Cook County, Illinois. As a result of the release, we have entered into an Agreed Preliminary Injunction and Order ("Agreed Order") with the State of Illinois. The Agreed Order requires us, in part, to complete a site investigation plan to delineate the scope of any potential contamination resulting from the release and to remediate any contamination. The Agreed Order does not contain any provision for any fines or penalties; however, it does not preclude the State of Illinois from assessing these at a later date. We do not expect that the completion of the remediation program will have a future material adverse effect on our financial position, results of operations or cash flows.

At March 31, 2003, we have an accrued liability of \$6.6 million related to various TCTM sites requiring environmental remediation activities. We also have a receivable at March 31, 2003, of \$1.9 million from DEFS which is based on a contractual indemnity obligation we received in connection with our acquisition of assets from a DEFS affiliate in November 1998. The indemnity relates to future environmental remediation activities attributable to operations of these assets prior to our acquisition. Under this indemnity obligation, we are responsible for the first \$3.0 million in specified environmental liabilities, and DEFS is responsible for those environmental liabilities in excess of \$3.0 million, up to a maximum amount of \$25.0 million. At December 31, 2002, the receivable balance from DEFS was \$4.2 million. The majority of the receivable from DEFS related to remediation activities at the Velma crude oil site in Stephens County, Oklahoma. During the first quarter of 2003, we received \$2.4 million from DEFS as partial payment on the receivable balance. The accrued liability balance at March 31, 2003, also includes an accrual of \$2.3 million related to the Shelby crude oil site in Stephens County, Oklahoma. At March 31, 2003, it is uncertain if these costs related to Shelby are covered under the indemnity obligation from DEFS. We are currently in discussions with DEFS regarding these matters. We do not expect that the completion of remediation programs associated with TCTM activities will have a future material adverse effect on our financial position, results of operations or cash flows.

Centennial entered into credit facilities totaling \$150.0 million, and as of March 31, 2003, \$150.0 million was outstanding under those credit facilities. The proceeds were used to fund construction and conversion costs of its pipeline system. Each of the participants in Centennial, including TE Products, originally guaranteed one-third of Centennial's debt up to a maximum amount of \$50.0 million. During the third quarter of 2002, PEPL, one of the participants in Centennial, was downgraded by Moody's and Standard & Poors to below investment grade, which

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS – (Continued) (Unaudited)

resulted in PEPL being in default under its portion of the Centennial guaranty. Effective September 27, 2002, TE Products and Marathon increased their guaranteed amounts to one-half of the debt of Centennial, up to a maximum amount of \$75.0 million each, to avoid a default on the Centennial debt. As compensation to TE Products and Marathon for providing their additional guarantees, PEPL was required to pay interest at a rate of 4% per annum to each of TE Products and Marathon on the portion of the additional guaranty that each had provided for PEPL. In connection with the acquisition of the additional interest in Centennial on February 10, 2003, the guaranty agreement between TE Products, Marathon, and PEPL was terminated. TE Products' guaranty of up to a maximum of \$75.0 million of Centennial's debt remains in effect.

NOTE 13. COMPREHENSIVE INCOME

SFAS No. 130, *Reporting Comprehensive Income* requires certain items such as foreign currency translation adjustments, minimum pension liability adjustments, and unrealized gains and losses on certain investments to be reported in a financial statement. As of and for the three months ended March 31, 2003 and 2002, the components of comprehensive income were due to the interest rate swap related to our variable rate revolving credit facility, which is designated as a cash flow hedge. Changes in the fair value of the cash flow hedge, to the extent the hedge is effective, are recognized in other comprehensive income until the hedge interest costs are recognized in earnings. The table below reconciles reported net income to total comprehensive income for the three months ended March 31, 2003 and 2002 (in thousands).

	Three Months Er	nded March 31,
	2003	
Net income	\$33,925	\$26,808
Net income on cash flow hedges	5,469	3,300
Total comprehensive income	\$39,394	\$30,108

The accumulated balance of other comprehensive loss related to our cash flow hedge is as follows (in thousands):

Balance at December 31, 2001	\$(20,324)
Net income on cash flow hedges	269
Balance at December 31, 2002	\$(20,055)
Net income on cash flow hedges	5,469
Balance at March 31, 2003	\$(14,586)

NOTE 14. SUPPLEMENTAL CONDENSED CONSOLIDATING FINANCIAL INFORMATION

In connection with our issuance of Senior Notes on February 20, 2002, TE Products Pipeline Company, Limited Partnership, TCTM, L.P., TEPPCO Midstream Companies, L.P. and Jonah Gas Gathering Company, our significant operating subsidiaries, issued unconditional guarantees of our debt securities. Effective with the acquisition of the Val Verde assets on June 30, 2002, our subsidiary, Val Verde Gas Gathering Company, L.P. also became a significant operating subsidiary and issued unconditional guarantees of our debt securities. The guarantees are full, unconditional, and joint and several. TE Products Pipeline Company, Limited Partnership, TCTM, L.P., TEPPCO Midstream Companies, L.P., Jonah Gas Gathering Company and Val Verde Gas Gathering Company, L.P. are collectively referred to as the "Guarantor Subsidiaries." The Guarantor Subsidiaries have also issued guarantees of our 6.125% Senior Notes issued in January 2003.

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS – (Continued) (Unaudited)

The following supplemental condensed consolidating financial information reflects our separate accounts, the combined accounts of the Guarantor Subsidiaries, the combined accounts of our other non-guarantor subsidiaries, the combined consolidating adjustments and eliminations and our consolidated accounts for the dates and periods indicated. For purposes of the following consolidating information, our investments in our subsidiaries and the Guarantor Subsidiaries' investments in their subsidiaries are accounted for under the equity method of accounting.

	March 31, 2003				
	TEPPCO Partners, L.P.	Guarantor Subsidiaries	Non-Guarantor Subsidiaries	Consolidating Adjustments	TEPPCO Partners, L.P. Consolidated
			(in thousands)		-
Assets					
Current assets	\$ 12,978	\$ 98,008	\$ 375,552	\$ (56,192)	\$ 430,346
Property, plant and equipment – net	—	1,054,896	457,009	_	1,511,905
Equity investments	1,001,068	928,127	216,278	(1,767,664)	377,809
Intercompany notes receivable	998,031	—	—	(998,031)	—
Intangible assets	—	424,093	29,804	—	453,897
Other assets	6,553	29,494	35,441	—	71,488
Total assets	\$2,018,630	\$2,534,618	\$1,114,084	\$(2,821,887)	\$2,845,445
iabilities and partners' capital					
Current liabilities	\$ 25,043	\$ 91,296	\$ 371,822	\$ (53,857)	\$ 434,304
Long-term debt	1,005,011	402,152			1,407,163
Intercompany notes payable		568,466	431,898	(1,000,364)	
Other long term liabilities	248	15,443	209		15,900
Redeemable Class B Units held by		,			,
related party	102,693	_	_	_	102,693
Total partners' capital	885,635	1,457,261	310,155	(1,767,666)	885,385
1 1					
Total liabilities and partners'					
capital	\$2,018,630	\$2,534,618	\$1,114,084	\$(2,821,887)	\$2,845,445
<u>r</u>			• • • • • •	· ())	. ,,
		20			

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS – (Continued) (Unaudited)

	December 31, 2002					
	TEPPCO Partners, L.P.	Guarantor Subsidiaries	Non-Guarantor Subsidiaries	Consolidating Adjustments	TEPPCO Partners, L.P. Consolidated	
			(in thousands)			
Assets						
Current assets	\$ 241	\$ 92,798	\$ 286,379	\$ (18,851)	\$ 360,567	
Property, plant and equipment – net	_	1,128,803	459,021	_	1,587,824	
Equity investments	1,011,935	846,991	211,229	(1,785,450)	284,705	
Intercompany notes receivable	986,852	_	—	(986,852)	—	
Intangible assets	—	434,941	30,433	—	465,374	
Other assets	6,200	31,135	34,837		72,172	
Total assets	\$2,005,228	\$2,534,668	\$1,021,899	\$(2,791,153)	\$2,770,642	
Liabilities and partners' capital	• • • • • •		* • • • • • •	A (70.000)	* • • • • • •	
Current liabilities	\$ 30,715	\$ 123,169	\$ 272,538	\$ (59,639)	\$ 366,783	
Long-term debt	974,264	403,428	_	—	1,377,692	
Intercompany notes payable	_	508,652	437,411	(946,063)		
Other long term liabilities	6,523	24,230	209	—	30,962	
Redeemable Class B Units held by						
related party	103,363	_	_	—	103,363	
Total partners' capital	890,363	1,475,189	311,741	(1,785,451)	891,842	
Total liabilities and partners'						
capital	\$2,005,228	\$2,534,668	\$1,021,899	\$(2,791,153)	\$2,770,642	

Three Months Ended March 31, 2003

	TEPPCO Partners, L.P.	Guarantor Subsidiaries	Non-Guarantor Subsidiaries 	Consolidating Adjustments	TEPPCO Partners, L.P. Consolidated
Operating revenues	\$ —	\$102,079	\$998,170	\$ (1,010)	\$1,099,239
Costs and expenses	—	60,877	988,030	(1,010)	1,047,897
Operating income	—	41,202	10,140	—	51,342
Interest expense – net	(17,409)	(13,519)	(7,828)	17,447	(21,309)
Equity earnings	33,925	8,916	4,969	(44,100)	3,710
Other income – net	17,409	(11)	231	(17,447)	182
Net income	\$ 33,925	\$ 36,588	\$ 7,512	\$(44,100)	\$ 33,925

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS – (Continued) (Unaudited)

		Three Months Ended March 31, 2002			
	TEPPCO Partners, L.P.	Guarantor Subsidiaries	Non-Guarantor Subsidiaries	Consolidating Adjustments	TEPPCO Partners, L.P. Consolidated
			(in thousands)		
Operating revenues	\$ —	\$69,788	\$562,056	\$ (707)	\$631,137
Costs and expenses	—	43,024	551,234	(707)	593,551
Operating income	_	26,764	10,822	_	37,586
Interest expense – net	(11,433)	(7,650)	(7,028)	11,433	(14,678)
Equity earnings	26,808	8,108	4,368	(35,712)	3,572
Other income – net	11,433	133	195	(11,433)	328
Net income	\$ 26,808	\$27,355	\$ 8,357	\$(35,712)	\$ 26,808
		Three Months Ended March 31, 2003			

	TEPPCO Partners, L.P.	Guarantor Subsidiaries	Non-Guarantor Subsidiaries	Consolidating Adjustments	TEPPCO Partners, L.P. Consolidated
			(in thousands)		
Cash flows from operating activities					
Net income	\$ 33,925	\$ 36,588	\$ 7,512	\$(44,100)	\$ 33,925
Adjustments to reconcile net income to net cash provided by operating activities:					
Depreciation and amortization	_	21,864	5,449		27,313
Equity earnings, net of distributions	12,608	12,788	(5,038)	(24,137)	(3,779)
Changes in assets and liabilities and other	(30,269)	(12,829)	(4,309)	25,602	(21,805)
Net cash provided by operating activities	16,264	58,411	3,614	(42,635)	35,654
Cash flows from investing activities	(13)	(33,605)	(2,829)	13	(36,434)
Cash flows from financing activities	(16,251)	(27,997)	(14,623)	42,622	(16,249)
Net decrease in cash and cash equivalents	_	(3,191)	(13,838)	_	(17,029)
Cash and cash equivalents at beginning of period	_	8,247	22,721	_	30,968
Cash and cash equivalents at end of period	\$ —	\$ 5,056	\$ 8,883	\$ —	\$ 13,939
	_		_		_

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS – (Continued) (Unaudited)

	Three Months Ended March 31, 2002				
	TEPPCO Partners, L.P.	Guarantor Subsidiaries	Non-Guarantor Subsidiaries	Consolidating Adjustments	TEPPCO Partners, L.P. Consolidated
			(in thousands)		
Cash flows from operating activities					
Net income	\$ 26,808	\$ 27,355	\$ 8,357	\$ (35,712)	\$ 26,808
Adjustments to reconcile net income to net cash provided by (used in) operating activities:					
Depreciation and amortization	—	12,199	3,842	_	16,041
Equity earnings, net of distributions	6,737	2,512	2,607	(8,454)	3,402
Changes in assets and liabilities and					
other	(124,549)	(4,725)	(19,847)	126,471	(22,650)
Net cash provided by (used in) operating activities	(91,004)	37,341	(5,041)	82,305	23,601
Cash flows from investing activities	(58,483)	(40,052)	(135,598)	58,483	(175,650)
Cash flows from financing activities	149,487	6,085	134,703	(140,788)	149,487
Net increase (decrease) in cash and cash					
equivalents	_	3,374	(5,936)	_	(2,562)
Cash and cash equivalents at beginning of period	_	3,655	21,824	_	25,479
Cash and cash equivalents at end of period	\$ —	\$ 7,029	\$ 15,888	\$ —	\$ 22,917

NOTE 15. SUBSEQUENT EVENT

On April 2, 2003, we sold 3,938,750 Units in an underwritten public offering at \$30.35 per Unit. The net proceeds from the offering totaled approximately \$114.5 million, of which approximately \$113.8 million was used to repurchase and retire all of the 3,916,547 outstanding Class B Units held by Duke Energy Transport and Trading Company, LLC, an affiliate of Duke Energy Corporation. We received approximately \$0.7 million in proceeds from the offering in excess of the amount needed to repurchase and retire the Class B Units.

Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations

General

You should read the following review of our financial position and results of operations in conjunction with the Consolidated Financial Statements. Material period-toperiod variances in the consolidated statements of income are discussed under "Results of Operations." The "Financial Condition and Liquidity" section analyzes cash flows and financial position. "Other Considerations" addresses trends, future plans and contingencies that are reasonably likely to materially affect future liquidity or earnings. The Consolidated Financial Statements should be read in conjunction with the financial statements and related notes, together with our discussion and analysis of financial position and results of operations included in our Annual Report on Form 10-K for the year ended December 31, 2002.

We operate and report in three business segments:

- Downstream Segment transportation and storage of refined products, LPGs and petrochemicals;
- Upstream Segment gathering, transportation, marketing and storage of crude oil and distribution of lubrication oils and specialty chemicals; and
- Midstream Segment gathering of natural gas, fractionation of NGLs and transportation of NGLs.

Our reportable segments offer different products and services and are managed separately because each requires different business strategies. TEPPCO GP, Inc., our wholly owned subsidiary, acts as managing general partner of our Operating Partnerships, with a 0.001% general partner interest and manages our subsidiaries.

Our Downstream Segment revenues are earned from transportation and storage of refined products and LPGs, intrastate transportation of petrochemicals, sale of product inventory and other ancillary services. The two largest operating expense items of the Downstream Segment are labor and electric power. We generally realize higher revenues during the first and fourth quarters of each year since our operations are somewhat seasonal. Refined products volumes are generally higher during the second and third quarters because of greater demand for gasolines during the spring and summer driving seasons. LPGs volumes are generally higher from November through March due to higher demand in the Northeast for propane, a major fuel for residential heating. Our Downstream Segment also includes the results of operations of the northern portion of the Dean pipeline. Beginning in January 2003, the northern portion of the Dean pipeline was converted to transport refinery grade propylene ("RGPs") from Mont Belvieu to Point Comfort, Texas. As a result, the revenues and expenses of the northern portion of the Dean pipeline are included in the Downstream Segment. Our Downstream Segment also includes our equity investments in Centennial Pipeline LLC ("Centennial") and Mont Belvieu Storage Partners, L.P. ("MB Storage") (see Note 8. Equity Investments).

Our Upstream Segment revenues are earned from gathering, transportation, marketing and storage of crude oil and distribution of lubrication oils and specialty chemicals, principally in Oklahoma, Texas and the Rocky Mountain region. Marketing operations consist primarily of aggregating purchased crude oil along our pipeline systems, or from third party pipeline systems, and arranging the necessary logistics for the ultimate sale of the crude oil to local refineries, marketers or other end users. Our Upstream Segment also includes the equity earnings from our investment in Seaway Crude Pipeline Company ("Seaway"). Seaway consists of large diameter pipelines that transport crude oil from Seaway's marine terminals on the U.S. Gulf Coast to Cushing, Oklahoma, a central crude oil distribution point for the Central United States, and to refineries in the Texas City and Houston areas.

Our Midstream Segment revenues are earned from the fractionation of NGLs in Colorado, transportation of NGLs from two trunkline NGL pipelines in South Texas, two NGL pipelines in East Texas and a pipeline system ("Chaparral") from West Texas and New Mexico to Mont Belvieu; the gathering of natural gas in the Green River Basin in southwestern Wyoming, through Jonah Gas Gathering Company ("Jonah") and the gathering of CBM from

the Fruitland Coal Formation of the San Juan Basin in New Mexico and Colorado, through Val Verde Gas Gathering Company ("Val Verde"). Chaparral was acquired on March 1, 2002, for \$132.4 million from Diamond-Koch II, L.P. and Diamond-Koch III, L.P., including acquisition related costs of approximately \$0.4 million. The Val Verde assets were acquired on June 30, 2002, for \$444.2 million from a subsidiary of Burlington Resources Inc., including acquisition related costs of approximately \$1.2 million. The Val Verde, Jonah and Chaparral assets are managed and operated by DEFS under contractual agreements. The results of operations of the Chaparral and Val Verde acquisitions are included in periods subsequent to their respective acquisition dates (see Note 6. Acquisitions).

Results of Operations

The following table summarizes financial data by business segment (in thousands):

	Three Months Ended March 31,		
	2003	2002	
Operating revenues:			
Downstream Segment	\$ 67,965	\$ 59,586	
Upstream Segment	985,380	553,888	
Midstream Segment	46,904	18,370	
Intercompany eliminations	(1,010)	(707)	
Total operating revenues	1,099,239	631,137	
Operating income:			
Downstream Segment	29,269	23,647	
Upstream Segment	3,624	6,207	
Midstream Segment	18,449	7,732	
Total operating income	51,342	37,586	
	······		
Earnings before interest:			
Downstream Segment	28,004	22,975	
Upstream Segment	8,782	10,760	
Midstream Segment	18,486	7,751	
Intercompany eliminations	(38)		
Total earnings before interest	55,234	41,486	
Interest expense	(21,905)	(16,787)	
Interest capitalized	596	2,109	
-			
Net income	\$ 33,925	\$ 26,808	

Below is a detailed analysis of the results of operations, including reasons for changes in results, by each of our operating segments.

Downstream Segment

The following table presents volumes delivered in barrels and average tariff per barrel for the three months ended March 31, 2003 and 2002:

		Three Months Ended March 31,		
	2003	2002	Increase (Decrease)	
	(in thousa	ıds, except tariff inform	mation)	
Volumes Delivered				
Refined products	30,232	25,765	17%	
LPGs	13,700	12,035	14%	
			_	
Total	43,932	37,800	16%	
			-	
Average Tariff per Barrel				
Refined products	\$ 0.89	\$ 0.98	(9%)	
LPGs	2.25	1.94	16%	
Average system tariff per barrel	\$ 1.31	\$ 1.28	2%	
			_	

Our Downstream Segment reported earnings before interest of \$28.0 million for the three months ended March 31, 2003, compared with earnings before interest of \$23.0 million for the three months ended March 31, 2002. Earnings before interest increased \$5.0 million primarily due to an increase of \$8.4 million in operating revenues, partially offset by an increase of \$2.8 million in costs and expenses, increased losses of \$0.5 million from equity investments and a decrease of \$0.1 million in other income – net. We discuss the factors influencing these variances below.

Revenues from refined products transportation increased \$1.8 million for the three months ended March 31, 2003, compared with the three months ended March 31, 2002, due to an overall increase of 17% in the refined products volumes delivered. This increase was primarily due to deliveries of product received into our pipeline from Centennial at Creal Springs, Illinois. Centennial, which commenced refined products deliveries to us in April 2002, has provided our system with an additional source of supply for product originating in the U.S. Gulf Coast area. With this incremental supply source, our previously constrained system has expanded services in markets both south and north of Creal Springs. The 17% increase in our overall refined product delivery was composed of a 17% increase in motor fuel delivery, a 26% increase in distillate volume delivery and a 3% increase in jet fuel delivery from the prior year period. Volume increases were due to increased demand for product supplied by the U.S. Gulf Coast into Midwest markets resulting from cold weather during the first quarter of 2003, which both increased demand and caused higher natural gas prices. As a result of high natural gas prices, utilities use distillates as a substitute for natural gas for their facilities. The refined products average rate per barrel decreased 9% from the prior year period primarily due to the impact of the Midwest origin point for volumes received from Centennial, which resulted in an increase in short-haul volumes transported on our system.

Revenues from LPGs transportation increased \$7.5 million for the three months ended March 31, 2003, compared with the three months ended March 31, 2002, primarily due to increased deliveries of propane in the upper Midwest and Northeast market areas attributable to colder than normal weather during the first quarter of 2003. Butane deliveries also increased in the Midwest as a result of declining future prices for gasoline which resulted in increased demand for butane used in gasoline blending. The LPGs average rate per barrel increased 16% from the prior year period as a result of an increased percentage of long-haul deliveries during the three months ended March 31, 2003.

Effective January 1, 2003, TE Products' 50% ownership interest in MB Storage is accounted for as an equity investment. See discussion regarding changes in equity earnings/losses below. Revenues generated from Mont Belvieu operations decreased \$4.5 million during the three months ended March 31, 2003, compared with the three months ended March 31, 2002, as a result of the formation of MB Storage. The purpose of MB Storage is to expand services to the upper Texas Gulf Coast energy marketplace by increasing pipeline throughput and the mix of products handled through the existing system and establishing new receipt and delivery connections.

Table of Contents

Other operating revenues increased \$3.6 million during the three months ended March 31, 2003, compared with the three months ended March 31, 2002, primarily due to higher margins on product inventory sales, higher propane deliveries at our Providence, Rhode Island import facility and higher refined product rental charges. Other operating revenues also increased \$1.0 million due to the addition of the northern Dean pipeline to the Downstream Segment in January 2003, which began transporting RGPs in January 2003. These increases were partially offset by lower revenues from product location exchanges which are used to position product in the Midwest market area.

Costs and expenses increased \$2.8 million for the three months ended March 31, 2003, compared with the three months ended March 31, 2002. The increase was made up of an increase of \$1.7 million in operating, general and administrative expenses, an increase of \$1.1 million in operating fuel and power and an increase of \$0.3 million increase in depreciation and amortization expense. These increases were partially offset by a decrease of \$0.3 million in taxes – other than income taxes. Operating, general and administrative expenses increased primarily due to higher pipeline maintenance expenses, increased consulting and contract services, increased labor costs and increased general and administrative expenses and the addition of the northern Dean pipeline to the Downstream Segment, which increased operating, general and administrative expense by \$0.2 million. Depreciation expense increased from the prior year period because of assets placed in service during 2002. Operating fuel and power expense increased as a result of increased mainline throughput and higher power costs. Taxes – other than income taxes decreased as a result of actual property taxes being lower than previously estimated.

Net losses from equity investments increased by \$0.5 million during the three months ended March 31, 2003, compared with the three months ended March 31, 2002. Centennial, which commenced operations in April 2002, accounted for \$2.4 million of the equity losses. On February 10, 2003, TE Products acquired an additional 16.7% interest in Centennial, bringing its ownership interest to 50%. The losses from Centennial are partially offset by equity earnings of \$1.9 million from our 50% ownership interest in MB Storage, which was formed effective January 1, 2003. Amounts in the prior year period recorded to revenues and costs and expenses are now being recorded to equity earnings based upon our ownership interest in MB Storage, effective with its formation on January 1, 2003.

Upstream Segment

Information presented in the following table includes the margin of the Upstream Segment, which may be viewed as a non-GAAP (Generally Accepted Accounting Principles) financial measure under the rules of the Securities and Exchange Commission. A reconciliation of margin to operating revenues and operating expenses is provided in the following table. We calculate the margin of the Upstream Segment as revenues generated from the sale of crude oil and lubrication oil, and transportation of crude oil, less the costs of purchases of crude oil and lubrication oil. Margin is a more meaningful measure of financial performance than operating revenues and operating expenses due to the significant fluctuations in revenues and expenses caused by variations in the level of marketing activity and prices for products marketed. Margin and volume information for the three months ended March 31, 2003, and 2002, is presented below (in thousands, except per barrel and per gallon amounts):

	Three Months Ended March 31,		Percentage
	2003	2002	Increase (Decrease)
Margins:			
Crude oil transportation	\$10,656	\$ 9,071	17%
Crude oil marketing	4,883	5,003	(2%)
Crude oil terminaling	2,120	2,449	(13%)
Lubrication oil sales	1,352	1,135	19%
Total margin	\$19,011	\$17,658	8%
			_
Total barrels:			
Crude oil transportation	22,625	21,116	7%
Crude oil marketing	37,672	30,352	24%
Crude oil terminaling	27,369	29,275	(7%)
Lubrication oil volume (total gallons)	2,843	2,194	30%
Margin per barrel:			
Crude oil transportation	\$ 0.471	\$ 0.430	10%
Crude oil marketing	0.130	0.165	(21%)
Crude oil terminaling	0.077	0.084	(7%)
Lubrication oil margin (per gallon)	0.476	0.517	(8%)

The following table reconciles the Upstream Segment margin to the consolidated statements of income using the information presented in the consolidated statements of income and the statements of income in Note 11. Segment Data (in thousands):

	Three Months	Three Months Ended March 31,		
	2003	2002		
Sales of petroleum products	\$ 975,960	\$ 545,208		
Transportation – Crude oil	6,905	6,128		
Less: Purchases of petroleum products	(963,854)	(533,678)		
Total margin	\$ 19,011	\$ 17,658		

Our Upstream Segment reported earnings before interest of \$8.8 million for the three months ended March 31, 2003, compared with earnings before interest of \$10.8 million for the three months ended March 31, 2002. Earnings before interest decreased \$2.0 million primarily due to an increase of \$4.0 million in costs and expenses (excluding purchases of crude oil and lubrication oil), partially offset by an increase of \$1.3 million in margin, a decrease of \$0.1 million in other operating revenues and an increase of \$0.6 million in equity earnings of Seaway. We discuss factors influencing these variances below.



Our margin increased \$1.3 million during the three months ended March 31, 2003, compared with the three months ended March 31, 2002. Crude oil transportation margin increased \$1.5 million primarily due to increased revenues on our Red River, Basin, South Texas and West Texas systems. Lubrication oil sales margin increased \$0.2 million due to increased sales of chemical volumes. Crude oil marketing margin decreased \$0.1 million primarily due to a pricing settlement on a marketing contract, partially offset by increased volumes marketed, renegotiated supply contracts and lower trucking expenses. Crude oil terminaling margin decreased \$0.3 million as a result of lower pumpover volumes at Midland, Texas, and Cushing, Oklahoma.

Other operating revenue of the Upstream Segment decreased \$0.1 million for the three months ended March 31, 2003, compared with the three months ended March 31, 2002, due to lower revenues from documentation and other services to support customers' trading activity at Midland and Cushing.

Costs and expenses, excluding expenses associated with purchases of crude oil and lubrication oil, increased \$4.0 million during the three months ended March 31, 2003, compared with the three months ended March 31, 2002. Operating, general and administrative expenses increased \$3.3 million from the prior year period. The increase includes \$1.6 million of higher environmental remediation costs in 2003 and \$1.7 million from the net settlement of crude oil imbalances with customers. Depreciation and amortization expense increased \$1.0 million due to assets placed in service in 2002. These increases were partially offset by a decrease of \$0.2 million in operating fuel and power attributable to a more efficient use of transportation assets in 2003. Taxes – other than income taxes decreased \$0.1 million due to reductions in property tax accruals.

Equity earnings in Seaway for the three months ended March 31, 2003, increased \$0.6 million from the three months ended March 31, 2002, due to the settlement of crude oil imbalances with customers and lower general and administrative expenses, partially offset by lower long-haul third-party transportation volumes and by our portion of equity earnings decreasing from 80% to 60% on a pro-rated basis in 2002 (averaging approximately 67% for the year ended December 31, 2002), to 60% in 2003.

Midstream Segment

The following table presents volume and average rate information for the three months ended March 31, 2003 and 2002:

		Three Months Ended March 31,	
	2003	2002	Increase (Decrease)
Gathering – Natural Gas:			
Million cubic feet	115,974	50,171	131%
Million British thermal units ("MMBtu")	117,212	55,728	110%
Average fee per MMBtu	\$ 0.29	\$ 0.17	71%
Transportation – NGLs:			
Thousand barrels	14,191	4,606	208%
Average rate per barrel	\$ 0.698	\$ 0.554	26%
Fractionation – NGLs:			
Thousand barrels	1,071	1,012	6%
Average rate per barrel	\$ 1.734	\$ 1.813	(5%)
Sales – Condensate:			
Thousand barrels	30,771	32,352	(5%)
Average rate per barrel	\$ 32.84	\$ 25.38	29%
Average rate per barrel	\$ 32.84	\$ 25.38	

Our Midstream Segment reported earnings before interest of \$18.5 million for the three months ended March 31, 2003, compared with earnings before interest of \$7.8 million for the three months ended March 31, 2002. Earnings before interest increased \$10.7 million due to an increase of \$28.5 million in operating revenues, partially offset by an increase of \$17.8 million in costs and expenses. We discuss factors influencing these variances below.

Operating revenues increased \$28.5 million during the three months ended March 31, 2003, compared with the three months ended March 31, 2002, due to an increase of \$24.8 million in natural gas gathering revenues, an increase of \$3.6 million in NGL transportation revenues and an increase of \$0.1 million in other revenues. Natural gas gathering revenues from the Jonah system increased \$6.1 million and volumes delivered increased 24.1 billion cubic feet during the three months ended March 31, 2003, due to the expansion of the Jonah system during 2002. The expansion of the Jonah system increased its capacity from 450 million cubic feet per day ("MMcf/day") to approximately 880 MMcf/day. Natural gas gathering revenues from the Val Verde system, which was acquired on June 30, 2002, totaled \$18.7 million and volumes delivered totaled 41.7 billion cubic feet during the three months ended March 31, 2003. Other revenues increased \$0.1 million primarily due to sales of gas condensate from the Jonah system. NGL transportation revenues increased \$3.6 million primarily due to the acquisition of the Chaparral NGL system on March 1, 2002.

Costs and expenses increased \$17.8 million during the three months ended March 31, 2003, compared with the three months ended March 31, 2002, due to an increase of \$10.0 million in depreciation and amortization expense, an increase of \$6.2 million in operating, general and administrative expense, an increase of \$0.9 million in taxes – other than income taxes and an increase of \$0.7 million in operating fuel and power. Depreciation and amortization expense increased \$7.4 million due to the acquisition of the Chaparral and Val Verde assets acquired on March 1, 2002, and June 30, 2002, respectively, and \$2.6 million due to assets placed in service in 2002 related to the expansion of the Jonah system. Operating, general and administrative expense increased \$4.7 million from the assets acquired and due to higher general and administrative labor and supplies expense. Operating fuel and power costs increased \$0.7 million due to the assets acquired. Taxes – other than income taxes increased \$0.7 million due to the assets acquired and \$0.2 million due to a higher property tax base on Jonah as a result of the expansion.

Interest Expense and Capitalized Interest

Interest expense increased \$5.1 million during the three months ended March 31, 2003, compared with the three months ended March 31, 2002, primarily due to higher outstanding debt balances used for capital expenditures and to finance the acquisition of assets acquired through the Midstream Segment, partially offset by lower LIBOR interest rates in effect during the three months ended March 31, 2003.

Capitalized interest decreased \$1.5 million during the three months ended March 31, 2003, compared with the three months ended March 31, 2002, due to interest capitalized on expenditures during the construction of the Jonah expansion in 2002 and decreased balances during 2003 on construction work-in-progress.

Financial Condition and Liquidity

Net cash from operating activities totaled \$35.7 million for the three months ended March 31, 2003. This cash was made up of \$61.2 million of income before charges for depreciation and amortization, partially offset by \$25.5 million of cash used for working capital. This compares with net cash from operating activities of \$23.6 million for the corresponding period in 2002, comprised of \$42.8 million of income before charges for depreciation and amortization, partially offset by \$19.2 million of cash used for working capital. Net cash from operatings for the three months ended March 31, 2003 and 2002, included interest payments of \$32.5 million and \$16.8 million, respectively.

Cash flows used in investing activities totaled \$36.4 million during the three months ended March 31, 2003, and were comprised of \$15.4 million of capital expenditures, \$20.0 million for TE Products' acquisition of an

additional ownership interest in Centennial and \$1.0 million of cash contributions for our ownership interest in Centennial. Cash flows used in investing activities totaled \$175.6 million during the three months ended March 31, 2002, and were primarily comprised of \$7.3 million for the final purchase price adjustments on the acquisition of Jonah, \$33.0 million of capital expenditures, \$3.3 million of cash contributions for our ownership interest in Centennial and \$132.0 million for the purchase of Chaparral on March 1, 2002.

Cash flows used in financing activities totaled \$16.2 million during the three months ended March 31, 2003, and were comprised of \$40.0 million in proceeds from term and revolving credit facilities; and \$198.6 million from the issuance in January 2003 of our 6.125% Senior Notes due 2013, partially offset by debt issuance costs of \$1.3 million. These sources of cash during the three months ended March 31, 2003, were partially offset by \$207.0 million of repayments on our term and revolving credit facilities and \$46.5 million of distributions to unitholders. Cash flows provided by financing activities totaled \$149.5 million during the three months ended March 31, 2002, and were comprised of \$172.0 million of proceeds from term and revolving credit facilities; \$497.8 million from the issuance in February 2002 of our 7.625% Senior Notes due 2012, partially offset by debt issuance costs of \$4.1 million; and \$56.8 million from the issuance of 1.92 million Units and \$1.2 million of repayments on our term and revolving credit facilities and \$33.5 million of distributions to unitholders.

In August 2000, TE Products entered into agreements with Panhandle Eastern Pipeline Company ("PEPL") and Marathon Ashland Petroleum LLC ("Marathon") to form Centennial. Centennial owns an interstate refined petroleum products pipeline extending from the upper Texas Gulf Coast to Illinois. Through February 9, 2003, each original participant owned a one-third interest in Centennial. On February 10, 2003, TE Products and Marathon each acquired an additional 16.7% interest in Centennial, bringing their ownership interest to 50% each. Excluding TE Products' purchase of its additional ownership interest of 16.7% on February 10, 2003, we expect to contribute an additional \$10.0 million to Centennial in 2003.

Centennial entered into credit facilities totaling \$150.0 million, and as of March 31, 2003, \$150.0 million was outstanding under those credit facilities. The proceeds were used to fund construction and conversion costs of its pipeline system. Each of the participants in Centennial, including TE Products, originally guaranteed one-third of Centennial's debt, up to a maximum amount of \$50.0 million. During the third quarter of 2002, PEPL, one of the participants in Centennial, was downgraded by Moody's and Standard & Poors to below investment grade, which resulted in PEPL being in default under its portion of the Centennial guaranty. Effective September 27, 2002, TE Products and Marathon increased their guaranteed amounts to one-half of the debt of Centennial, up to a maximum amount of \$75.0 million each, to avoid a default on the Centennial debt. As compensation to TE Products and Marathon for providing their additional guarantees, PEPL was required to pay interest at a rate of 4% per annum to each of TE Products and Marathon on the portion of the additional guaranty that each had provided for PEPL. In connection with the acquisition of the additional interest in Centennial on February 10, 2003, the guaranty agreement between TE Products, Marathon and PEPL was terminated. TE Products' guaranty of up to a maximum of \$75.0 million of Centennial's debt remains in effect.

Credit Facilities and Interest Rate Swap Agreements

On April 6, 2001, we entered into a \$500.0 million revolving credit facility including the issuance of letters of credit of up to \$20.0 million ("Three Year Facility"). The interest rate is based, at our option, on either the lender's base rate plus a spread, or LIBOR plus a spread in effect at the time of the borrowings. The credit agreement for the Three Year Facility contains certain restrictive financial covenant ratios. During 2002, borrowings under the Three Year Facility were used to finance the acquisitions of the Chaparral NGL system on March 1, 2002, and Val Verde on June 30, 2002. During 2002, repayments were made on the Three Year Facility with proceeds from the issuance of our 7.625% Senior Notes, proceeds from the issuance of additional Units and proceeds from the termination of interest rate swaps (see Note 5. Derivative Financial Instruments). During the first quarter of 2003, we repaid \$182.0 million of the outstanding balance of the Three Year Facility with proceeds from the issuance of our 6.125% Senior Notes on January 30, 2003. At March 31, 2003, \$265.0 million was outstanding under the Three

Year Facility at a weighted average interest rate of 2.3%. As of March 31, 2003, we were in compliance with the covenants contained in this credit agreement.

On April 6, 2001, we entered into a 364-day, \$200.0 million revolving credit agreement ("Short-term Revolver"). The interest rate was based, at our option, on either the lender's base rate plus a spread, or LIBOR plus a spread in effect at the time of the borrowings. The credit agreement contained certain restrictive financial covenant ratios. On March 28, 2002, the Short-term Revolver was extended for an additional period of 364 days, ending in March 2003. During 2002, borrowings under the Short-term Revolver were used to finance the acquisition of the Val Verde assets and for other purposes. During 2002, we repaid the existing amounts outstanding under the Short-term Revolver with proceeds we received from the issuance of Units in 2002. The Short-term Revolver expired on March 27, 2003.

On February 20, 2002, we completed the issuance of \$500.0 million principal amount of 7.625% Senior Notes due 2012. The 7.625% Senior Notes were issued at a discount of \$2.2 million and are being accreted to their face value over the term of the notes. We used the proceeds from the offering to reduce a portion of the outstanding balances of our credit facilities, including those issued in connection with the acquisition of Jonah. The Senior Notes may be redeemed at any time at our option with the payment of accrued interest and a make-whole premium determined by discounting remaining interest and principal payments using a discount rate equal to the rate of the United States Treasury securities of comparable remaining maturity plus 35 basis points. The indenture governing the 7.625% Senior Notes contains covenants, including, but not limited to, covenants limiting the creation of liens securing indebtedness and sale and leaseback transactions. However, the indenture does not limit our ability to incur additional indebtedness. As of March 31, 2003, we were in compliance with the covenants of these Senior Notes.

On June 27, 2002, we entered into a \$200.0 million six-month term loan with SunTrust Bank ("Six-Month Term Loan") payable in December 2002. We borrowed \$200.0 million under the Six-Month Term Loan to acquire the Val Verde assets (see Note 6. Acquisitions). The interest rate was based, at our option, on either the lender's base rate plus a spread, or LIBOR plus a spread in effect at the time of the borrowings. The credit agreement contained certain restrictive financial covenant ratios. On July 11, 2002, we repaid \$90.0 million of the outstanding principal from proceeds primarily received from the issuance of Units in July 2002. On September 10, 2002, we repaid the remaining outstanding balance of \$110.0 million with proceeds received from the issuance of Units in September 2002, and canceled the facility.

On January 30, 2003, we completed the issuance of \$200.0 million principal amount of 6.125% Senior Notes due 2013. The 6.125% Senior Notes were issued at a discount of \$1.4 million and are being accreted to their face value over the term of the notes. We used \$182.0 million of the proceeds from the offering to reduce the outstanding principal on the Three Year Facility to \$250.0 million. The balance of the net proceeds received was used for general purposes. The Senior Notes may be redeemed at any time at our option with the payment of accrued interest and a make-whole premium determined by discounting remaining interest and principal payments using a discount rate equal to the rate of the United States Treasury securities of comparable remaining maturity plus 35 basis points. The indenture governing our 6.125% Senior Notes contains covenants, including, but not limited to, covenants limiting the creation of liens securing indebtedness and sale and leaseback transactions. However, the indenture does not limit our ability to incur additional indebtedness. As of March 31, 2003, we were in compliance with the covenants of these Senior Notes.

We have entered into interest rate swap agreements to hedge our exposure to cash flows and fair value changes. These agreements are more fully described in Item 3. "Quantitative and Qualitative Disclosures About Market Risk."

The following table summarizes our credit facilities as of March 31, 2003 (in millions):

		As of March 31, 2003		
Description:	Outstanding Principal	Available Borrowing Capacity	Maturity Date	
Three Year Facility	\$ 265.0	\$235.0	April 2004	
6.45% Senior Notes (1)	180.0	_	January 2008	
7.51% Senior Notes (1)	210.0	_	January 2028	
7.625% Senior Notes (1)	500.0	_	February 2012	
6.125% Senior Notes (1)	200.0	_	February 2013	
Total	\$1,355.0	\$235.0		

(1) Our TE Products subsidiary entered into an interest rate swap agreement to hedge its exposure to changes in the fair value of its 7.51% Senior Notes due 2028. At March 31, 2003, the 7.51% Senior Notes include an adjustment to increase the fair value of the debt by \$12.3 million related to this interest rate swap agreement. We also entered into interest rate swap agreements to hedge our exposure to changes in the fair value of our 7.625% Senior Notes due 2012. At March 31, 2003, the 7.625% Senior Notes include a deferred gain, net of amortization, from previous interest rate swap terminations of \$43.4 million. At March 31, 2003, our 6.45% Senior Notes, our 7.625% Senior Notes and our 6.125% Senior Notes include \$3.5 million of unamortized debt discounts. The fair value adjustments, the deferred gain adjustment and the uanmortized debt discounts are excluded from this table.

Distributions and Issuance of Additional Limited Partner Units

We paid cash distributions of \$46.5 million (\$0.60 per Unit) and \$33.5 million (\$0.575 per Unit) for each of the three months ended March 31, 2003 and 2002, respectively. Additionally, we declared a cash distribution of \$0.625 per Unit for the quarter ended March 31, 2003. We will pay the distribution of \$49.4 million on May 9, 2003, to unitholders of record on April 30, 2003.

On March 22, 2002, we sold in an underwritten public offering 1.92 million Units at \$31.18 per Unit. The proceeds from the offering, net of underwriting discount, totaled approximately \$57.3 million and were used to repay \$50.0 million of the outstanding balance on the Three Year Facility, with the remaining amount being used for general purposes.

On July 11, 2002, we sold in an underwritten public offering 3.0 million Units at \$30.15 per Unit. The proceeds from the offering, net of underwriting discount, totaled approximately \$86.6 million and were used to reduce borrowings under our Six-Month Term Loan. On August 14, 2002, 175,000 Units were sold upon exercise of the underwriters' over-allotment option granted in connection with the offering on July 11, 2002. Proceeds from that sale totaled \$5.1 million and were used for general purposes.

On September 5, 2002, we sold in an underwritten public offering 3.8 million Units at \$29.72 per Unit. The proceeds from the offering, net of underwriting discount, totaled approximately \$108.1 million and were used to reduce borrowings under our Six-Month Term Loan. On September 19, 2002, 570,000 Units were sold upon exercise of the underwriters' over-allotment option granted in connection with the offering on September 5, 2002. Proceeds from that sale totaled \$16.2 million and were used to reduce borrowings under our Six-Month Term Loan.

On November 7, 2002, we sold in an underwritten public offering 3.3 million Units at \$26.83 per Unit. The proceeds from the offering, net of underwriting discount, totaled approximately \$84.8 million and were used to reduce borrowings under our Short-term Revolver and Three Year Facility. On December 4, 2002, 495,000 Units were sold upon exercise of the underwriters' over-allotment option granted in connection with the offering on

November 7, 2002. Proceeds from that sale totaled \$12.7 million and were used to reduce borrowings under our Short-term Revolver and Three Year Facility.

Future Capital Needs and Commitments

We estimate that capital expenditures, excluding acquisitions, for 2003 will be approximately \$64.1 million (which includes \$3.6 million of capitalized interest). We expect to use approximately \$22.2 million for revenue generating projects. Capital spending on revenue generating projects will include approximately \$7.5 million for the expansion of our pumping capacity of LPGs into the Northeast markets, approximately \$2.3 million for expansion of our Downstream Segment's deliverability capacity, \$5.0 million to expand Upstream Segment facilities and approximately \$7.4 million for the expansion of Midstream assets. We expect to spend approximately \$31.8 million to sustain existing operations, of which approximately \$25.9 million will be for Downstream Segment pipeline projects, including the replacement of storage tanks and pipeline rehabilitation projects to comply with regulations enacted by the United States Department of Transportation Office of Pipeline Safety, \$4.8 million for upgrades of our Upstream Segment and \$1.1 million of capital expenditures to sustain existing operations on the Midstream Segment. An additional \$10.1 million will be expended on system upgrade projects among all of our business segments including the replacement of a portion of our 20 inch pipeline crossing the Mississippi River for approximately \$4.0 million. We continually review and evaluate potential capital improvements and expansions that would be complementary to our present business segments. These expenditures can vary greatly depending on the magnitude of our transactions. We may finance capital expenditures through internally generated funds, debt or the issuance of additional equity.

As of March 31, 2003, we had a working capital deficit of \$4.0 million. In the event of any working capital shortfalls, we have approximately \$235.0 million in available borrowing capacity under our Three Year Facility to cover these items.

Our debt repayment obligations consist of payments for principal and interest on (i) outstanding principal amounts under the Three Year Facility due in April 2004 (\$265.0 million outstanding at March 31, 2003), (ii) the TE Products \$180.0 million 6.45% Senior Notes due January 15, 2008, (iii) the TE Products \$210.0 million 7.51% Senior Notes due January 15, 2028, (iv) our \$500.0 million 7.625% Senior Notes due February 15, 2012, and (v) our \$200.0 million 6.125% Senior Notes due February 1, 2013.

TE Products is contingently liable as guarantor for the lesser of one-half or \$75.0 million principal amount (plus interest) of the borrowings of Centennial. We expect to contribute an additional \$10.0 million to Centennial in 2003 to provide for its working capital needs. In January 2003, TE Products entered into a pipeline capacity lease agreement with Centennial for a period of five years. On February 10, 2003, TE Products acquired an additional 16.7% ownership interest in Centennial, bringing its ownership percentage to 50%.

We do not rely on off-balance sheet borrowings to fund our acquisitions. We have no off-balance sheet commitments for indebtedness other than the limited guaranty of Centennial debt and leases covering assets utilized in several areas of our operations.

The following table summarizes our debt repayment obligations and material contractual commitments as of March 31, 2003 (in millions):

		Amount of Commitment Expiration Per Period			
	Total	Less than 1 Year	1-3 Years	4-5 Years	After 5 Years
Three Year Facility	\$ 265.0	\$ —	\$265.0	\$ —	\$ —
6.45% Senior Notes due 2008 (1) (2)	180.0		_	180.0	
7.51% Senior Notes due 2028 (1) (2)	210.0	_	_	_	210.0
7.625% Senior Notes due 2012 (2)	500.0		_		500.0
6.125% Senior Notes due 2013 (2)	200.0	_	_	_	200.0
Debt subtotal	1,355.0	_	265.0	180.0	910.0
Centennial cash contributions	10.0	10.0	_	_	
Operating leases	33.1	9.4	16.6	6.3	0.8
Contractual commitments subtotal	43.1	19.4	16.6	6.3	0.8
Total	\$1,398.1	\$19.4	\$281.6	\$186.3	\$910.8

Amount of Commitment Expiration Per Period

(1) Obligations of TE Products.

(2) Our TE Products subsidiary entered into an interest rate swap agreement to hedge its exposure to changes in the fair value of its 7.51% Senior Notes due 2028. At March 31, 2003, the 7.51% Senior Notes include an adjustment to increase the fair value of the debt by \$12.3 million related to this interest rate swap agreement. We also entered into interest rate swap agreements to hedge our exposure to changes in the fair value of our 7.625% Senior Notes due 2012. At March 31, 2003, the 7.625% Senior Notes include a deferred gain, net of amortization, from previous interest rate swap terminations of \$43.4 million. At March 31, 2003, our 6.45% Senior Notes, our 7.625% Senior Notes and our 6.125% Senior Notes include \$3.5 million of unamortized debt discounts. The fair value adjustments, the deferred gain adjustments and the unamortized debt discounts are excluded from this table.

We expect to repay the long-term, senior unsecured obligations and bank debt through the issuance of additional long-term senior unsecured debt at the time the 2008, 2012, 2013 and 2028 debt matures, issuance of additional equity, proceeds from dispositions of assets, cash flow from operations or any combination of the above items.

Sources of Future Capital

Historically, we have funded our capital commitments from operating cash flow and borrowings under bank credit facilities or bridge loans. We repaid these loans in part by the issuance of long term debt in capital markets and the public offering of Units. We expect future capital needs would be similarly funded to the extent not otherwise available from cash flow from operations.

As of March 31, 2003, we had approximately \$235.0 million in available borrowing capacity under the Three Year Facility.

We expect that cash flows from operating activities will be adequate to fund cash distributions and capital additions necessary to sustain existing operations. However, future expansionary capital projects and acquisitions may require funding through proceeds from the sale of additional debt or equity capital markets offerings.

On May 29, 2002, Moody's Investors Service downgraded our senior unsecured debt rating to Baa3 from Baa2. Our subsidiary, TE Products was also included in this downgrade. These ratings were given with stable outlooks and followed our announcement of the acquisition of Val Verde. The downgrades reflect Moody's concern that we have a high level of debt relative to many of our peers and that our debt may be continually higher than our long-term targets if we continue to make a series of acquisitions of increasingly larger size. Because of our high distribution rate, we are particularly reliant on external financing to finance our acquisitions. Moody's indicated that our cash flows are becoming less predictable as a result of our acquisitions and expansion into the crude oil and natural gas gathering businesses. Further reductions in our credit ratings could increase the debt financing costs or possibly reduce the availability of financing. A rating reflects only the view of a rating agency and is not a recommendation to buy, sell or hold any indebtedness. Any rating can be revised upward or downward or withdrawn at any time by a rating agency if it determines that the circumstances warrant such a change. In February 2003, Moody's reaffirmed the Baa3 ratings for us and our subsidiary, TE Products.

Other Considerations

Our operations are subject to federal, state and local laws and regulations governing the discharge of materials into the environment. Failure to comply with these laws and regulations may result in the assessment of administrative, civil and criminal penalties, imposition of injunctions delaying or prohibiting certain activities and the need to perform investigatory and remedial activities. Although we believe our operations are in material compliance with applicable environmental laws and regulations, risks of significant costs and liabilities are inherent in pipeline operations, and we cannot assure you that significant costs and liabilities will not be incurred. Moreover, it is possible that other developments, such as increasingly strict environmental laws and regulations and enforcement policies thereunder, and claims for damages to property or persons resulting from our operations, could result in substantial costs and liabilities to us. We believe that changes in environmental laws and regulations will not have a material adverse effect on our financial position, results of operations or cash flows in the near term.

In 1994, we entered into an Agreed Order with the Indiana Department of Environmental Management ("IDEM") that resulted in the implementation of a remediation program for groundwater contamination attributable to our operations at the Seymour, Indiana, terminal. In 1999, the IDEM approved a Feasibility Study, which includes our proposed remediation program. In March 2003, the IDEM issued a Record of Decision formally approving the remediation program. As the Record of Decision has been issued, we will enter into an Agreed Order for the continued operation and maintenance of the remediation program. We have an accrued liability of \$0.3 million at March 31, 2003, for future remediation costs at the Seymour terminal. We do not expect that the completion of the remediation program will have a future material adverse effect on our financial position, results of operations or cash flows.

In 1994, the Louisiana Department of Environmental Quality ("LDEQ") issued a compliance order for environmental contamination at our Arcadia, Louisiana, facility. This contamination may be attributable to our operations, as well as adjacent petroleum terminals operated by other companies. In 1999, our Arcadia facility and adjacent terminals were directed by the Remediation Services Division of the LDEQ to pursue remediation of this containment phase. At March 31, 2003, we have an accrued liability of \$0.2 million for remediation costs at our Arcadia facility. We do not expect that the completion of the remediation program that we have proposed will have a future material adverse effect on our financial position, results of operations or cash flows.

On March 17, 2003, we experienced a release of 511 barrels of jet fuel from a tank at our Blue Island terminal located in Cook County, Illinois. As a result of the release, we have entered into an Agreed Preliminary Injunction and Order ("Agreed Order") with the State of Illinois. The Agreed Order requires us, in part, to complete a site investigation plan to delineate the scope of any potential contamination resulting from the release and to remediate any contamination. The Agreed Order does not contain any provision for any fines or penalties; however, it does not preclude the State of Illinois from assessing these at a later date. We do not expect that the completion of the remediation program will have a future material adverse effect on our financial position, results of operations or cash flows.

At March 31, 2003, we have an accrued liability of \$6.6 million related to various TCTM sites requiring environmental remediation activities. We also have a receivable at March 31, 2003, of \$1.9 million from DEFS which is based on a contractual indemnity obligation we received in connection with our acquisition of assets from a DEFS affiliate in November 1998. The indemnity relates to future environmental remediation activities attributable to operations of these assets prior to our acquisition. Under this indemnity obligation, we are responsible for the first \$3.0 million in specified environmental liabilities, and DEFS is responsible for those environmental liabilities in excess of \$3.0 million, up to a maximum amount of \$25.0 million. At December 31, 2002, the receivable balance from DEFS was \$4.2 million. The majority of the receivable from DEFS related to remediation activities at the Velma crude oil site in Stephens County, Oklahoma. During the first quarter of 2003, we received \$2.4 million from DEFS as partial payment on the receivable balance. The accrued liability balance at March 31, 2003, also includes an accrual of \$2.3 million related to the Shelby crude oil site in Stephens County, Oklahoma. At March 31, 2003, it is uncertain if these costs related to Shelby are covered under the indemnity obligation from DEFS. We are currently in discussions with DEFS regarding these matters. We do not expect that the completion of remediation programs associated with TCTM activities will have a future material adverse effect on our financial position, results of operations or cash flows.

New Accounting Pronouncements

In December 2002, the Financial Accounting Standards Board ("FASB") issued Statement of Financial Accounting Standards ("SFAS") No. 148, Accounting for Stock-Based Compensation – Transition and Disclosure. SFAS 148 amends SFAS No. 123, Accounting for Stock-Based Compensation, to provide alternative methods of transition for a voluntary change to the fair value based method of accounting for stock-based employee compensation. In addition, SFAS 148 amends the disclosure requirements of SFAS 123 to require prominent disclosure in both annual and interim financial statements about the method of accounting for stock-based employee compensation and the effect of the method used on the reported results. The provisions of SFAS 148 are effective for financial statements for fiscal years ending after December 15, 2002. The adoption of SFAS 148 did not affect our financial position, results of operations or cash flows.

In January 2003, the FASB issued FASB Interpretation No. 46, *Consolidation of Variable Interest Entities, an interpretation of ARB No. 51* ("FIN 46"). FIN 46 requires certain variable interest entities to be consolidated by the primary beneficiary of the entity if the equity investors in the entity do not have the characteristics of a controlling financial interest or do not have sufficient equity at risk for the entity to finance its activities without additional subordinated financial support from other parties. We are required to apply FIN 46 to all new variable interest entities created or acquired after January 31, 2003. For variable interest entities created or acquired prior to February 1, 2003, we are required to apply FIN 46 on July 1, 2003. In connection with the adoption of FIN 46, we evaluated our investments in Centennial, Seaway and MB Storage and determined that these entities are not variable interest entities as defined by FIN 46, and thus we have accounted for them as equity method investments (see Note 8. Equity Investments). The adoption of FIN 46 did not have an effect on our financial position, results of operations or cash flows.

Forward-Looking Statements

The matters discussed in this Report include "forward-looking statements" within the meaning of various provisions of the Securities Act of 1933 and the Securities Exchange Act of 1934. All statements, other than statements of historical facts, included in this document that address activities, events or developments that we expect or anticipate will or may occur in the future, including such things as estimated future capital expenditures (including the amount and nature thereof), business strategy and measures to implement strategy, competitive strengths, goals, expansion and growth of our business and operations, plans, references to future success, references to intentions as to future matters and other such matters are forward-looking statements. These statements are based on certain assumptions and analyses made by us in light of our experience and our perception of historical trends, current conditions and expected future developments as well as other factors we believe are appropriate under the circumstances. However, whether actual results and developments will conform with our expectations and

predictions is subject to a number of risks and uncertainties, including general economic, market or business conditions, the opportunities (or lack thereof) that may be presented to and pursued by us, competitive actions by other pipeline companies, changes in laws or regulations and other factors, many of which are beyond our control. Consequently, all of the forward-looking statements made in this document are qualified by these cautionary statements and we cannot assure you that actual results or developments that we anticipate will be realized or, even if substantially realized, will have the expected consequences to or effect on us or our business or operations. For additional discussion of such risks and uncertainties, see our Annual Report on Form 10-K, for the year ended December 31, 2002, and other filings we have made with the Securities and Exchange Commission.

Item 3. Quantitative and Qualitative Disclosures About Market Risk

We may be exposed to market risk through changes in commodity prices and interest rates. We do not have foreign exchange risks. Our Risk Management Committee has established policies to monitor and control these market risks. The Risk Management Committee is comprised, in part, of senior executives of the Company.

At March 31, 2003, we had \$265.0 million outstanding under our variable interest rate revolving credit agreement. The interest rate is based, at our option, on either the lender's base rate plus a spread or LIBOR plus a spread in effect at the time of the borrowings and is adjusted monthly, bimonthly, quarterly or semiannually. Utilizing the balances of variable interest rate debt outstanding at March 31, 2003, including the effects of hedging activities discussed below, and assuming market interest rates increase 100 basis points, the potential annual increase in interest expense is \$0.2 million.

We have utilized and expect to continue to utilize interest rate swap agreements to hedge a portion of our cash flow and fair value risks. Interest rate swap agreements are used to manage the fixed and floating interest rate mix of our total debt portfolio and overall cost of borrowing. The interest rate swap related to our cash flow risk is intended to reduce our exposure to increases in the benchmark interest rates underlying our variable rate revolving credit facility. The interest rate swaps related to our fair value risks are intended to reduce our exposure to changes in the fair value of the fixed rate Senior Notes. The interest rate swap agreements involve the periodic exchange of payments without the exchange of the notional amount upon which the payments are based. The related amount payable to or receivable from counterparties is included as an adjustment to accrued interest.

At March 31, 2003, TE Products had outstanding \$180.0 million principal amount of 6.45% Senior Notes due 2008, and \$210.0 million principal amount of 7.51% Senior Notes due 2028 (collectively the "TE Products Senior Notes"). At March 31, 2003, the estimated fair value of the TE Products Senior Notes was approximately \$403.9 million. At March 31, 2003, we had outstanding \$500.0 million principal amount of 7.625% Senior Notes due 2012 and \$200.0 million principal amount of 6.125% Senior Notes due 2013. At March 31, 2003, the estimated fair value of the \$500.0 million 7.625% Senior Notes and the \$200.0 million 6.125% Senior Notes was approximately \$566.9 million and \$203.4 million, respectively.

As of March 31, 2003, TE Products had an interest rate swap agreement in place to hedge its exposure to changes in the fair value of its fixed rate 7.51% TE Products Senior Notes due 2028. We designated this swap agreement as a fair value hedge. The swap agreement has a notional amount of \$210.0 million and matures in January 2028 to match the principal and maturity of the TE Products Senior Notes. Under the swap agreement, TE Products pays a floating rate based on a three month U.S. Dollar LIBOR rate, plus a spread, and receives a fixed rate of interest of 7.51%. During the three months ended March 31, 2003, and 2002, we recognized reductions in interest expense of \$2.4 million and \$1.7 million, respectively, related to the difference between the fixed rate and the floating rate of interest on the interest rate swap. During the three months ended March 31, 2003, we measured the hedge effectiveness of this interest rate swap and noted that no gain or loss from ineffectiveness was required to be recognized. The fair value of this interest rate swap agreement was a gain of approximately \$12.3 million and \$13.6 million at March 31, 2003, and December 31, 2002, respectively. Utilizing the balance of the 7.51% TE Products Senior Notes outstanding at March 31, 2003, and including the effects of hedging activities, assuming market interest rates increase 100 basis points, the potential annual increase in interest expense is \$2.1 million.

As of March 31, 2003, we had an interest rate swap agreement in place to hedge our exposure to increases in the benchmark interest rate underlying our variable rate revolving credit facility. The term of the interest rate swap matches the maturity of the credit facility. We designated this swap agreement, which hedges exposure to variability in expected future cash flows attributed to changes in interest rates, as a cash flow hedge. The swap agreement is based on a notional amount of \$250.0 million. Under the swap agreement, we pay a fixed rate of interest of 6.955% and receive a floating rate based on a three month U.S. Dollar LIBOR rate. Since this swap is designated as a cash flow hedge, the changes in fair value, to the extent the swap is effective, are recognized in other comprehensive income until the hedged interest costs are recognized in earnings. During the three months ended March 31, 2003, and 2002, we recognized increases in interest expense of \$3.4 million and \$3.2 million, respectively, related to the difference between the fixed rate and the floating rate of interest on the interest rate swap. During the three months ended March 31, 2003, we measured the hedge effectiveness of this interest rate swap and noted that no gain or loss from ineffectiveness was required to be recognized. The fair value of the interest rate swap agreement was a loss of approximately \$14.6 million and \$20.1 million at March 31, 2003, and December 31, 2002, respectively. We anticipate that approximately \$14.3 million of the fair value will be transferred into earnings over the next twelve months.

On February 20, 2002, we entered into interest rate swap agreements, designated as fair value hedges, to hedge our exposure to changes in the fair value of our fixed rate 7.625% Senior Notes due 2012. The swap agreements had a combined notional amount of \$500.0 million and matured in 2012 to match the principal and maturity of the Senior Notes. Under the swap agreements, we paid a floating rate based on a U.S. Dollar LIBOR rate, plus a spread, and received a fixed rate of interest of 7.625%. On July 16, 2002, the swap agreements were terminated resulting in a gain of approximately \$18.0 million. Concurrent with the swap terminations, we entered into new interest rate swap agreements, with identical terms as the previous swap agreements; however, the floating rate was based upon a spread of an additional 50 basis points. In December 2002, the swap agreements entered into on July 16, 2002, were terminated, resulting in a gain of approximately \$26.9 million. The gains realized from the July 2002 and December 2002 swap terminations have been deferred as adjustments to the carrying value of the Senior Notes and are being amortized using the effective interest method as reductions to future interest expense over the remaining term of the Senior Notes. In the event of early extinguishment of the Senior Notes, any remaining unamortized gains would be recognized in the consolidated statement of income at the time of extinguishment.

Item 4. Controls and Procedures

Included in its recent Release No. 34-46427, effective August 29, 2002, the Securities and Exchange Commission adopted rules requiring reporting companies to maintain disclosure controls and procedures to provide reasonable assurance that a registrant is able to record, process, summarize and report the information required in the registrant's quarterly and annual reports under the Securities Exchange Act of 1934 (the "Exchange Act"). While we believe that our existing disclosure controls and procedures have been effective to accomplish these objectives, we intend to continue to examine, refine and formalize our disclosure controls and procedures and to monitor ongoing developments in this area.

The principal executive officer and principal financial officer of our general partner, after evaluating the effectiveness of our disclosure controls and procedures (as defined in Exchange Act Rules 13a-14(c) and Rule 15d-14(c)) as of a date within 90 days before the filing date of this Report, have concluded that, as of such date, our disclosure controls and procedures are adequate and effective to ensure that material information relating to us and our consolidated subsidiaries would be made known to them by others within those entities.

There have been no changes in our internal controls or in other factors known to us that could significantly affect those internal controls subsequent to the date of the evaluation, nor were there any significant deficiencies or material weaknesses in our internal controls. As a result, no corrective actions were required or undertaken.

Item 6. Exhibits and Reports on Form 8-K.

(a) Exhibits:

Exhibit Number	Description
3.1	Certificate of Limited Partnership of TEPPCO Partners, L.P. (Filed as Exhibit 3.2 to the Registration Statement of TEPPCO Partners, L.P. (Commission File No. 33-32203) and incorporated herein by reference).
3.2	Third Amended and Restated Agreement of Limited Partnership of TEPPCO Partners, L.P., dated September 21, 2001 (Filed as Exhibit 3.7 to Form 10-Q of TEPPCO Partners, L.P. (Commission File No. 1-10403) for the quarter ended September 30, 2001 and incorporated herein by reference).
4.1	Form of Certificate representing Limited Partner Units (Filed as Exhibit 4.1 to the Registration Statement of TEPPCO Partners, L.P. (Commission File No. 33-32203) and incorporated herein by reference).
4.2	Form of Indenture between TE Products Pipeline Company, Limited Partnership and The Bank of New York, as Trustee, dated as of January 27, 1998 (Filed as Exhibit 4.3 to TE Products Pipeline Company, Limited Partnership's Registration Statement on Form S-3 (Commission File No. 333-38473) and incorporated herein by reference).
4.3	Form of Certificate representing Class B Units (Filed as Exhibit 4.3 to Form 10-K of TEPPCO Partners, L.P. (Commission File No. 1-10403) for the year ended December 31, 1998 and incorporated herein by reference).
4.4	Form of Indenture between TEPPCO Partners, L.P., as issuer, TE Products Pipeline Company, Limited Partnership, TCTM, L.P., TEPPCO Midstream Companies, L.P. and Jonah Gas Gathering Company, as subsidiary guarantors, and First Union National Bank, NA, as trustee, dated as of February 20, 2002 (Filed as Exhibit 99.2 to Form 8-K of TEPPCO Partners, L.P. (Commission File No. 1-10403) dated as of February 20, 2002 and incorporated herein by reference).
4.5	First Supplemental Indenture between TEPPCO Partners, L.P., as issuer, TE Products Pipeline Company, Limited Partnership, TCTM, L.P., TEPPCO Midstream Companies, L.P. and Jonah Gas Gathering Company, as subsidiary guarantors, and First Union National Bank, NA, as trustee, dated as of February 20, 2002 (Filed as Exhibit 99.3 to Form 8-K of TEPPCO Partners, L.P. (Commission File No. 1-10403) dated as of February 20, 2002 and incorporated herein by reference).
4.6	Second Supplemental Indenture, dated as of June 27, 2002, among TEPPCO Partners, L.P., as issuer, TE Products Pipeline Company, Limited Partnership, TCTM, L.P., TEPPCO Midstream Companies, L.P., and Jonah Gas Gathering Company, as Initial Subsidiary Guarantors, and Val Verde Gas Gathering Company, L.P., as New Subsidiary Guarantor, and Wachovia Bank, National Association, formerly known as First Union National Bank, as trustee (Filed as Exhibit 4.6 to Form 10-Q of TEPPCO Partners, L.P. (Commission File No. 1-10403) for the quarter ended June 30, 2002 and incorporated herein by reference).
4.7	Third Supplemental Indenture among TEPPCO Partners, L.P. as issuer, TE Products Pipeline Company, Limited Partnership, TCTM, L.P., TEPPCO Midstream Companies, L.P., Jonah Gas Gathering Company and Val Verde Gas Gathering Company, L.P. as Subsidiary Guarantors, and Wachovia Bank, National Association, as trustee, dated as of January 30, 2003 (Filed as Exhibit 4.7 to Form 10-K of TEPPCO Partners, L.P. (Commission File No. 1-10403) for the year ended December 31, 2002 and incorporated herein by reference).

- 10.1+ Duke Energy Corporation Executive Savings Plan (Filed as Exhibit 10.7 to Form 10-K of TEPPCO Partners, L.P. (Commission File No. 1-10403) for the year ended December 31, 1999 and incorporated herein by reference).
- 10.2+ Duke Energy Corporation Executive Cash Balance Plan (Filed as Exhibit 10.8 to Form 10-K of TEPPCO Partners, L.P. (Commission File No. 1-10403) for the year ended December 31, 1999 and incorporated herein by reference).
- 10.3+ Duke Energy Corporation Retirement Benefit Equalization Plan (Filed as Exhibit 10.9 to Form 10-K for TEPPCO Partners, L.P. (Commission File No. 1-10403) for the year ended December 31, 1999 and incorporated herein by reference).
- 10.4+Texas Eastern Products Pipeline Company 1994 Long Term Incentive Plan executed on March 8,
1994 (Filed as Exhibit 10.1 to Form 10-Q of TEPPCO Partners, L.P. (Commission File No. 1-
10403) for the quarter ended March 31, 1994 and incorporated herein by reference).
- 10.5+ Texas Eastern Products Pipeline Company 1994 Long Term Incentive Plan, Amendment 1, effective January 16, 1995 (Filed as Exhibit 10.12 to Form 10-Q of TEPPCO Partners, L.P. (Commission File No. 1-10403) for the quarter ended June 30, 1999 and incorporated herein by reference).
- 10.6 Asset Purchase Agreement between Duke Energy Field Services, Inc. and TEPPCO Colorado, LLC, dated March 31, 1998 (Filed as Exhibit 10.14 to Form 10-Q of TEPPCO Partners, L.P. (Commission File No. 1-10403) for the quarter ended March 31, 1998 and incorporated herein by reference).
- 10.7 Contribution Agreement between Duke Energy Transport and Trading Company and TEPPCO Partners, L.P., dated October 15, 1998 (Filed as Exhibit 10.16 to Form 10-K of TEPPCO Partners, L.P. (Commission File No. 1-10403) for the year ended December 31, 1998 and incorporated herein by reference).
- 10.8 Guaranty Agreement by Duke Energy Natural Gas Corporation for the benefit of TEPPCO Partners, L.P., dated November 30, 1998, effective November 1, 1998 (Filed as Exhibit 10.17 to Form 10-K of TEPPCO Partners, L.P. (Commission File No. 1-10403) for the year ended December 31, 1998 and incorporated herein by reference).
- 10.9+ Form of Employment Agreement between the Company and Thomas R. Harper, Charles H. Leonard, James C. Ruth, John N. Goodpasture, Leonard W. Mallett, Stephen W. Russell, David E. Owen, and Barbara A. Carroll (Filed as Exhibit 10.20 to Form 10-K of TEPPCO Partners, L.P. (Commission File No. 1-10403) for the year ended December 31, 1998 and incorporated herein by reference).
- 10.10 Services and Transportation Agreement between TE Products Pipeline Company, Limited Partnership and Fina Oil and Chemical Company, BASF Corporation and BASF Fina Petrochemical Limited Partnership, dated February 9, 1999 (Filed as Exhibit 10.22 to Form 10-Q of TEPPCO Partners, L.P. (Commission File No. 1-10403) for the quarter ended March 31, 1999 and incorporated herein by reference).
- 10.11 Call Option Agreement, dated February 9, 1999 (Filed as Exhibit 10.23 to Form 10-Q of TEPPCO Partners, L.P. (Commission File No. 1-10403) for the quarter ended March 31, 1999 and incorporated herein by reference).
- 10.12+Texas Eastern Products Pipeline Company Retention Incentive Compensation Plan, effective
January 1, 1999 (Filed as Exhibit 10.24 to Form 10-Q of TEPPCO Partners, L.P. (Commission File
No. 1-10403) for the quarter ended March 31, 1999 and incorporated herein by reference).
- 10.13+ Form of Employment and Non-Compete Agreement between the Company and J. Michael Cockrell effective January 1, 1999 (Filed as Exhibit 10.29 to Form 10-Q of TEPPCO Partners, L.P. (Commission File No. 1-10403) for the quarter ended September 30, 1999 and incorporated herein by reference).
- 10.14+Texas Eastern Products Pipeline Company Non-employee Directors Unit Accumulation Plan,
effective April 1, 1999 (Filed as Exhibit 10.30 to Form 10-Q of TEPPCO Partners,

L.P. (Commission File No. 1-10403) for the quarter ended September 30, 1999 and incorporated herein by reference).

- 10.15+ Texas Eastern Products Pipeline Company Non-employee Directors Deferred Compensation Plan, effective November 1, 1999 (Filed as Exhibit 10.31 to Form 10-Q of TEPPCO Partners, L.P. (Commission File No. 1-10403) for the quarter ended September 30, 1999 and incorporated herein by reference).
- 10.16+Texas Eastern Products Pipeline Company Phantom Unit Retention Plan, effective August 25,
1999 (Filed as Exhibit 10.32 to Form 10-Q of TEPPCO Partners, L.P. (Commission File No. 1-
10403) for the quarter ended September 30, 1999 and incorporated herein by reference).
- 10.17 Amended and Restated Purchase Agreement By and Between Atlantic Richfield Company and Texas Eastern Products Pipeline Company With Respect to the Sale of ARCO Pipe Line Company, dated as of May 10, 2000. (Filed as Exhibit 2.1 to Form 10-Q of TEPPCO Partners, L.P. (Commission File No. 1-10403) for the quarter ended March 31, 2000 and incorporated herein by reference).
- 10.18+ Texas Eastern Products Pipeline Company, LLC 2000 Long Term Incentive Plan, Amendment and Restatement, effective January 1, 2000 (Filed as Exhibit 10.28 to Form 10-K of TEPPCO Partners, L.P. (Commission File No. 1-10403) for the year ended December 31, 2000 and incorporated herein by reference).
- 10.19+ TEPPCO Supplemental Benefit Plan, effective April 1, 2000 (Filed as Exhibit 10.29 to Form 10-K of TEPPCO Partners, L.P. (Commission File No. 1-10403) for the year ended December 31, 2000 and incorporated herein by reference).
- 10.20+ Employment Agreement with Barry R. Pearl (Filed as Exhibit 10.30 to Form 10-Q of TEPPCO Partners, L.P. (Commission File No. 1-10403) for the quarter ended March 31, 2001 and incorporated herein by reference).
- 10.21 Amended and Restated Credit Agreement among TEPPCO Partners, L.P. as Borrower, SunTrust Bank as Administrative Agent and LC Issuing Bank, and Certain Lenders, dated as of April 6, 2001 (\$500,000,000 Revolving Facility) (Filed as Exhibit 10.31 to Form 10-Q of TEPPCO Partners, L.P. (Commission File No. 1-10403) for the quarter ended March 31, 2001 and incorporated herein by reference).
- 10.22 Credit Agreement among TEPPCO Partners, L.P. as Borrower, SunTrust Bank as Administrative Agent, and Certain Lenders, dated as of April 6, 2001 (\$200,000,000 Revolving Facility) (Filed as Exhibit 10.32 to Form 10-Q of TEPPCO Partners, L.P. (Commission File No. 1-10403) for the quarter ended March 31, 2001 and incorporated herein by reference).
- 10.23 Purchase and Sale Agreement By and Among Green River Pipeline, LLC and McMurry Oil Company, Sellers, and TEPPCO Partners, L.P., Buyer, dated as of September 7, 2000. (Filed as Exhibit 10.31 to Form 10-Q of TEPPCO Partners, L.P. (Commission File No. 1-10403) for the quarter ended September 30, 2001 and incorporated herein by reference).
- 10.24 Credit Agreement Among TEPPCO Partners, L.P. as Borrower, SunTrust Bank, as Administrative Agent and Certain Lenders, dated as of September 28, 2001 (\$400,000,000 Term Facility) (Filed as Exhibit 10.32 to Form 10-Q of TEPPCO Partners, L.P. (Commission File No. 1-10403) for the quarter ended September 30, 2001 and incorporated herein by reference).
- 10.25 Amendment 1, dated as of September 28, 2001, to the Amended and Restated Credit Agreement among TEPPCO Partners, L.P. as Borrower, SunTrust Bank as Administrative Agent and LC Issuing Bank, and Certain Lenders, dated as of April 6, 2001 (\$500,000,000 Revolving Facility) (Filed as Exhibit 10.33 to Form 10-Q of TEPPCO Partners, L.P. (Commission File No. 1-10403) for the quarter ended September 30, 2001 and incorporated herein by reference).
- 10.26 Amendment 1, dated as of September 28, 2001, to the Credit Agreement among TEPPCO Partners, L.P. as Borrower, SunTrust Bank as Administrative Agent, and Certain Lenders,

dated as of April 6, 2001 (\$200,000,000 Revolving Facility) (Filed as Exhibit 10.34 to Form 10-Q of TEPPCO Partners, L.P. (Commission File No. 1-10403) for the quarter ended September 30, 2001 and incorporated herein by reference).

- 10.27 Amendment and Restatement, dated as of November 13, 2001, to the Credit Agreement among TEPPCO Partners, L.P. as Borrower, SunTrust Bank as Administrative Agent, and Certain Lenders, dated as of April 6, 2001 (\$200,000,000 Revolving Facility) (Filed as Exhibit 10.35 to Form 10-K of TEPPCO Partners, L.P (Commission File No. 1-10403) for the year ended December 31, 2001 and incorporated herein by reference).
- 10.28 Second Amendment and Restatement, dated as of November 13, 2001, to the Amended and Restated Credit Agreement amount TEPPCO Partners, L.P. as Borrower, SunTrust Bank as Administrative Agent and LC Issuing Bank, and Certain Lenders, dated as of April 6, 2001 (\$500,000,000 Revolving Facility) (Filed as Exhibit 10.36 to Form 10-K of TEPPCO Partners, L.P (Commission File No. 1-10403) for the year ended December 31, 2001 and incorporated herein by reference).
- 10.29 Second Amended and Restated Agreement of Limited Partnership of TE Products Pipeline Company, Limited Partnership, dated September 21, 2001 (Filed as Exhibit 3.8 to Form 10-Q of TEPPCO Partners, L.P. (Commission File No. 1-10403) for the quarter ended September 30, 2001 and incorporated herein by reference).
- 10.30 Amended and Restated Agreement of Limited Partnership of TCTM, L.P., dated September 21,
 2001 (Filed as Exhibit 3.9 to Form 10-Q of TEPPCO Partners, L.P. (Commission File No. 1-10403) for the quarter ended September 30, 2001 and incorporated herein by reference).
- 10.31 Contribution, Assignment and Amendment Agreement among TEPPCO Partners, L.P., TE Products Pipeline Company, Limited Partnership, TCTM, L.P., Texas Eastern Products Pipeline Company, LLC, and TEPPCO GP, Inc., dated July 26, 2001 (Filed as Exhibit 3.6 to Form 10-Q of TEPPCO Partners, L.P. (Commission File No. 1-10403) for the quarter ended June 30, 2001 and incorporated herein by reference).
- 10.32 Certificate of Formation of TEPPCO Colorado, LLC (Filed as Exhibit 3.2 to Form 10-Q of TEPPCO Partners, L.P. (Commission File No. 1-10403) for the quarter ended March 31, 1998 and incorporated herein by reference).
- 10.33 Agreement of Limited Partnership of TEPPCO Midstream Companies, L.P., dated September 24, 2001 (Filed as Exhibit 3.10 to Form 10-Q of TEPPCO Partners, L.P. (Commission File No. 1-10403) for the quarter ended September 30, 2001 and incorporated herein by reference).
- 10.34 Agreement of Partnership of Jonah Gas Gathering Company dated June 20, 1996 as amended by that certain Assignment of Partnership Interests dated September 28, 2001 (Filed as Exhibit 10.40 to Form 10-K of TEPPCO Partners, L.P. (Commission File No. 1-10403) for the year ended December 31, 2001 and incorporated herein by reference).
- 10.35 Unanimous Written Consent of the Board of Directors of TEPPCO GP, Inc. dated February 13, 2002 (Filed as Exhibit 10.41 to Form 10-K of TEPPCO Partners, L.P. (Commission File No. 1-10403) for the year ended December 31, 2001 and incorporated herein by reference).
- 10.36 Credit Agreement among TEPPCO Partners, L.P. as Borrower, SunTrust Bank as Administrative Agent and Certain Lenders, as Lenders dated as of March 28, 2002 (\$200,000,000 Revolving Credit Facility) (Filed as Exhibit 10.44 to Form 10-Q of TEPPCO Partners, L.P. (Commission File No. 1-10403) for the three months ended March 31, 2002 and incorporated herein by reference).
- 10.37 Amended and Restated Credit Agreement among TEPPCO Partners, L.P. as Borrower, SunTrust Bank, as Administrative Agent and LC Issuing Bank and Certain Lenders, as Lenders dated as of March 28, 2002 (\$500,000,000 Revolving Facility) (Filed as Exhibit 10.45 to Form 10-Q of TEPPCO Partners, L.P. (Commission File No. 1-10403) for the three months ended March 31, 2002 and incorporated herein by reference).

- 10.38 Purchase and Sale Agreement between Burlington Resources Gathering Inc. as Seller and TEPPCO Partners, L.P., as Buyer, dated May 24, 2002 (Filed as Exhibit 99.1 to Form 8-K of TEPPCO Partners, L.P. (Commission File No. 1-10403) dated as of July 2, 2002 and incorporated herein by reference).
- 10.39 Credit Agreement among TEPPCO Partners, L.P., as Borrower, SunTrust Bank, as Administrative Agent and Certain Lenders, as Lenders dated as of June 27, 2002 (\$200,000,000 Term Facility) (Filed as Exhibit 99.2 to Form 8-K of TEPPCO Partners, L.P. (Commission File No. 1-10403) dated as of July 2, 2002 and incorporated herein by reference).
- 10.40 Amendment, dated as of June 27, 2002 to the Amended and Restated Credit Agreement among TEPPCO Partners, L.P., as Borrower, SunTrust Bank, as Administrative Agent, and Certain Lenders, dated as of March 28, 2002 (\$500,000,000 Revolving Credit Facility) (Filed as Exhibit 99.3 to Form 8-K of TEPPCO Partners, L.P. (Commission File No. 1-10403) dated as of July 2, 2002 and incorporated herein by reference).
- 10.41 Amendment 1, dated as of June 27, 2002 to the Credit Agreement among TEPPCO Partners, L.P., as Borrower, SunTrust Bank, as Administrative Agent and Certain Lenders, dated as of March 28, 2002 (\$200,000,000 Revolving Credit Facility) (Filed as Exhibit 99.4 to Form 8-K of TEPPCO Partners, L.P. (Commission File No. 1-10403) dated as of July 2, 2002 and incorporated herein by reference).
- 10.42 Agreement of Limited Partnership of Val Verde Gas Gathering Company, L.P., dated May 29, 2002 (Filed as Exhibit 10.48 to Form 10-Q of TEPPCO Partners, L.P. (Commission File No. 1-10403) for the quarter ended June 30, 2002 and incorporated herein by reference).
- 10.43+ Texas Eastern Products Pipeline Company, LLC 2002 Phantom Unit Retention Plan, effective June 1, 2002 (Filed as Exhibit 10.43 to Form 10-Q of TEPPCO Partners, L.P. (Commission File No. 1-10403) for the quarter ended June 30, 2002 and incorporated herein by reference).
- 10.44+ Amended and Restated TEPPCO Supplemental Benefit Plan, effective November 1, 2002 (Filed as Exhibit 10.44 to Form 10-K of TEPPCO Partners, L.P. (Commission File No. 1-10403) for the year ended December 31, 2002 and incorporated herein by reference).
- 10.45+ Texas Eastern Products Pipeline Company, LLC 2000 Long Term Incentive Plan, Second Amendment and Restatement, effective January 1, 2003 (Filed as Exhibit 10.45 to Form 10-K of TEPPCO Partners, L.P. (Commission File No. 1-10403) for the year ended December 31, 2002 and incorporated herein by reference).
- 10.46+ Amended and Restated Texas Eastern Products Pipeline Company, LLC Management Incentive Compensation Plan, effective January 1, 2003 (Filed as Exhibit 10.46 to Form 10-K of TEPPCO Partners, L.P. (Commission File No. 1-10403) for the year ended December 31, 2002 and incorporated herein by reference).
- 10.47+ Amended and Restated TEPPCO Retirement Cash Balance Plan, effective January 1, 2002 (Filed as Exhibit 10.47 to Form 10-K of TEPPCO Partners, L.P. (Commission File No. 1-10403) for the year ended December 31, 2002 and incorporated herein by reference).
- 10.48 Formation Agreement between Panhandle Eastern Pipe Line Company and Marathon Ashland Petroleum LLC and TE Products Pipeline Company, Limited Partnership, dated as of August 10, 2000 (Filed as Exhibit 10.48 to Form 10-K of TEPPCO Partners, L.P. (Commission File No. 1-10403) for the year ended December 31, 2002 and incorporated herein by reference).
- 10.49 Amended and Restated Limited Liability Company Agreement of Centennial Pipeline LLC dated as of August 10, 2000 (Filed as Exhibit 10.49 to Form 10-K of TEPPCO Partners, L.P. (Commission File No. 1-10403) for the year ended December 31, 2002 and incorporated herein by reference).
- 10.50 Guaranty Agreement, dated as of September 27, 2002, between TE Products Pipeline Company, Limited Partnership and Marathon Ashland Petroleum LLC for Note

	Partners, L.P. (Commission File No. 1-10403) for the year ended December 31, 2002 and incorporated herein by reference).
10.51	LLC Membership Interest Purchase Agreement By and Between CMS Panhandle Holdings, LLC, As Seller and Marathon Ashland Petroleum LLC and TE Products Pipeline Company, Limited Partnership, Severally as Buyers, dated February 10, 2003 (Filed as Exhibit 10.51 to Form 10-K of TEPPCO Partners, L.P. (Commission File No. 1-10403) for the year ended December 31, 2002 and incorporated herein by reference).
10.52*	Joint Development Agreement between TE Products Pipeline Company, Limited Partnership and Louis Dreyfus Plastics Corporation dated February 10, 2000.
12.1*	Statement of Computation of Ratio of Earnings to Fixed Charges.
21	Subsidiaries of the Partnership (Filed as Exhibit 21 to Form 10-Q of TEPPCO Partners, L.P. (Commission File No. 1-10403) for the quarter ended June 30, 2002 and incorporated herein by reference).
99.1*	Certification of Chief Executive Officer Pursuant to Section 906 of the Sarbanes-Oxley Act of 2002.
99.2*	Certification of Chief Financial Officer Pursuant to Section 906 of the Sarbanes-Oxley Act of 2002.

Agreements of Centennial Pipeline LLC (Filed as Exhibit 10.50 to Form 10-K of TEPPCO

* Filed herewith.

+ A management contract or compensation plan or arrangement.

(b) Reports on Form 8-K filed during the quarter ended March 31, 2003:

Reports on Form 8-K were filed with the Securities and Exchange Commission on January 21, 2003, January 30, 2003, February 6, 2003 and March 4, 2003.

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the Registrants have duly caused this report to be signed on its behalf by the undersigned duly authorized officer and principal financial officer.

TEPPCO Partners, L.P.

(Registrant) (A Delaware Limited Partnership)

By: Texas Eastern Products Pipeline Company, LLC, as General Partner

By: /s/ BARRY R. PEARL

Barry R. Pearl, President and Chief Executive Officer

By: /s/ CHARLES H. LEONARD

Charles H. Leonard, Senior Vice President and Chief Financial Officer

Date: May 1, 2003

CERTIFICATION OF CHIEF EXECUTIVE OFFICER PURSUANT TO SECTION 302 OF THE SARBANES-OXLEY ACT OF 2002

I, BARRY R. PEARL, certify that:

- 1. I have reviewed this quarterly report on Form 10-Q of TEPPCO Partners, L.P.;
- 2. Based on my knowledge, this quarterly report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this quarterly report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this quarterly report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this quarterly report;
- 4. The registrant's other certifying officers and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-14 and 15d-14) for the registrant and have:
 - a) designed such disclosure controls and procedures to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this quarterly report is being prepared;
 - evaluated the effectiveness of the registrant's disclosure controls and procedures as of a date within 90 days prior to the filing date of this quarterly report (the "Evaluation Date"); and
 - c) presented in this quarterly report our conclusions about the effectiveness of the disclosure controls and procedures based on our evaluation as of the Evaluation Date;
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation, to the registrant's auditors and the audit committee of registrant's board of directors (or persons performing the equivalent function):
 - a) all significant deficiencies in the design or operation of internal controls which could adversely affect the registrant's ability to record, process, summarize and report financial data and have identified for the registrant's auditors any material weaknesses in internal controls; and
 - b) any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal controls; and
- 6. The registrant's other certifying officers and I have indicated in this quarterly report whether there were significant changes in internal controls or in other factors that could significantly affect internal controls subsequent to the date of our most recent evaluation, including any corrective actions with regard to significant deficiencies and material weaknesses.
- May 1, 2003

Date

/s/ BARRY R. PEARL

Barry R. Pearl President and Chief Executive Officer Texas Eastern Products Pipeline Company, LLC, General Partner

CERTIFICATION OF CHIEF FINANCIAL OFFICER PURSUANT TO SECTION 302 OF THE SARBANES-OXLEY ACT OF 2002

I, CHARLES H. LEONARD, certify that:

- 1. I have reviewed this quarterly report on Form 10-Q of TEPPCO Partners, L.P.;
- 2. Based on my knowledge, this quarterly report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this quarterly report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this quarterly report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this quarterly report;
- 4. The registrant's other certifying officers and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-14 and 15d-14) for the registrant and have:
 - a) designed such disclosure controls and procedures to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this quarterly report is being prepared;
 - evaluated the effectiveness of the registrant's disclosure controls and procedures as of a date within 90 days prior to the filing date of this quarterly report (the "Evaluation Date"); and
 - c) presented in this quarterly report our conclusions about the effectiveness of the disclosure controls and procedures based on our evaluation as of the Evaluation Date;
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation, to the registrant's auditors and the audit committee of registrant's board of directors (or persons performing the equivalent function):
 - a) all significant deficiencies in the design or operation of internal controls which could adversely affect the registrant's ability to record, process, summarize and report financial data and have identified for the registrant's auditors any material weaknesses in internal controls; and
 - b) any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal controls; and
- 6. The registrant's other certifying officers and I have indicated in this quarterly report whether there were significant changes in internal controls or in other factors that could significantly affect internal controls subsequent to the date of our most recent evaluation, including any corrective actions with regard to significant deficiencies and material weaknesses.
- May 1, 2003

Date

/s/ CHARLES H. LEONARD

Charles H. Leonard Senior Vice President and Chief Financial Officer Texas Eastern Products Pipeline Company, LLC, General Partner

Exhibit Index

Exhibit No.	Description	
10.52	Joint Development Agreement between TE Products Pipeline Company, Limited Partnership and Louis Dreyfus Plastics Corporation dated February 10, 2000.	
12.1	Statement of Computation of Ratio of Earnings to Fixed Charges	
99.1	Certification of Chief Executive Officer Pursuant to Section 906 of the Sarbanes-Oxley Act of 2002.	
99.2	Certification of Chief Financial Officer Pursuant to Section 906 of the Sarbanes-Oxley Act of 2002.	

EXHIBIT 10.52

JOINT DEVELOPMENT AGREEMENT

BETWEEN

TE PRODUCTS PIPELINE COMPANY, LIMITED PARTNERSHIP

AND

LOUIS DREYFUS PLASTICS CORPORATION

DATED FEBRUARY 10, 2000

TABLE OF CONTENTS

PAGE ARTICLE I DEFINITIONS AND
CONSTRUCTION
OTHER DEFINITIONS AND PROVISIONS
HEADINGS
ASSETS
MARKETING AND DEVELOPMENT ACTIVITIES
BUDGETS
12 2.4 COMMUNICATION
12 2.5 CAPITAL EXPENDITURES 12 2.6
TEPPC0 EXCULPATION
2.7 DREYFUS EXCULPATION
ARTICLE III FINANCIAL OPERATIONS AND PAYMENTS
RESULTS 13 3.2
PAYMENTS TO DREYFUS 13 3.3
FUNDING OF MB BUSINESS 14 3.4
ACCOUNTS, BOOKS AND RECORDS 14 3.5 EXCEPTIONS
TO FINANCIAL REPORTS 15 ARTICLE IV REPRESENTATIONS AND WARRANTIES OF
TEPPCO 15 4.1 CORPORATE MATTERS; ASSET OWNERSHIP 15 0
AGREEMENT; NO CONFLICT 15 4.3 LIABILITIES OF THE MB
BUSINESS 15 4.4 FINANCIAL STATEMENTS 16 4.5
FINDER'S
ARTICLE V REPRESENTATIONS AND WARRANTIES OF DREYFUS
OWNERSHIP 16 5.2 VALIDITY OF
AGREEMENT; NO CONFLICT 16 5.3 FINANCIAL
STATEMENTS 16 5.4 FINDER'S
FEE 17 ARTICLE VI TRANSFER OF TEPPCO ASSETS OR RIGHT TO USE
MILESTONE
MILESTONE 17 6.3 CLOSING
TIME AND LOCATION FOR TRANSFER
NEWCO
18 6.6 INABILITY TO TRANSFER ASSETS 19 6.7 FURTHER ASSURANCE BY THE PARTIES 19 6.8
THIRD PARTY CONSENTS19

i

PAGE ARTICLE VII CONDITIONS TO
CLOSING 20 7.1 CONDITIONS TO OBLIGATION OF
DREYFUS
RIGHTS 22 8.1 TRANSFER OF CONTRACT RIGHTS PROHIBITED 22 8.2 FAILURE TO MEET
MILESTONE
DISPUTE 23 8.5 CERTAIN TERMINATION
PAYMENTS
25 8.8 TRANSFER PROCEDURES
ARTICLE IX ADDITIONAL AGREEMENTS
EFFORTS TO SATISFY CONDITIONS
OF RECORDS 27 9.3 NONDISCLOSURE OF PROPRIETARY
INFORMATION
COVENANT
30 10.3 REASONABLENESS OF COVENANT
INJUNCTIVE RELIEF
ARTICLE XI INDEMNIFICATION
11.1 INDEMNITY OBLIGATION 31 11.2
PROCEDURE
THRESHOLD
33 11.5 EXCLUSIVE REMEDY; LIMITATIONS
LIMITATIONS
LIMITATIONS
LIMITATIONS
LIMITATIONS

ii

TABLE OF CONTENTS

PAGE 15.1 FURTHER
ASSURANCES
NOTICES
36 15.3 GOVERNING
LAW
ENTIRE
AGREEMENT
ASSIGNMENT
37 15.6
SUCCESSORS
37 15.7 AMENDMENTS;
WAIVER
38 15.9
SEVERABILITY
38 15.10 NO THIRD PARTY
BENEFICIARIES 38 15.11
INDEPENDENT CONTRACTOR; NO PARTNERSHIP
TRANSACTION
BUSINESS
DAYS

iii

JOINT DEVELOPMENT AGREEMENT

This JOINT DEVELOPMENT AGREEMENT (this "Agreement") dated as of February 10, 2000, by and between TE Products Pipeline Company, Limited Partnership, a Delaware limited partnership ("TEPPCO"), and Louis Dreyfus Plastics Corporation, a Delaware corporation ("Dreyfus").

WITNESSETH:

WHEREAS, TEPPCO is engaged in the MB Business in Chambers and Harris Counties, Texas utilizing the TEPPCO Assets; and

WHEREAS, Dreyfus is engaged in the merchandising and trading of the Products, and in connection therewith currently utilizes the TEPPCO Assets; and

WHEREAS, TEPPCO and Dreyfus desire to achieve certain synergistic benefits by combining the MB Business and the expertise of Dreyfus in marketing such business activities to achieve increased utilization of the TEPPCO Assets during and after the Development Period; and

WHEREAS, TEPPCO intends to account for the MB Business as a separate division of TEPPCO during the Term of this Agreement as hereinafter set forth to better reflect the results of such operations; and

WHEREAS, upon successful completion of certain objectives during the Development Period in accordance with the provisions of this Agreement, the Parties intend to enter into a joint venture regarding the MB Business through the contribution of the assets and liabilities constituting the MB Business, the TEPPCO Assets and certain other assets and/or improvements to a newly formed Delaware limited partnership ("Newco"), which would be jointly owned by TEPPCO and Dreyfus, such contribution to be in accordance with the terms and conditions of this Agreement and the other agreements contemplated hereby;

NOW, THEREFORE, in consideration of the premises and the mutual promises and obligations contained herein, and intending to be legally bound, TEPPCO and Dreyfus agree as follows:

ARTICLE I

DEFINITIONS AND CONSTRUCTION

1.1 DEFINED TERMS. The capitalized terms used in this Agreement shall have the meanings ascribed to them as follows:

"Acceptance" shall have the meaning given that term in Section 8.6;

"Accounting Procedures" shall mean the accounting policies and procedures set forth in Annex I to the Operating Procedures;

"Acquisition Proposal" shall have the meaning given that term in Section 8.6;

"Affiliate" means, when used with respect to a specified Person, any other Person directly or indirectly controlling or controlled by or under direct or indirect common control with the specified Person, provided that Newco shall not be deemed to be an Affiliate of TEPPCO, Dreyfus or any of their respective subsidiaries or Affiliates. For purposes of this definition, "control", when used with respect to any specified Person, means the power to direct the management and policies of the Person, directly or indirectly, whether through the ownership of voting securities, by contract, by family relationship or otherwise; and the terms "controlling" and "controlled" have the meanings correlative to the foregoing. Notwithstanding the foregoing, for purposes of this Agreement, Louis Dreyfus Natural Gas Corporation, an Oklahoma corporation, shall not be deemed an Affiliate of Dreyfus;

"Assumed Liabilities" shall mean the liabilities, debts and obligations, whether known or unknown, contingent or liquidated, including those set forth on Exhibit A hereto, of or relating to the MB Business as of a specific date, including all liabilities, debts and obligations related to the MB Business arising during the Term or based upon events occurring or facts or circumstances existing prior to the Effective Date, but excluding all indebtedness for borrowed money except for indebtedness for Discretionary Capital Expenditures previously approved by Dreyfus;

"Base EBITDA" means, for the MB Business, for any 12 month period the lesser of (i) its EBITDA for such period and (ii) 7.3 Million;

"Budgets" shall have the meaning given that term in Section 2.3 hereof;

"Business Day" means any day other than a Saturday, Sunday or bank holiday in Houston, Texas;

"Capital Budget" shall have the meaning given that term in Section 2.3 hereof;

"Claim" means any demand, demand letter, claim or notice of noncompliance or violation (written or oral) or Proceeding;

"Claim Notice" shall have the meaning given that term in Section 11.211.2(a) hereof;

"Closing" shall have the meaning given that term in Section 6.3;

"Code" means the Internal Revenue Code of 1986, as amended, or any amending or superseding tax laws of the United States of America;

"Competing Business Transaction" shall have the meaning given that term in Section 10.1(c);

"Contract Rights" shall have the meaning given that term in Section 8.1(a);

"Conveyance Document" means the General Conveyance and Assumption Agreement in the form set forth in Exhibit B hereto;

"Development Period" shall mean the period beginning on January 1, 2000 and ending at midnight on December 31, 2002;

"Discretionary Capital Expenditures" means any expenditure during the Term relating to the TEPPCO Assets or the MB Business that is classified under GAAP as a capital expenditure and (i) is not a Mandatory Capital Expenditure, or (ii) if classified as a Mandatory Capital Expenditure, is associated or is in conjunction with, or is the result of, a prior or current Discretionary Capital Expenditure, in which case the portion of such Mandatory Capital Expenditure that is allocable to such prior or current Discretionary Capital Expenditure shall be classified as a Discretionary Capital Expenditure;

"Disposition" shall have the meaning given that term in Section 8.6;

"Dreyfus" shall have the meaning given that term in the preamble and any successor or assign permitted by this Agreement;

"Dreyfus Change of Control" means an event that causes S. A. Louis Dreyfus et Cie., a French societe anonyme, together with its Affiliates, to cease to own, directly or indirectly, at least a 50% voting interest in (x) Dreyfus or (y) in the event of the transfer of Dreyfus' Interest to an Affiliate of Dreyfus, in such Affiliate;

"Dreyfus Investment" shall have the meaning given that term in Section 8.2;

"Dreyfus Storage Agreement" shall mean the Storage and Service Agreement in the form attached as Exhibit C hereto;

"EBITDA" means, for the MB Business, its earnings for any period on a GAAP basis before all charges for such period for depreciation and amortization, interest (except as provided below) and income taxes. Notwithstanding the foregoing, all interest incurred in connection with borrowings the proceeds of which are used for Discretionary Capital Expenditures shall be included as an expense in calculating EBITDA. As an example only, Exhibit D sets forth a calculation of EBITDA for the MB Business based on historical averages except as otherwise noted in such Exhibit;

"Effective Date" shall have the meaning given that term in Section 6.3;

"Effective Time" shall have the meaning given that term in Section 6.3;

"Environmental Liability" means any liability, contingent or otherwise (including any liability for damages, costs of environmental remediation, fines, or penalties) resulting from or based upon (a) a violation of any Environmental Law, (b) the generation, use, handling, transportation, storage, treatment or disposal of any Hazardous Materials, (c) exposure to any Hazardous Materials, (d) the release or threatened release of any Hazardous Materials into the environment; or (e) any contract, agreement or other consensual arrangement pursuant to which liability is assumed or imposed with respect to any of the foregoing.

"Environmental Law" or "Environmental Laws" means all laws, rules, regulations, statutes, ordinances, decrees or orders of any governmental entity relating to (i) the control of any potential pollutant or protection of the air, water or land, (ii) solid, gaseous or liquid waste generation, handling, treatment, storage, disposal or transportation, and (iii) exposure to hazardous, toxic or other substances alleged to be harmful, and includes without limitation, (a) the terms and conditions of any license, permit, approval, or other authorization by any governmental entity, and (b) judicial, administrative, or other regulatory decrees, judgments, and orders of any governmental entity.

"First Party" shall have the definition given that term in Section 10.1(c);

"Fiscal Year" means the calendar year;

"GAAP" means Generally Accepted Accounting Principles in the United States of America consistently applied for the time periods involved;

"General Partner" means a Delaware limited liability company, formed by the Parties to serve as the general partner of Newco;

"Governmental Authority" means any entity of or pertaining to government, including any federal, state, local, other governmental or administrative authority, agency, court, tribunal, arbitrator, commission, board or bureau;

"Hazardous Materials" means any (i) toxic or hazardous materials or substances; (ii) solid or hazardous wastes, including asbestos, polychlorinated biphenyls, mercury, buried contaminants, chemicals, flammable or explosive materials; (iii) radioactive materials; (iv) petroleum wastes and spills or releases of petroleum products; and (v) any other chemical, pollutant, contaminant, substance or waste that is regulated by any governmental entity under any Environmental Law.

"Indemnified Party" shall have the meaning given that term in Section 11.2;

"Indemnifying Party" shall have the meaning given that term in Section 11.2;

"IRS" means the Internal Revenue Service of the United States of America;

"Lien" means any lien, mortgage, pledge, claim, charge, security interest or other encumbrance, option, defect or other rights of any third person of any nature whatsoever;

"LLC Agreement" means the Limited Liability Company Agreement of the General Partner, in the form attached as Exhibit E hereto;

"Losses" means any and all damages, losses, liabilities, judgments, payments, obligations, penalties, assessments, costs, disbursements or expenses (including reasonable fees, disbursements and expenses of attorneys, accountants and other professional advisors and of expert witnesses and costs of investigation and preparation of any kind or nature whatsoever);

"Mandatory Capital Expenditure" means all expenditures during the Term relating to the TEPPCO Assets or the MB Business that are classified under GAAP as a capital expenditure, the primary function of which is for regulatory compliance, safety or operational integrity of the

MB Business or TEPPCO Assets or is otherwise the result of actions by TEPPCO in emergency situations as provided in the Operating Procedures;

"Material Adverse Effect" means, with respect to a Person, a material adverse effect on the business, assets, financial condition or results of operations of such Person and its subsidiaries, taken as a whole;

"MB Balance Sheet" shall have the meaning given that term in Section 4.4;

"MB Business" means the operation, maintenance and marketing of the underground storage facility and related operations of shuttle, gathering and local delivery pipelines owned by TEPPCO located near Mont Belvieu in Chambers and Harris Counties, Texas for the transportation of Products to and from such underground storage facility and industries located along the Houston Ship Channel and at Baytown and Mont Belvieu, Texas;

"Milestone" shall have the meaning given that term in Section 6.1;

"Net EBITDA" means, for the MB Business, the amount of EBITDA, if any, in excess of the Base EBITDA during any period;

"Newco" shall have the meaning given that term in the recitals;

"Notice Month" shall have the meaning given that term in Section 8.3;

"Offer" shall have the meaning given that term in Section 8.6;

"Offer Period" shall have the meaning given that term in Section 8.6;

"Operating Budget" shall have the meaning given that term in Section 2.3. The initial Operating Budget for the calendar year beginning January 1, 2000 is attached as Exhibit F;

"Operating Procedures" means the Operating Procedures in the form attached as an exhibit to the Partnership Agreement and further includes the Accounting Procedures;

"Organizational Agreements" means the Partnership Agreement, the LLC Agreement and the agreements contemplated thereby;

"Parent" shall mean, with respect to TEPPCO, TEPPCO Partners, L.P., a Delaware limited partnership, and with respect to Dreyfus, S. A. Louis Dreyfus et Cie, a French societe anonyme.

"Partnership Agreement" means the Agreement of Limited Partnership of Newco, in the form attached as Exhibit G hereto;

"Party" means TEPPCO or Dreyfus, as the case may be, and "Parties" means TEPPCO and Dreyfus;

"Party Indemnitees" shall have the meaning given that term in Section 11.1;

"Permit" means any license, permit, concession, warrant, franchise or other governmental authorization or approval of any Governmental Authority;

"Permitted Encumbrances" means (a) Liens for current taxes and assessments not yet due, (b) inchoate mechanic and materialmen liens for construction in progress, (c) inchoate workmen, repairmen, warehousemen, customer, employee and carrier liens arising in the ordinary course of business, (d) other minor imperfections in title that do not affect marketability or use, (e) all validly existing restrictions, covenants, agreements, encumbrances, conditions, rights-of-way, easements, ordinances, mineral reservations, royalty reservations and other matters of record, if any, affecting all or any part of the TEPPCO Assets and (f) all of the terms and conditions of the instruments creating or evidencing the TEPPCO Assets;

"Person" means any individual, corporation, partnership, joint venture, association, limited liability company, joint stock company, trust, unincorporated organization, Governmental Authority or government (or agency or political subdivision thereof);

"Pricing Period" shall mean the trailing 12-month period ending on a date or month as specified in the Agreement;

"Proceeding" means any action, suit, claim, investigation, review or other judicial or administrative proceeding, at law or in equity, before any Governmental Authority;

"Products" means propane, normal butane, isobutane, natural gasoline and other commodities as mutually agreed between the Parties, and for which products the TEPPCO Assets has at the relevant time all necessary Permits;

"Records" means all agreements, documents, accounts, books, records and files relating to the TEPPCO Assets and the MB Business;

"Restricted Affiliate" shall have the meaning given that term in Section 10.2;

"Second Party" shall have the meaning given that term in Section 10.1(c);

"Start Date" means January 1, 2000;

"TEPPCO" shall have the meaning given that term in the preamble and any successor or assign permitted by this Agreement;

"TEPPCO Assets" means (i) all of the real property identified in pink or blue on the map labeled "Real Property at Mont Belvieu" attached to Exhibit H as Attachment 1 (the "Current Real Property") and the improvements (other than the TEPPCO Retained Assets) located thereon, (ii) equipment, personalty and other assets located at places other than on the Current Real Property which are exclusively used in the conduct of the MB Business in the ordinary course consistent with past practices and (iii) all of TEPPCO's right, title and interest in and to the real property interests, permits (to the extent such permits are assignable and are required to be held in the name of the owner of the TEPPCO Assets as opposed to the operator of the TEPPCO Assets), equipment, personalty, contracts and other assets that TEPPCO determines, in its reasonable discretion, as being necessary for the conduct of the MB Business in the ordinary course consistent with past practices (collectively, the "Necessary Assets"), but excluding the TEPPCO Retained Assets. The Parties recognize and agree that (a) the TEPPCO Assets may change from time to time and may be different at Closing than as of the date hereof, and may, for example, include, at Closing, all or a portion of the TEPPCO Retained Assets, there being no obligation, however, on the part of TEPPCO to include any of the TEPPCO Retained Assets in the TEPPCO Assets at Closing and (b) a portion of the TEPPCO Retained Assets are located on the Current Real Property;

"TEPPCO Retained Assets" means the real property interests, equipment, personalty, contracts and other assets described on Exhibit H and such other real property interests, permits (other than permits that are assignable and are required to be held in the name of the owner of the TEPPCO Assets as opposed to the operator of the TEPPCO Assets), equipment, personalty, contracts and other assets that TEPPCO determines, in its reasonable discretion, are not necessary for the conduct of the MB Business in the ordinary course consistent with past practices. The Parties recognize and agree that, absent mutual agreement by the

Parties, the Necessary Assets shall never be included within the TEPPCO Retained Assets;

"TEPPCO Storage Agreement" shall mean the Storage and Service Agreement in the form attached as Exhibit I hereto;

"Term" or "Term of this Agreement" shall mean the period beginning on the Start Date and ending on the termination of this Agreement and shall include any extensions of this Agreement;

"Territory" shall have the meaning given that term in Section 10.2; and

"Threshold" shall have the meaning given that term in Section 11.3.

1.2 OTHER DEFINITIONS AND PROVISIONS.

(a) As used in this Agreement, unless expressly stated otherwise, references to (a) "including" mean "including, without limitation", and the words "hereof", "herein", and "hereunder", and similar words, refer to this Agreement as a whole and not to any particular Article, provision, section or paragraph of this Agreement and (b) "or" mean "either or both". Unless otherwise specified, all references in this Agreement to Articles, Sections, paragraphs, Exhibits or Schedules are deemed references to the corresponding Articles, Sections, paragraphs, Exhibits or Schedules in this Agreement.

(b) Whenever a statement is qualified by the term "knowledge," "best knowledge" or similar term or phrase, it is intended to indicate the actual knowledge of the persons identified below gained through the exercise of such person's normal duties and without investigation or inquiry. As applied to TEPPCO, such terms or phrases shall be limited to the actual knowledge of Charles H. Leonard, David L. Langley and Kenneth A. Betts, and as applied to Dreyfus, such terms or phrases shall be limited to the actual knowledge of Timothy J. Stuart, William C. Rippe and Michael G. Dowling. If any such Person is no longer employed by their respective Party at the time in question, there shall be substituted the employee of such Party fulfilling the same job responsibilities as the Parties shall agree.

1.3 HEADINGS. The headings of the Articles and Sections of this Agreement and of the Schedules and Exhibits are included for convenience only and shall not be deemed to constitute part of this Agreement or to affect the construction or interpretation hereof or thereof.

ARTICLE II OPERATION OF TEPPCO ASSETS

2.1 STORAGE OPERATIONS.

(a) Beginning on the Start Date, TEPPCO shall use its commercially reasonable efforts to continue to operate, manage and maintain the TEPPCO Assets and to carry on the MB Business during the Term consistent with its past practices and in the manner provided in the Operating Procedures, except where the Parties may agree otherwise in writing. In the event the terms and provisions of the Operating Procedures conflict or are inconsistent with the terms and provisions of this Agreement, the terms and provisions of this Agreement shall control. The Parties recognize and agree that all direct, unallocated (except as provided in the attached Exhibit D) expenses of TEPPCO in operating the MB Business, including any extraordinary expenses or nonreoccurring expenses, such as Losses relating to environmental or litigation matters, shall be charged to the MB Business in accordance with TEPPCO's past practices and the Accounting Procedures. The Parties acknowledge and agree that TEPPCO retains ownership of the TEPPCO Assets and the MB Business during the Term of this Agreement, and will continue to operate such assets and business as it, in its discretion, deems appropriate except as otherwise specifically provided herein or as the Parties may otherwise agree in writing. Except as specifically set forth in this Agreement, no right, title or interest in the TEPPCO Assets or the MB Business is being conveyed to Dreyfus. Notwithstanding the foregoing, the operation of the MB Business and the TEPPCO Assets during the Term of this Agreement shall be for the economic benefit of both TEPPCO and Dreyfus as and to the extent herein provided.

(b) Simultaneously with the execution of this Agreement, TEPPCO is entering into the Dreyfus Storage Agreement. Additionally, TEPPCO shall be entitled to utilize during the Term the storage capacity and services specified in, and in accordance with the terms and provisions of, the TEPPCO Storage Agreement although the parties are not entering into such agreement at this time.

(c) Notwithstanding any other provision in this Agreement or the Operating Procedures, Dreyfus recognizes and agrees that deliveries to the mainline system of TEPPCO and its Affiliates, including all other operational activities and facilities required to implement such deliveries, take precedence over all other activities of the MB Business or relating to the TEPPCO Assets, and that TEPPCO shall be authorized to act accordingly in operating the MB Business without liability to Dreyfus even if such action shall benefit TEPPCO to the detriment of Dreyfus or any other Person.

2.2 MARKETING AND DEVELOPMENT ACTIVITIES.

(a) Dreyfus shall have the right (i) to carry out the marketing of the TEPPCO Assets and other assets that may become part of the MB Business and manage the personnel involved in those marketing activities, and (ii) to make proposals

to TEPPCO for Discretionary Capital Expenditures and the development and use of the TEPPCO Assets and such other assets as may become part of the MB Business, in each case with the purpose of increasing the EBITDA of the MB Business. Drevfus will use commercially reasonable efforts in carrying out such activities. Dreyfus's activities shall include soliciting new users of the TEPPCO Assets or increased use by existing customers and investigating and proposing capital projects for the TEPPCO Assets and the MB Business. At least 120 days prior to the beginning of each Fiscal Year, Dreyfus will provide TEPPCO with a forecast of revenue for such Fiscal Year starting with the 2001 Fiscal Year. On or before January 15, April 15, July 15 and October 15 of each Fiscal Year (beginning with April 15, 2000), Dreyfus will provide a report to TEPPCO on Dreyfus's activities for the three month period ending on the last day of the immediately preceding month including, among other things, the status of contracts entered into during such period, potential contracts being explored, performance for that period and revisions, if any, to the revenue forecast for the Fiscal Year in which such three-month period falls. TEPPCO will keep Dreyfus advised of any storage contracts or renewals thereof entered into by TEPPCO and will seek Dreyfus' advice with respect thereto. Dreyfus will commit the equivalent of at least one full-time person to the marketing activities to be carried out under this Section 2.2(a), and TEPPCO will commit the equivalent of at least one part-time person to aid Dreyfus in those marketing activities under Dreyfus' direction. The cost of the compensation and benefits of such employees shall not be included in the expenses of the MB Business but shall be borne by Dreyfus or TEPPCO respectively as the employer of that person. TEPPCO and Dreyfus will agree on whether the cost of the compensation and benefits of any additional employees of either party who may be engaged in the marketing activities shall be included in the expenses of the MB Business.

(b) The Parties understand and agree that Dreyfus shall not act as TEPPCO's agent in carrying out its marketing responsibilities. All storage, service and other agreements relating to the TEPPCO Assets and the MB Business shall be subject to the approval of, and signed by and in the name of, TEPPCO or one of its Affiliates. Neither Party shall have the authority, nor shall it attempt, to bind the other Party in any regard. Dreyfus will keep TEPPCO informed of its marketing activities and provide to TEPPCO such information as it may reasonably request with respect to customers and potential customers of the TEPPCO Assets. TEPPCO will provide all reasonable assistance in Dreyfus' marketing and sales efforts and shall make available appropriate authorized representatives to execute any storage agreements developed by Dreyfus. All such contracts shall become the property of TEPPCO and shall be included in the definition of "TEPPCO Assets".

2.3 BUDGETS. At least 30 days prior to the end of each Fiscal Year during the Term, TEPPCO, as owner of the TEPPCO Assets and the MB Business, shall prepare and disclose to Dreyfus a proposed operating budget (the "Operating Budget") for operating the TEPPCO Assets and the MB Business for the immediately following Fiscal Year. TEPPCO will also prepare and disclose to Dreyfus a proposed budget for any capital expenditure or capital project ("Capital Budget", collectively, with the Operating Budget, the "Budgets"). Each Operating Budget and each Capital Budget for

Mandatory Capital Expenditures shall be submitted to Dreyfus for its review and comment, however, such Budgets as presented by TEPPCO shall be final and binding for the subject Fiscal Year. Each Capital Budget for Discretionary Capital Expenditures shall be submitted to Dreyfus for approval, and upon approval, each Party shall be obligated to make such capital and other contributions as may be called for by the approved Budget. The Operating Budget for the Fiscal Year beginning January 1, 2000 has been approved by both Parties and is attached as Exhibit F.

2.4 COMMUNICATION. During the Term, TEPPCO shall use its commercially reasonable efforts to (i) periodically advise Dreyfus regarding the status of the operation and maintenance of the TEPPCO Assets and the MB Business, except for marketing activities carried out by Dreyfus, and (ii) respond to inquiries and information requests for Dreyfus with respect to information available to TEPPCO. In this regard, TEPPCO agrees that it will obtain Dreyfus' prior approval of any financing to be entered into relating to Discretionary Capital Expenditures.

2.5 CAPITAL EXPENDITURES. During the Term:

(a) Discretionary Capital Expenditures approved by the Parties shall be funded, first by Dreyfus up to an aggregate of \$5,000,000, and then by each of the Parties paying 50% of all such amounts.

(b) Mandatory Capital Expenditures that do not relate to improvements, additions, expansions or other modifications resulting from Discretionary Capital Expenditures shall be funded by TEPPCO.

(c) Mandatory Capital Expenditures that do relate to improvements, additions, expansions or other modifications resulting from Discretionary Capital Expenditures shall be funded 50% by each Party.

All improvements, additions, expansions and other modifications to the TEPPCO Assets, regardless of how funded, shall become the property of TEPPCO and a part of the TEPPCO Assets subject to this Agreement.

2.6 TEPPCO EXCULPATION. NEITHER TEPPCO (IN ITS CAPACITY AS OWNER AND OPERATOR OF THE TEPPCO ASSETS OR THE MB BUSINESS), NOR ITS AFFILIATES, NOR ANY OWNER, OFFICER, DIRECTOR, PARTNER, MEMBER, EMPLOYEE OR AGENT OF SUCH PARTIES, SHALL BE LIABLE, RESPONSIBLE OR ACCOUNTABLE IN DAMAGES OR OTHERWISE TO DREYFUS FOR ANY ACTION TAKEN OR FAILURE TO ACT (EVEN IF SUCH ACTION OR FAILURE TO ACT CONSTITUTED THE NEGLIGENCE OF SUCH PERSON) IN THE OPERATION OF THE TEPPCO ASSETS OR THE CONDUCT OF THE MB BUSINESS UNLESS SUCH ACT OR OMISSION WAS PERFORMED OR OMITTED FRAUDULENTLY OR CONSTITUTED GROSS NEGLIGENCE OR WILLFUL MISCONDUCT. NOTWITHSTANDING THE FOREGOING, TEPPCO SHALL NOT BE RELIEVED OF ITS OBLIGATIONS HEREUNDER RELATING TO (I) PAYMENTS TO DREYFUS UNDER SECTION 3.2 OR OF DREYFUS' INVESTMENT UPON TERMINATION OF THIS AGREEMENT AS

HEREIN PROVIDED, (II) TEPPCO'S OBLIGATIONS UNDER ARTICLE XI HEREOF, OR (III) TEPPCO'S RESPONSIBILITIES AS SPECIFIED IN THE DREYFUS STORAGE AGREEMENT.

2.7 DREYFUS EXCULPATION. NEITHER DREYFUS (IN ITS MARKETING CAPACITY), NOR ITS AFFILIATES, NOR ANY OWNER, OFFICER, DIRECTOR, PARTNER, MEMBER, EMPLOYEE OR AGENT OF SUCH PARTIES, SHALL BE LIABLE, RESPONSIBLE OR ACCOUNTABLE IN DAMAGES OR OTHERWISE TO TEPPCO FOR ANY ACTION TAKEN OR FAILURE TO ACT (EVEN IF SUCH ACTION OR FAILURE TO ACT CONSTITUTED THE NEGLIGENCE OF SUCH PERSON) IN THE PERFORMANCE OF ITS OBLIGATIONS IN CONNECTION WITH THE CONDUCT OF THE MB BUSINESS UNLESS SUCH ACT OR OMISSION WAS PERFORMED OR OMITTED FRAUDULENTLY OR CONSTITUTED GROSS NEGLIGENCE OR WILLFUL MISCONDUCT. NOTWITHSTANDING THE FOREGOING, DREYFUS SHALL NOT BE RELIEVED OF ITS OBLIGATIONS (I) UNDER ARTICLE XI HEREOF, OR (II) AS SPECIFIED IN THE DREYFUS STORAGE AGREEMENT.

ARTICLE III FINANCIAL OPERATIONS AND PAYMENTS

3.1 FINANCIAL RESULTS.

(a) Unaudited financial statements for the MB Business shall be prepared by or at the direction of TEPPCO on a quarterly and annual basis and provided to Dreyfus within 45 days of each calendar quarter and 90 days following the end of each Fiscal Year. Such statements shall be prepared in accordance with GAAP and shall be consistent, to the extent applicable, with the calculations of EBITDA set forth on Exhibit D.

(b) Within 90 days following the Development Period, TEPPCO shall prepare and deliver to Dreyfus TEPPCO's determination of whether the MB Business met both of the requirements of the Milestone specified in Section 6.1 (i) and (ii), together with the financial information and calculations on which such determination is based. If Dreyfus does not agree with such determination, the procedures of Section 3.5 shall be followed to obtain a final determination.

3.2 PAYMENTS TO DREYFUS.

(a) For each Fiscal Year during the Term:

(i) All Base EBITDA amounts shall be retained by

TEPPC0.

(ii) Thereafter all Not EPITDA shall be divided

(ii) Thereafter, all Net EBITDA shall be divided equally between TEPPCO and Dreyfus and 50% of Net EBITDA shall be paid to Dreyfus.

(b) Payments to Dreyfus pursuant to Section 3.2(a) shall be made on an annual basis and shall be made within ninety (90) days following the end of the applicable Fiscal Year. If approved by the Parties and following funding by Dreyfus of

\$5,000,000 in Discretionary Capital Expenditures, Net EBITDA amounts may be retained by TEPPCO to fund additional Discretionary Capital Expenditures.

(c) In the event this Agreement is terminated during any Fiscal Year, amounts of Base EBITDA and Net EBITDA shall nevertheless be retained by TEPPCO or paid to Dreyfus pursuant to Section 3.2(a) in amounts that are prorated to the portion of the Fiscal Year prior to such termination. Accordingly, the Base EBITDA amount for such partial Fiscal Year shall be determined by multiplying \$7.3 million by a fraction the numerator of which is the number of days in the Fiscal Year prior to such termination and the denominator of which is 365. This amount shall be retained by TEPPCO, and the remaining EBITDA for that partial Fiscal Year shall be divided equally between TEPPCO and Dreyfus. Payment of such amount to Dreyfus shall be in accordance with Section 3.2(b).

3.3 FUNDING OF MB BUSINESS. Dreyfus will not be responsible for funding any of the operating or other costs of the MB Business during the Term except for Discretionary Capital Expenditures in accordance with Section 2.5(a) and, to the extent provided in Section 2.5(c), Mandatory Capital Expenditures, although the Parties recognize that operating and other costs may increase in the future, thereby impacting EBITDA and possibly the resulting payments to Dreyfus.

3.4 ACCOUNTS, BOOKS AND RECORDS.

(a) TEPPCO shall at all times during the Term maintain true and correct accounts, books and records for the MB Business as if it were a separate operating division of TEPPCO in accordance with GAAP and sufficient to determine the financial position and results of operations for the MB Business and the amounts, if any, of EBITDA available for retention by or payment to the Parties.

(b) Upon notice, the accounts, books and records referred to in Section 3.4(a) shall be made available for inspection, copying and audit by Dreyfus and/or its representatives (at Dreyfus' sole cost and expense and no more frequently than once every calendar quarter) at the offices of TEPPCO during normal business hours. Dreyfus and its representatives shall make every effort to conduct such reviews in a manner as to cause the least disruption of TEPPCO's normal business operations.

3.5 EXCEPTIONS TO FINANCIAL REPORTS. If Dreyfus shall take exception to any item or items included in the financial reports and determinations provided by TEPPCO pursuant to Section 3.1 hereof (collectively, the "reports"), Dreyfus shall notify TEPPCO in writing within 30 days after receipt of any such report, setting forth in such notice the specific items to which exception is taken and the specific charges or credits which should have been made or allowed; and with respect to such exceptions as are justified, adjustments (including payments or refunds) shall be promptly made. If Dreyfus shall fail to give written notice of such exceptions prior to the expiration of such 30 day period, then the reports for such period as originally rendered by TEPPCO shall be deemed to be correct as rendered, no adjustment shall be made, and except as

provided below in this Section, Dreyfus shall no longer be entitled to take exception or otherwise object to such reports, or to audit TEPPCO's accounts, books and records with respect to such period, and, if applicable, the determination prepared pursuant to Section 3.1(b) shall be final for purposes of Section 6.2. Any exceptions as to which written notice is not given within such 30 day period shall be waived. In the event an exception is timely made by Dreyfus and should the parties disagree on any such exception, such disputes shall be handled in accordance with the provisions of Article XIV hereof. If the dispute involves the determination prepared pursuant to Section 3.1(b), resolution pursuant to Article XIV shall be deemed a final determination for purposes of Section 6.2. All time periods set forth in this Agreement shall be extended as necessary to comply with such procedures. Notwithstanding the foregoing, in the event of fraud or mutual mistake, Dreyfus shall be entitled to audit such accounts, books and records subsequent to the above stated period or periods and to make exception to any such report; provided, however, such extended right to audit and to make exceptions shall terminate 180 days following the end of the Term, and thereafter, all such reports, accounts, books and records shall be deemed to be final. Additionally, Dreyfus shall be entitled to inspect and copy such accounts, books and records in accordance with, and for the period specified in, Section 9.2. The parties recognize that the information contained in such reports shall be considered confidential information and treated as such in accordance with the provisions of Section 9.3 hereof.

ARTICLE IV REPRESENTATIONS AND WARRANTIES OF TEPPCO

TEPPCO hereby represents and warrants to, and agrees and covenants with, Dreyfus as follows:

4.1 CORPORATE MATTERS; ASSET OWNERSHIP. TEPPCO is a limited partnership duly organized, validly existing and in good standing under the laws of the State of Delaware, having all requisite limited partnership power and authority to enter into this Agreement and the Organizational Agreements and to perform its obligations hereunder and thereunder.

4.2 VALIDITY OF AGREEMENT; NO CONFLICT. This Agreement has been duly authorized, executed and delivered by TEPPCO and is a legal, valid and binding obligation of TEPPCO, enforceable against it in accordance with its terms, except as enforceability may be limited by applicable bankruptcy, insolvency, reorganization, moratorium or similar laws from time to time in effect that affect creditors' rights generally and by legal and equitable limitations on the availability of specific remedies.

4.3 LIABILITIES OF THE MB BUSINESS. Except as set forth on Exhibit J-1, to the knowledge of TEPPCO, there are no other liabilities or obligations of the MB Business which would have a material adverse effect on the MB Business.

4.4 FINANCIAL STATEMENTS. Attached as Schedule 4.4 hereto is a copy of the unaudited pro forma balance sheet for the MB Business, as of December 31, 1999 (the "MB Balance Sheet") and the unaudited pro forma income statement for the MB Business for the 12 months then ended, which have been prepared in accordance with generally accepted accounting principles consistently applied and fairly present the consolidated financial condition for the MB Business as of the date thereof and the results of operations for the respective periods set forth therein, subject to normal interim adjustments, footnotes and other presentation items. The MB Balance Sheet has been adjusted to (i) reflect certain correcting adjustments and reclassifications described therein, and (ii) exclude the TEPPCO Retained Assets.

4.5 FINDER'S FEE. No investment banker, broker, finder or other Person is entitled to any brokerage or finder's fee or similar commission in respect of the transactions contemplated by this Agreement. TEPPCO agrees to indemnify and hold Dreyfus and its Affiliates harmless from and against any and all claims, liabilities and obligations with respect to any such fees, commissions or expenses asserted by any such Person on the basis of any act, statement, agreement or commitment alleged to have been made by TEPPCO or any of its Affiliates with respect thereto.

ARTICLE V REPRESENTATIONS AND WARRANTIES OF DREYFUS

Dreyfus hereby represents and warrants to, and agrees and covenants with, TEPPCO as follows:

5.1 CORPORATE MATTERS; ASSET OWNERSHIP. Dreyfus is a corporation duly organized, validly existing and in good standing under the laws of the State of Delaware, having all requisite corporate power and authority to enter into this Agreement and the Organizational Agreements, as the case may be, and to perform its respective obligations hereunder and thereunder.

5.2 VALIDITY OF AGREEMENT; NO CONFLICT. This Agreement has been duly authorized, executed and delivered by Dreyfus and is a legal, valid and binding obligation of Dreyfus enforceable against it in accordance with its terms, except as enforceability may be limited by applicable bankruptcy, insolvency, reorganization, moratorium or similar laws from time to time in effect that affect creditors' rights generally and by legal and equitable limitations on the availability of specific remedies.

5.3 FINANCIAL STATEMENTS. Attached as Schedule 5.3 hereto is a copy of the audited consolidated balance sheet of Dreyfus as of May 31, 1999 and the audited consolidated income statement of Dreyfus for the 12 months then ended, which have been prepared in accordance with generally accepted accounting principles consistently applied and fairly present the consolidated financial condition of Dreyfus as of the date thereof and the results of operations for the respective periods set forth therein, subject to normal interim adjustments, footnotes and other presentation items.

5.4 FINDER'S FEE. No investment banker, broker, finder or other Person is entitled to any brokerage or finder's fee or similar commission in respect of the transactions contemplated by this Agreement. Dreyfus agrees to indemnify and hold TEPPCO and its Affiliates harmless from and against any and all claims, liabilities and obligations with respect to any such fees, commissions or expenses asserted by any such Person on the basis of any act, statement, agreement or commitment alleged to have been made by Dreyfus or any of its Affiliates with respect thereto.

ARTICLE VI TRANSFER OF TEPPCO ASSETS OR RIGHT TO USE

6.1 PERFORMANCE MILESTONE. If EBITDA exceeds:

(i) \$8.7 million for the Fiscal Year ending on

December 31, 2002; and

(ii) \$23.5 million for the Development Period (the provisions of clauses (i) and (ii) are herein collectively referred to as the "Milestone"), then

TEPPCO shall contribute the TEPPCO Assets and the MB Business to Newco for no additional consideration. Such contribution shall be through the execution and delivery by TEPPCO of the Conveyance Document, and TEPPCO shall have no further obligation with respect thereto except as may be specifically set forth herein or therein. Solely for the purpose of determining whether the Milestone has been met, EBITDA, for the period in question, shall be calculated without regard to amounts that would be classified in accordance with GAAP as extraordinary or nonrecurring items.

6.2 FAILURE TO MEET MILESTONE. If either of the requirements of the Milestone set out in Sections 6.1(i) and (ii) is not met as finally determined pursuant to Sections 3.1 and 3.5, TEPPCO shall have no obligation to transfer the TEPPCO Assets or the MB Business, neither Party shall be obligated to close the transactions contemplated by this Agreement and either Party shall have the option to terminate this Agreement within the 30 day period following the final determination of the failure to meet the Milestone. If neither Party exercises the option to terminate, the Agreement shall automatically terminate on June 30,2003. In the event of termination fee as provided in Section 8.2.

6.3 CLOSING TIME AND LOCATION FOR TRANSFER. Subject to the satisfaction of the conditions set forth in this Agreement, the contribution and transfer of the TEPPCO Assets and the MB Business to Newco through the execution and delivery of the Conveyance Document and the assumption by Newco of the Assumed Liabilities (the "Closing") shall take place at the offices of Fulbright & Jaworski L.L.P., 1301 McKinney Street, Houston, Texas 77010, or at such other place as the Parties may mutually agree in writing, at 9:00 a.m. on the later of (i) March 31, 2003, and (ii) such other date as may be mutually agreed to by the Parties in writing, to be effective as of 12:01 a.m. on

January 1, 2003 (the "Effective Time"). The date on which the Effective Time occurs is referred to herein as the "Effective Date".

6.4 FORMATION OF NEWCO. On or before the Closing, TEPPCO and Dreyfus shall execute the LLC Agreement and the Partnership Agreement. Following the formation of Newco, TEPPCO and Dreyfus will cause Newco to make reasonable arrangements for qualifying Newco to do business in the necessary jurisdictions, securing insurance and necessary Permits, establishing bank accounts and such other action as would be appropriate in order for Newco to become the owner of the TEPPCO Assets in the manner herein contemplated immediately following the Effective Time.

6.5 CLOSING. In accordance with the terms and subject to satisfaction of the conditions set forth in this Agreement, TEPPCO and Dreyfus, as the case may be, shall undertake, or shall cause Newco to undertake, the following actions at the Closing:

(a) TEPPCO will execute, acknowledge and deliver to Newco the Conveyance Document and other instruments of transfer, assignment and conveyance as shall be necessary to vest in Newco the right, title and interest of TEPPCO and its Affiliates in and to the TEPPCO Assets and the MB Business herein and therein contemplated, subject only to the representations contained in Exhibit K.

(b) Dreyfus shall contribute to Newco the aggregate Dreyfus Investment made through the Closing.

(c) Each party will have established pursuant to the Partnership Agreement a Capital Account (as that term is defined in the Partnership Agreement) which will reflect, in the case of TEPPCO, the net book value of the TEPPCO Assets for GAAP purposes as of the Effective Date (exclusive of any basis attributable to capital expenditures made by Dreyfus during the Term), and, in the case of Dreyfus, the Dreyfus Investment made during the Term.

(d) Newco will assume and agree to timely discharge, pay or perform (as the case may be) the Assumed Liabilities as of the Effective Date and shall execute and deliver, as appropriate, the Conveyance Document.

(e) Newco and TEPPCO or its Affiliate shall execute and deliver the TEPPCO Storage Agreement. The Dreyfus Storage Agreement shall be assigned to and assumed by Newco.

(f) Simultaneously with the consummation of the transfer of the TEPPCO Assets and the MB Business, each of the Parties, through its officers, agents and employees, shall take all action necessary and appropriate to put Newco into full possession and enjoyment of all of such assets.

(g) Upon consummation of the Closing, this Agreement shall terminate as of the Effective Time except as otherwise specifically provided herein.

6.6 INABILITY TO TRANSFER ASSETS. In the event the Milestone has been met as herein provided but the TEPPCO Assets and the MB Business have not been transferred to Newco, the Parties will use all commercially reasonable efforts to:

(a) Continue the current structure and operations through December 31, 2003 so that the economic benefit as specified in Section 3.2 will continue for the benefit of the Parties. Unless the Parties agree otherwise in writing, the terms and provisions of this Agreement shall continue in effect, and the Parties shall continue to be bound thereby, during any extension of this Agreement.

(b) Resolve or remove the impediment that prohibits the transfer of such assets, although neither Party will be obligated to incur any liability or obligation or expend funds in conjunction with such efforts.

In the event the transfer of the TEPPCO Assets and the MB Business has still not occurred by December 31, 2003, this Agreement may be terminated by either Party giving to the other Party written notice of its election to terminate within 30 days following December 31, 2003. If no notice of termination is given within such 30 day period, this Agreement shall be automatically extended through December 31 of 2004 and for each succeeding year thereafter unless and until either Party gives written notice of its election to terminate within 30 days following December 31 of any such extended annual period, in which event, this Agreement shall terminate. In the event such notice of termination is given, the provisions of Section 8.5(b) shall apply.

6.7 FURTHER ASSURANCE BY THE PARTIES. Each of the Parties to this Agreement, at any time or times, on and after the Effective Date, shall, without further consideration and at its sole expense, execute, acknowledge and deliver any further bills of sales, assignments, conveyances and other assurances, documents and instruments of transfer reasonably requested by Newco, and shall take any other action consistent with the terms of this Agreement that may be reasonably requested by Newco, for the purpose of assigning, transferring, granting, conveying and confirming to Newco, or reducing to its possession, any or all of the TEPPCO Assets as herein contemplated and for effectuating the transfer of operational control of the MB Business. Each of the Parties shall, after the Closing, also furnish Newco with such information and documents in such Party's possession or under such Party's control or which any of the Parties can execute or cause to be executed, as will enable Newco to prosecute any and all pending claims, applications and the like which may be assigned hereunder.

6.8 THIRD PARTY CONSENTS. Each of TEPPCO and Dreyfus shall use its commercially reasonable efforts to obtain the consents of third parties as are necessary for the conveyance of the TEPPCO Assets or rights thereto and the consummation of the transactions, all as herein contemplated. To the extent that any of the TEPPCO Assets or rights thereto are not so transferrable by the terms thereof or consents to the assignment thereof cannot be obtained and the respective condition to closing has been waived by Dreyfus, subject to the provisions of Section 6.6, such assets shall be held by TEPPCO in trust for Newco and shall be performed by Newco in the name of TEPPCO

and all benefits and obligations derived thereunder shall be for the account of Newco; provided that, where entitlement of Newco to those TEPPCO Assets or rights is not recognized by any third party, TEPPCO shall, at the request and expense of Newco, enforce in a reasonable manner, for the account of Newco, any and all rights of TEPPCO against the third party. All reasonable costs and expenses incurred by TEPPCO with respect to the performance or enforcement of the above-described TEPPCO Assets and rights thereto shall be paid or reimbursed by Newco, and Newco shall indemnify TEPPCO and its Party Indemnitees from any and all Losses that may arise or be incurred as a result of TEPPCO's undertaking the foregoing actions.

ARTICLE VII CONDITIONS TO CLOSING

7.1 CONDITIONS TO OBLIGATION OF DREYFUS. The obligation of Dreyfus to enter into the agreements and to consummate the transactions contemplated hereby at the Closing is, at the option of Dreyfus, subject to the satisfaction on or before the date of Closing of the conditions set forth below, any of which may be waived by Dreyfus in writing.

(a) Satisfaction of Milestone. The Milestone specified in Section 6.1 shall have been met.

(b) Representations, Warranties and Covenant. TEPPCO shall have performed, satisfied, and complied with, in all material respects, all covenants and agreements required by this Agreement to be performed, satisfied, or complied with by it on or before the date of Closing. All representations and warranties of TEPPCO attached as Exhibit K to this Agreement or in any certificate, delivered to Dreyfus by or on behalf of TEPPCO under this Agreement shall be true and correct, in all material respects, on and as of the date of Closing with the same force and effect as though they had been made on such date. Dreyfus shall have received a certificate, dated as of the date of Closing, signed by the President or Vice President of the general partner of TEPPCO certifying the matters set forth in this subsection.

(c) Good Standing. TEPPCO shall have delivered to Dreyfus certificates issued by appropriate Governmental Authorities evidencing the good standing and existence of TEPPCO in Texas as of a date not more than ten calendar days prior to the date of Closing.

(d) Approvals. All approvals of Governmental Authorities and other third parties required for the consummation of the transactions contemplated by this Agreement shall have been obtained.

(e) Closing Actions Taken. The Parties and Newco shall have executed, acknowledged and delivered the documents, and shall have taken the respective actions, called for by Sections 6.4 and 6.5.

(f) Bring-down Schedules. As of or prior to Closing, TEPPCO shall have provided Dreyfus: (i) Exhibit J-2 setting forth, as to the knowledge of TEPPCO, the Assumed Liabilities as of the month end immediately preceding the month prior to the Effective Date, which shall thereafter be deemed a part of this Agreement, and the information set forth therein shall not include any indebtedness for borrowed money except for indebtedness for Discretionary Capital Expenditures previously approved by Dreyfus, and (ii) Exhibit L setting forth the financial statements called for by the representations and warranties of Exhibit K which shall thereafter be deemed a part of this Agreement and the information contained in those financial statements shall not demonstrate a material adverse change in the financial statements provided to Dreyfus pursuant to Section 4.4.

7.2 CONDITIONS TO OBLIGATION OF TEPPCO. The obligation of TEPPCO to enter into the agreements and to consummate the transactions contemplated hereby at the Closing and to transfer the TEPPCO Assets as contemplated hereby is, at the option of TEPPCO, subject to the satisfaction on or before the date of Closing of the conditions set forth below, any of which may be waived by TEPPCO in writing.

(a) Satisfaction of Milestone. The Milestone specified in Section 6.1 shall have been met.

(b) Representations, Warranties and Covenants. Dreyfus shall have performed, satisfied, and complied with, in all material respects, all covenants and agreements required by this Agreement to be performed, satisfied, or complied with by it on or before the date of Closing. All representations and warranties of Dreyfus attached as Exhibit M to this Agreement or in any certificate, delivered to TEPPCO by or on behalf of Dreyfus under this Agreement shall be true and correct, in all material respects, on and as of the date of Closing with the same force and effect as though they had been made on such date. TEPPCO shall have received a certificate, dated as of the date of Closing, signed by the President or Vice President of Dreyfus certifying the matters set forth in this subsection.

(c) Good Standing. Dreyfus shall have delivered to TEPPCO certificates issued by appropriate Governmental Authorities evidencing the good standing and existence of Dreyfus, and as of a date not more than ten calendar days prior to the date of Closing, in the state in which it was organized and in Texas. To the extent provided for under applicable law, Dreyfus shall also have delivered to TEPPCO certificates or other writings issued by appropriate Governmental Authorities in such states with respect to Dreyfus evidencing that all applicable state franchise Taxes have been paid.

(d) Approvals. All approvals of Governmental Authorities and other third parties required for the consummation of the transactions contemplated by this Agreement shall have been obtained.

(e) Closing Actions Taken. The Parties and Newco shall have executed, acknowledged and delivered the documents, and shall have taken the respective actions, called for by Sections 6.4 and 6.5.

(f) Bring-down Schedules. As of or prior to Closing, Dreyfus shall have provided to TEPPCO Exhibit N setting forth the financial statements called for by the representations and warranties of Exhibit M which shall thereafter be deemed to be a part of this Agreement and the information contained in those financial statements shall not demonstrate a material adverse change in the financial condition or results of operations of Dreyfus since the financial statements provided to TEPPCO pursuant to Section 5.3.

ARTICLE VIII TRANSFER OF CONTRACT RIGHTS

8.1 TRANSFER OF CONTRACT RIGHTS PROHIBITED.

(a) During the Term, Dreyfus and TEPPCO agree that they shall not transfer, assign or otherwise convey or attempt to transfer, assign or otherwise convey, to any third party all or any portion of their respective rights, duties, obligations and other interests in and to this Agreement, any of the other agreements executed in connection herewith, the TEPPCO Assets or the MB Business (collectively with respect to each Party, as such rights, duties, obligations and other interests may exist from time to time, and regardless of whether held by the Party and/or its permitted Affiliates or assignees, the "Contract Rights"), except with the written consent of the other Party or as otherwise provided in this Agreement. Any such transfer, assignment or conveyance or attempted transfer, assignment or conveyance shall be void and of no force and effect. The Parties recognize that a sustained period of operation of the MB Business as herein contemplated is necessary to accurately determine the results and prospects for such business. Accordingly, the Parties agree that the foregoing prohibition on transfer of their respective Contract Rights in such agreements is reasonable for the Term of this Agreement.

(b) Notwithstanding the foregoing, each Party may transfer its Contract Rights in its entirety, but not in part, to an Affiliate that is wholly owned, directly or indirectly, by its Parent; provided, however, the Contract Rights remain subject to this Agreement and such Affiliate agrees in writing to be bound by this Agreement.

8.2 FAILURE TO MEET MILESTONE. If this Agreement is terminated pursuant to Section 6.2, TEPPCO shall pay to Dreyfus a termination fee equal to the aggregate capital expenditures by Dreyfus and its Affiliates in the TEPPCO Assets (the "Dreyfus Investment") as of the last date of the Term.

8.3 CHANGE OF CONTROL.

(a) If a Dreyfus Change of Control occurs, TEPPCO shall have the option to terminate this Agreement upon (i) the giving of a written notice of termination to Dreyfus within 30 days of the effective date of Dreyfus' notice of consummation given pursuant to Section 8.3(b), and (ii) the payment to Dreyfus of a termination fee which shall be the greater of (x) 50% of the Net EBITDA for the Pricing Period immediately preceding the Notice Month times 10.0 and (y) the Dreyfus Investment as of the end of such Pricing Period. "Notice Month" means the month during which the Dreyfus Change of Control occurs with respect to Dreyfus or its applicable Affiliate.

(b) Dreyfus shall keep TEPPCO promptly advised (subject to appropriate confidentiality arrangements) of any proposed or threatened Dreyfus Change of Control, whether voluntary or involuntary, within its knowledge, and shall provide written notice within one Business Day of the consummation thereof.

8.4 OPERATIONAL DISPUTE. In the event the Parties are unable to agree on a written proposal by either Party for a Discretionary Capital Expenditure with an estimated cost equal to or greater than \$1,000,000 by a date 180 days after the date on which such proposal was made, then either Party may terminate this Agreement by giving written termination notice no later than 30 days after the last day of such 180 day period or a Party's right to exercise the termination option pursuant to this Section 8.4 is waived with respect to the Discretionary Capital Expenditure proposal that is the subject of the disagreement. Upon the giving of such termination notice, the provisions of Section 8.5 shall apply.

8.5 CERTAIN TERMINATION PAYMENTS.

(a) Should either Party exercise the termination option of Section 8.4 or 10.1, it shall give written notice of such exercise and the following provisions shall apply:

(i) If these termination provisions are instituted by Dreyfus pursuant to Section 8.4 or Dreyfus is the First Party in Section 10.1, TEPPCO shall pay to Dreyfus a termination fee in an amount equal to the greater of: (x) 50% of the Net EBITDA for the Pricing Period ending with the month immediately preceding the month in which the written notice is given exercising the termination option times 7.0 and (y) the Dreyfus Investment at the end of such Pricing Period.

(ii) If these termination provisions are instituted by TEPPCO pursuant to Section 8.4 or TEPPCO is the First Party in Section 10.1, TEPPCO shall pay to Dreyfus a termination fee in an amount equal to the greater of: (x) 50% of the Net EBITDA for such Pricing Period times 13.0 and (y) the Dreyfus Investment at the end of such Pricing Period.

(b) In the event of termination of this Agreement pursuant to Section 6.6, TEPPCO shall pay to Dreyfus a termination fee equal to:

(i) If the termination provisions of Section 6.6 are instituted by Dreyfus, TEPPCO shall pay to Dreyfus a termination fee in an amount equal to the greater of: (x) 50% of the Net EBITDA for the Pricing Period that coincides with the most recently ended Fiscal Year times 7.0, and (y) the Dreyfus Investment at the end of such Pricing Period.

(ii) If the termination provisions of Section 6.6 are instituted by TEPPCO, TEPPCO shall pay to Dreyfus a termination fee in an amount equal to the greater of: (x) 50% of the Net EBITDA for such Pricing Period times 13.0, and (y) the Dreyfus Investment at the end of such Pricing Period.

(c) In the event of termination of this Agreement pursuant to Section 12.1(b) or (c), TEPPCO shall pay to Dreyfus a termination fee equal to:

(i) If the termination is pursuant to Section 12.1(b), TEPPCO shall pay to Dreyfus a termination fee in an amount equal to the greater of: (x) 50% of the Net EBITDA for the Pricing Period ending with the month immediately preceding the thirty day period specified in such section times 13.0, and (y) the Dreyfus Investment at the end of such Pricing Period.

(ii) If the termination is pursuant to Section 12.1(c), TEPPCO shall pay to Dreyfus a termination fee in an amount equal to the greater of: (x) 50% of the Net EBITDA for the Pricing Period ending with the month immediately preceding the thirty day period specified in such section times 7.0, and (y) the Dreyfus Investment at the end of such Pricing Period.

8.6 RIGHT OF FIRST REFUSAL. If, during the Term, TEPPCO intends to sell (i) any of the TEPPCO Assets other than assets in any transaction having a book value of \$5,000,000 or less, or (ii) its ownership interest in any Affiliate of which more than 75% of such entity's EBITDA for the preceding 12 month period is derived from the MB Business (each a "Disposition"), such Disposition may be made only if TEPPCO receives with respect thereto a bona fide binding written proposal for the acquisition of such assets (an "Acquisition Proposal"), and then only in compliance with the following procedures:

(a) Upon receipt of an Acquisition Proposal, TEPPCO shall offer, by written notice (the "Offer") to Dreyfus, to sell the assets to Dreyfus on the terms (including price) specified in the Acquisition Proposal pursuant to the terms of this Section 8.6. Such Offer shall contain a description of and a copy of the Acquisition Proposal subject to appropriate confidentiality provisions. In addition, TEPPCO shall provide to Dreyfus all other information with respect to the Acquisition Proposal and the proposed transferee reasonably requested by Dreyfus in order for it to evaluate the Acquisition Proposal and to verify the bona fide nature thereof.

(b) Dreyfus shall have the right, to be exercised by written notice (the "Acceptance") from Dreyfus to TEPPCO on or before the 30th day following receipt of

the Offer (the "Offer Period"), to elect to purchase such assets pursuant to the terms of the Offer. Should Dreyfus elect not to exercise the option, it shall have the option to terminate this Agreement upon the consummation of a sale of such assets to a third party if Dreyfus gives written notice to TEPPCO of such election to terminate on or before the end of such 30 day period. If Dreyfus does not respond during such 30 day period, it shall be deemed to have elected not to (i) purchase the assets pursuant to the Offer or (ii) terminate this Agreement.

(c) If Dreyfus accepts the Offer, the closing of the acquisition of the assets shall be consummated on or before the 60th day after TEPPCO receives the Acceptance but effective at the end of the calendar month occurring on or immediately prior to such closing. The acquisition shall be consummated at a closing in accordance with the provisions of Section 8.7, at which time Dreyfus shall deliver to TEPPCO the purchase price (in the form of immediately available funds), and TEPPCO shall deliver to Dreyfus such transfer documentation reasonably acceptable to Dreyfus as shall be required to evidence the transfer of such assets.

(d) If Dreyfus does not accept the Offer, TEPPCO shall be permitted for a period of six months after expiration of the Offer Period to sell all (but not less than all) of the subject assets on terms not more favorable to such transferee than the terms specified in the Acquisition Proposal and at a price that is not less than the price specified in the Acquisition Proposal. Should a third party acquire the assets in question and unless Dreyfus has elected to terminate the Agreement pursuant to Section 8.6(b), it would acquire those assets subject to this Agreement and the Dreyfus Storage Agreement which would continue in accordance with their respective terms. If Dreyfus has elected to terminate this Agreement pursuant to Section 8.6(b), this Agreement shall terminate effective as of the closing of such sale of the subject assets, subject, however, to the obligation to pay to Dreyfus the termination fee specified below in Section 8.6(e). TEPPCO agrees to give prompt notice to Dreyfus of such closing.

(e) Upon the closing of a sale pursuant to Section 8.6(d) and termination of this Agreement by Dreyfus pursuant to Section 8.6(b), TEPPCO shall pay to Dreyfus a termination fee in an amount equal to the greater of (i) 50% of the Net EBITDA for the Pricing Period immediately preceding the month in which the sale occurred, times 10 and (ii) the Dreyfus Investment at the end of such Pricing Period.

8.7 CLOSING.

(a) The closing of any transaction pursuant to this Article VIII shall be held at TEPPCO's principal office on a closing date which shall be no later than 60 days after, in the case of a termination of this Agreement pursuant to (i) Section 6.2, the date of the final determination that the Milestone has not been met, (ii) Section 6.6 or 12.1(b) or (c), the date of the written notice of termination, (iii) Section 8.3, or 10.1, the end of the applicable Pricing Period, (iv) Section 8.4, the end of the 30 day period set forth in such section, and (v) Section 8.6, as specified in that section.

(b) At such closing, any transferor Party shall deliver to the transferee Party such instruments of sale, assignment and transfer of its (and its transferee's, if any) entire right, title and interest in the Contract Rights or the subject assets, as may be the case, (including, without limitation, any rights of the transferor Party to receive repayment of its advances or payments of EBITDA and/or other payments for periods following such closing) and shall transfer to the transferee Party title to such Contract Rights or assets, free and clear of all Liens by or through such Party or its Affiliates, and shall deliver to the transferee Party such evidence of due authorization, execution and delivery and of the absence of any such Liens or competing claims as the transferee Party reasonably may request. The transferee Party shall pay the purchase price or termination fee, as applicable, as well as any and all stamp, recording and transactional taxes payable upon the sale, assignment and transfer (other than taxes imposed on or with respect to the transferor Party which are payable as result of the sale, assignment or transfer). The payment of the purchase price or termination fee, as may be the case, shall in each case be in immediately available funds payable at the closing and shall be deemed to include and constitute full payment and satisfaction for the transferring Party's entire right, title and interest in the Contract Rights or assets, as may be the case. No interest shall be due or owing on any amounts paid pursuant to the provisions of this Article VIII unless specifically contracted for in writing.

8.8 TRANSFER PROCEDURES. Assuming this Agreement is not terminated pursuant to its provisions, any Party who desires to transfer its Contract Rights shall arrange for any permitted transferee to be bound by the provisions of this Agreement, as it may then be amended, by having such transferee execute two counterparts of an instrument of assignment reasonably satisfactory in form and substance to the other Party. If and when the consent of the other Party to such assignment and the substitution of such transferee hereof is secured and the other requirements of this Article VIII are satisfied, the transferee shall become a party to this Agreement as to the Contract Rights thus transferred effective as of the first day of the calendar month during which the other Party actually receives the aforesaid instrument of assignment executed by both the transferor and transferee. The transferee shall be required to pay any and all reasonable out-of-pocket filing and recording fees, legal fees, accounting fees, and other charges and fees incurred by the other Party and its counsel as a result of such transfer. If the transferee is an Affiliate of the transferor, the transferor will continue to be liable for its obligations and liabilities under this Agreement. However, if the transferee is not an Affiliate of the transferor, the transferor shall be released from any and all obligations and liabilities hereunder, and the other Party to this Agreement shall look solely to the transferee for performance of any such obligations or satisfaction of such liabilities.

ARTICLE IX ADDITIONAL AGREEMENTS

9.1 EFFORTS TO SATISFY CONDITIONS. Dreyfus and TEPPCO agree to use their commercially reasonable efforts to bring about the satisfaction of the conditions

specified in Section 6.1 and Article VIIVII hereof. The Parties agree to execute, and shall cause Newco to execute, the other agreements called for by this Agreement on or prior to the Closing, including the Operating Procedures, the Dreyfus Storage Agreement and the TEPPCO Storage Agreement.

9.2 RETENTION OF RECORDS. TEPPCO shall retain all Records, including computer disks reflecting any books or records, documents or other papers, or other information or data relating to the operation of the MB Business or the TEPPCO Assets. Dreyfus and its authorized representatives shall have the right to inspect and, at its expense, copy, at any time during regular business hours for any proper purpose, the corporate, accounting, auditing and tax books, records (including work papers) and other books and records relating to the MB Business and the TEPPCO Assets in the possession of TEPPCO or its Affiliates. For a period of seven years following the Effective Date, TEPPCO and Dreyfus agree that they will not dispose of or destroy any such books and records.

9.3 NONDISCLOSURE OF PROPRIETARY INFORMATION.

(a) Each of TEPPCO and Dreyfus agrees that, for a period beginning on the date of this Agreement and ending on the second anniversary date of the earlier to occur of (i) the termination of this Agreement for whatever reason, or (ii) the disposition by TEPPCO or Dreyfus, respectively (or, to the extent such Contract Rights have been transferred to an Affiliate of such Party as permitted hereby, by such Affiliate), of its Contract Rights to a Person that is not a wholly-owned subsidiary of such Party's Parent, it and its Affiliates will:

(i) maintain the confidentiality of all confidential and proprietary non-public information relating to TEPPCO, Dreyfus, the MB Business or the TEPPCO Assets, applying the same standards as it does with respect to its own similar confidential or proprietary non-public information; and

(ii) not recruit any employees of the other Party employed in the MB Business.

Notwithstanding the foregoing, each Party and its Affiliates may disclose information that (x) is required to be disclosed by applicable state or federal tax or securities laws to the extent, and only to the extent, the laws require the disclosure and such Party provides the other Party prior written notice of its intent to provide the disclosure and the general text of the disclosure, or (y) is required to be disclosed by a court or administrative body of competent jurisdiction.

(b) Notwithstanding any other provision in this Agreement, Dreyfus recognizes and agrees that it shall not have access to information regarding inventory levels maintained by customers of the TEPPCO Assets.

9.4 TRADEMARK AND TRADE NAME LICENSE. Should the Closing occur and the Parties wish to continue to use any of the tradenames, trademarks or service marks

used by TEPPCO in the MB Business or in connection with the TEPPCO Assets, TEPPCO will negotiate in good faith a license for such continued use, subject to appropriate protections of TEPPCO's rights in and to such names and marks and its use thereof in its other business operations.

ARTICLE X NONCOMPETITION AGREEMENT

10.1 NONCOMPETITION COVENANT.

(a) Each Party agrees that, for a period beginning on the date of this Agreement and ending on the earlier to occur of (i) the termination of this Agreement for whatever reason, or (ii) the disposition, directly or indirectly, by TEPPCO or Dreyfus, respectively (or, to the extent such interest has been transferred to an Affiliate of such Party as permitted hereby, by such Affiliate), of its Contract Rights to a Person that is not a wholly-owned, direct or indirect, subsidiary of such Party's Parent, no such Party or any of its Restricted Affiliates will engage or participate in, or carry on, directly or indirectly, either as proprietor, partner, member, director, stockholder, agent, consultant, advisor, trustee, Affiliate, or otherwise, whether or not for compensation, any business that competes with the MB Business in the Territory. Each Party acknowledges and agrees that the foregoing noncompetition agreement is given in partial consideration for the other Party's entering into this Agreement.

(b) Each of the Parties or its Affiliates have, pursuant to Storage Agreements, leased or obtained the right to use storage capacity in the TEPPCO Assets. Without limiting the generality of the provisions of Section 10.1(a) and in furtherance thereof, each of the Parties agrees that it and its Affiliates will use such storage capacity solely for its own internal business operations and will not lease or sublease such storage capacity to third parties, and will direct to TEPPCO any such third party seeking access to the TEPPCO Assets so that such additional business shall be for the benefit of both Parties during the Term. The foregoing shall not be interpreted to prohibit or restrict the current practices of TEPPCO with respect to such storage capacity. Additionally, except as otherwise herein specifically provided, the Parties agree to act in good faith with respect to each other and the MB Business to maximize the returns from such business throughout the Term.

(c) In the event either Party (the "First Party") or any of its Restricted Affiliates desires to pursue a business opportunity that will compete with the MB Business in the Territory as prohibited by Section 10.1 ("Competing Business Transaction"), such Party shall be obligated to promptly bring such Competing Business Transaction to the attention of the other Party (the "Second Party"), to fully disclose all details regarding the proposed opportunity (subject to the execution of appropriate confidentiality undertakings), and to offer such business opportunity to become part of the MB Business. The Parties shall negotiate in good faith for a period of forty-five (45) days (unless the business opportunity dictates a shorter period of time, in which event,

for such shorter time period as will avoid the loss of such opportunity) to bring such Competing Business Transaction into the MB Business, either by contribution or purchase. Should the Parties be unable to agree within such forty-five (45) day period on the terms and provisions for incorporating the Competing Business Transaction into the MB Business and the First Party or its Restricted Affiliate, as may be the case, wishes to pursue the Competing Business Transaction rather than continue with the MB Business and the transactions contemplated by this Agreement, subject to the provisions of Section 10.1(e) below, either Party may terminate this Agreement by giving written notice to the other Party within 30 days of the consummation of the Competing Business Transaction, in which event the provisions of Section 8.5(a) shall apply. The First Party agrees to give written notice within one Business Day of the consummation of such transaction.

(d) In the event the First Party or any of its Restricted Affiliates desires to pursue a Competing Business Transaction which cannot become a part of the MB Business or Newco for any reason other than the failure of the Second Party to consent to such transaction, and the First Party or its Restricted Affiliate, as may be the case, elects to pursue such Competing Business Transaction rather than continue with the MB Business and the transactions contemplated by this Agreement, the First Party shall give written notice of such election to the Second Party and either Party may terminate this Agreement by giving written notice to the other Party within 30 days of the consummation of the Competing Business Transaction, in which event the provisions of Section 8.5(a) shall apply. The First Party agrees to give written notice within one Business Day of the consummation of such transaction.

(e) Notwithstanding any other provision of this Agreement, the Second Party may waive the non-competition provision of this Article X as it applies to any Competing Business Transaction offered by the First Party under Section 10.1(c). In such event, (i) the First Party or its Restricted Affiliate, as may be the case, may participate in such Competing Business Transaction without being in violation of this Agreement; (ii) the Second Party thereby waives any right to participate in, or to benefit from, such transaction; and (iii) neither Party shall be entitled to invoke the termination option of Section 8.5(a) as a result of such transaction. Any such waiver shall apply only to the specific Competing Business Transaction being presented and shall not be construed as applicable to any other Competing Business Transaction.

(f) Notwithstanding the foregoing or any other provision of this Agreement:

(i) Neither Party nor any of its Affiliates shall be prohibited from conducting business activities or operations outside the Territory of a similar nature as are conducted by such Party or Affiliate as of the date hereof; or

(ii) Neither Party nor any of its Affiliates shall be prohibited from constructing and/or operating other pipelines to, from or through the Territory so long as

they do not directly compete with the pipelines that comprise a part of the TEPPCO Assets as of the date hereof, or

(iii) Neither Party nor any of its Affiliates shall be prohibited from operating pipelines to, from or through the Territory which are currently operated by such Persons, including the servicing of Persons who are customers of the TEPPCO Assets, or from expanding, improving or otherwise modifying such pipelines, or

(iv) Neither Party nor any of its Affiliates has any obligation or duty to first offer to the other Party new business ideas that are outside the scope of the MB Business.

10.2 DEFINITIONS. For purposes of this Article X:

(a) "Restricted Affiliates" shall mean with respect to each Party, any Person in which such Party's Parent owns, directly or indirectly, at least a 51% voting or equity interest, or which it otherwise controls. For purposes of this definition, "control" shall have the definition given that term in the definition of "Affiliate" and Louis Dreyfus Natural Gas Corporation shall not be deemed to be a Restricted Affiliate of Dreyfus.

Texas.

(b) "Territory" shall mean Chambers and Harris Counties,

10.3 REASONABLENESS OF COVENANT. Each Party acknowledges that the covenant provided in Section 10.1 hereof is manifestly reasonable on its face and is no more restrictive than is required for the protection of the MB Business, including the interest of the other Party in the MB Business. In the event that the provisions of Section 10.1 should ever be deemed to exceed the time and geographic limitations permitted by applicable law, then such provisions shall be reformed to the maximum time or geographic limitations permitted by applicable law.

10.4 INJUNCTIVE RELIEF. It is specifically understood and agreed that any breach of the provisions of Section 10.1 hereof by a Party is likely to result in irreparable harm to the other Party and that an action at law for damages alone will be an inadequate remedy for such breach, and that the non-breaching Party would suffer irreparable harm in the event a Party fails to comply with its obligations hereunder. Therefore, in addition to any other remedy that may be available to it, the non-breaching Party shall be entitled to enforce the specific performance of Section 10.1 of the breaching Party and to seek both temporary and permanent injunctive relief (to the extent permitted by law) without the necessity of proving actual damages, and such other relief as the court may allow.

ARTICLE XI INDEMNIFICATION

11.1 INDEMNITY OBLIGATION. Subject to the provisions of this Agreement, including without limitation this Article XI and Article XII:

(a) Each of TEPPCO and Dreyfus shall indemnify and hold harmless the other Party, its Affiliates and the officers, directors, shareholders, agents, employees, representatives, successors and assigns of each of them (said Persons being sometimes referred to as "Party Indemnitees") in respect of any and all Losses, relating to matters reasonably incurred by such other Party to the extent arising from each and all of the following:

(i) any breach of a representation or warranty made by such Party in this Agreement or any Schedule or Exhibit attached hereto or delivered pursuant hereto; or

(ii) the material breach of any covenant, agreement or obligation of such Party or its Affiliates contained in this Agreement or any Schedule or Exhibit hereto, the Organizational Agreements or any other instrument contemplated by this Agreement.

(b) TEPPCO shall indemnify and hold harmless Dreyfus and its Party Indemnitees in respect of any and all Losses relating to matters reasonably incurred by such parties to the extent arising from:

(i) any Liens against the TEPPCO Assets other than the Permitted Encumbrances or Liens resulting from borrowings for Discretionary Capital Expenditures; or

(ii) (x) any actual or alleged release of any Hazardous Materials on or from the TEPPCO Assets, (y) any Environmental Liability related to the TEPPCO Assets, or (z) any actual or prospective claim, litigation, investigation or proceeding relating to any of the foregoing; provided that such indemnity shall not, as to any indemnitee, be available to the extent that such Losses are a result of the acts or omissions of Dreyfus or its Party Indemnitees; and provided further, that the indemnification obligations of this Section 11.1(b)(ii) shall apply only for Losses resulting from events specified in clauses (x), (y) and (z) above which occurred during the Term.

(c) Amounts covered by insurance or other proceeds, reimbursements or contributions paid by Persons not Affiliates of any of the Parties shall not be included in determining Losses hereunder. To the extent such amounts are included under this Section 11.1 and are later reimbursed by such non-Affiliates, equitable adjustments and repayments shall be made.

11.2 PROCEDURE. All claims for indemnification by a Person under this Article XI (the party claiming indemnification and the party against whom such claims are

asserted being hereinafter called the "Indemnified Party" and the "Indemnifying Party", respectively) shall be asserted and resolved as follows:

(a) Whenever any claim or demand for which an Indemnifying Party would be liable to an Indemnified Party hereunder (which shall be deemed to include any claim which exhausts any part of the applicable Threshold limitation specified in Section 11.3) is asserted against or sought to be collected from such Indemnified Party by a third party, such Indemnified Party shall, within 15 days of the receipt thereof, give notice (the "Claim Notice") to the Indemnifying Party of such claim or demand, specifying the nature of and specific basis for such claim or demand and the amount or the estimated amount thereof to the extent then feasible, which estimate shall not be binding upon the Indemnified Party in its effort to collect the final amount of such claim or demand. To the extent the Indemnifying Party is prejudiced thereby, the failure to so notify the Indemnifying Party of any such claims or action shall relieve the Indemnifying Party from liability that it may have to the Indemnified Party under the indemnification provisions contained in this Article XI, but only to the extent of the loss directly attributable to such failure to notify, and shall not relieve the Indemnifying Party from any liability that it may have to the Indemnified Party otherwise than under this Article XI. The Indemnifying Party shall, within 20 days of the receipt of a Claim Notice, notify the Indemnified Party as to whether it accepts, in whole or in part, its indemnity obligation, in which case, the Indemnifying Party shall assume and thereafter conduct the defense thereof; provided that the Indemnified Party shall be entitled to participate in the defense thereof at its own expense. If the Indemnifying Party disputes liability under this Section 11.2(a) or otherwise fails to defend within a reasonable time after notice, the Indemnified Party will have the right to undertake the defense, compromise or settle such third party claim, and to compromise or settle such claim on behalf of, and for the account of the Indemnifying Party, at the expense and risk of the Indemnifying Party. The consent to the entry of any judgment or settlement of any claim hereunder by the Indemnifying Party may only be made upon the prior approval by the Indemnified Party, which approval shall not be unreasonably withheld, unless the judgment or proposed settlement involves only the payment of money damages (which would be paid by the Indemnifying Party) and does not impose any injunction or other equitable relief on the Indemnified Party in which case consent is not required.

(b) If requested by the Indemnifying Party, the Indemnified Party agrees to cooperate with the Indemnifying Party and its counsel in contesting any claim or demand that the Indemnifying Party elects to contest, or, if appropriate and related to the claim in question, in making any counterclaim against the person asserting the third party claim or demand, or any cross-complaint against any Person. The Indemnifying Party shall reimburse such Indemnified Party for reasonable out-of-pocket expenses incurred by the Indemnified Party in such cooperation.

(c) If any Indemnified Party should have a claim against the Indemnifying Party hereunder that does not involve a claim or demand being asserted against or sought to be collected from it by a third party, the Indemnified Party shall send a Claim Notice with respect to such claim to the Indemnifying Party.

Reimbursement of any Losses incurred by the Indemnified Party pursuant to this Article XI shall be made within 30 days after documentation is sent to the Indemnifying Party by the Indemnified Party. If the Indemnifying Party disputes such claim, such dispute shall be resolved in the manner set forth in Article XIV hereof and reimbursement shall be withheld pending the outcome of the proceeding under Article XIV.

11.3 INDEMNIFICATION THRESHOLD. Except as otherwise specifically provided herein, the obligation of TEPPCO or Dreyfus to make payments to the other Party or the Party Indemnitees of the other Party pursuant to Section 11.1 shall be limited to the excess of the aggregate of the amounts of any such Losses over \$500,000 except for the TEPPCO indemnification obligations of:

(i) Section 11.1(b)(i) which shall not be subject to a minimum threshold limitation; and

(ii) Section 11.1(b)(ii) which shall be limited to the excess of the aggregate amount of such Losses over the Dreyfus Investment.(Each of the Dreyfus Investment and the \$500,000 specified above in this Section 11.3 is sometimes herein referred to as the "Threshold.").

11.4 SURVIVAL. The representations, warranties and indemnities of TEPPCO or Dreyfus contained in this Agreement shall survive the making of this Agreement, any investigation of the other party hereto and the Closing and, except as set forth below, shall terminate on the first anniversary of the Effective Date; provided that, if a claim or demand for indemnification has been made or given within the applicable survival period and has not been resolved as of the expiration of such period, such claim (and, if the claim results from a breach of a representation or warranty, such representation or warranty) shall survive until the final resolution of such claim. In the event that the Closing does not occur and this Agreement is terminated, for whatever reason, effective upon such termination, all representations, warranties and indemnities shall terminate other than the indemnification obligation of Section 11.1(b)(ii) which shall survive such termination.

11.5 EXCLUSIVE REMEDY; LIMITATIONS.

(a) THE PARTIES AGREE THAT ONLY ACTUAL DAMAGES SHALL BE RECOVERABLE UNDER THIS AGREEMENT AND HEREBY WAIVE ANY RIGHT TO RECOVER SPECIAL, PUNITIVE, CONSEQUENTIAL, INCIDENTAL OR EXEMPLARY DAMAGES EXCEPT TO THE EXTENT AN INDEMNIFIED PARTY SUFFERS SUCH DAMAGES TO AN UNAFFILIATED THIRD-PARTY IN CONNECTION WITH A THIRD-PARTY CLAIM, IN WHICH EVENT SUCH DAMAGES SHALL BE RECOVERABLE. Notwithstanding anything to the contrary in this Agreement, the indemnification provisions of this Agreement shall be the exclusive remedies for any claim based upon this Agreement or the transactions contemplated hereby (except with respect and to the extent the Parties or their Affiliates are parties to other agreements, in which case such other agreements shall govern their obligations

with respect to such agreements). In furtherance of the foregoing and except for the indemnification provisions of this Agreement, effective upon the execution hereof, the Parties, on behalf of themselves and their Affiliates, each release, remise and forever discharge the other Party and its Affiliates, and all such Parties' stockholders, officers, directors, employees, agents, managers, advisors and representatives from any and all claims and losses, in law or in equity, known or unknown, which such parties might now or subsequently may have, based on, relating to or arising out of this Agreement, the ownership, use or operation of TEPPCO Assets or the MB Business or the condition, quality, status or nature of such assets or business, including rights to contribution under applicable environmental laws, even if caused in whole or in part by the negligence (whether sole, joint or concurrent), strict liability or other legal fault of any released Person, excluding, however, any contractual rights arising under other written agreements entered into by the Parties or their respective Affiliates.

(b) Notwithstanding anything to the contrary contained elsewhere in this Agreement, an Indemnifying Party's aggregate indemnity obligation with respect to Losses under or in connection with this Agreement or the transactions contemplated hereby shall not exceed \$5,000,000 except for the indemnification obligation of TEPPCO under Section 11.1(b) which shall not be subject to the foregoing limitations.

ARTICLE XII TERMINATION

12.1 TERMINATION PRIOR TO CLOSING. This Agreement shall terminate, or may be terminated, as the case may be, as follows:

(a) By mutual agreement of Dreyfus and TEPPCO;

(b) By Dreyfus, by written notice if a material default shall be made in the observance or performance by TEPPCO or of any agreements and covenants of TEPPCO herein contained, or if there shall have been a material breach by TEPPCO of any warranties and representations; provided, however, that Dreyfus shall have given TEPPCO thirty days prior written notice of such default or breach, and TEPPCO shall not have cured such default or breach within such thirty day period, in which event the provisions of Section 8.5(c)(i) shall apply;

(c) By TEPPCO, by written notice if a material default shall be made by Dreyfus in the observance or performance by it of any agreements and covenants of Dreyfus herein contained, or if there shall have been a material breach by Dreyfus of any warranties and representations; provided, however, that TEPPCO shall have given Dreyfus thirty days prior written notice of such default or breach, and Dreyfus shall not have cured such default or breach within such thirty day period, in which event the provisions of Section 8.5(c)(ii) shall apply;

(d) Upon the exercise of an option to terminate as provided in Section 6.2, 6.6, 8.3, 8.4, 8.6 or 10.1; or

(e) Automatically, as provided in Section 6.2.

12.2 LIABILITY UPON TERMINATION. If this Agreement is terminated pursuant to any provision of Section 12.1 or due to the failure of either Party, for whatever reason, to satisfy the Closing conditions of Article VII, neither Party shall have any liability to the other Party hereunder. Notwithstanding the foregoing provisions or the termination of this Agreement pursuant to this Article, obligations that by their terms continue in existence or that are created as a result of such termination, including the obligation to pay a termination fee or other amounts as a result of such termination and amounts payable pursuant to Section 3.2, and the provisions of Sections 9.2, 9.3, and this 12.2 and of Articles XI, XIII, XIV and XV shall continue in effect.

12.3 EFFECTIVE DATE OF TERMINATION. All notices of termination given pursuant to this Agreement shall specify the effective date of such termination which shall not be less than 30 days, nor more than 60 days, from the date of such notice. Should a Party fail to state a termination date, it shall occur 30 days from the date of such notice unless the Parties agree otherwise.

12.4 TERMINATION OF INTEREST. Upon the termination of this Agreement (other than through the exercise by Dreyfus of an Offer pursuant to Section 8.6) and the payment to Dreyfus of all termination and other amounts due hereunder, Dreyfus shall (i) convey or release to TEPPCO the Dreyfus' Contract Rights and the Dreyfus Investment, and (ii) cease to have any ongoing right, interest or claim in or to the TEPPCO Assets, the MB Business or any income derived therefrom.

ARTICLE XIII EXPENSES

Except as otherwise set forth herein, and whether or not the transactions contemplated by this Agreement shall be consummated, each Party agrees to pay, without right of reimbursement from any other Party, the costs incurred by the Party incident to the preparation and execution of this Agreement and performance of its obligations hereunder, including the fees and disbursements of legal counsel, accountants and consultants employed by the Party in connection with the transactions contemplated by this Agreement. The pre-formation and pre-organization costs of Newco and the General Partner shall be shared equally by TEPPCO and Dreyfus.

ARTICLE XIV RESOLUTION OF DISPUTES; CONSENT TO JURISDICTION AND VENUE

The Parties agree that, except as otherwise specifically provided herein, all disputes under this Agreement shall be resolved in accordance with the procedures set forth in Exhibit O hereto.

ARTICLE XV GENERAL PROVISIONS

15.1 FURTHER ASSURANCES. At any time or from time to time at and after the Effective Time, each of the Parties shall, at the request of the other, execute and deliver or cause to be executed and delivered all the assignments, consents, documents and instruments, and take or cause to be taken all the other reasonable actions as may be necessary or desirable to more fully and effectively carry out the intent and purposes of this Agreement.

15.2 NOTICES. All notices, requests, demands and other communications required or permitted to be given under this Agreement shall be deemed to have been duly given if in writing and delivered personally or sent via first-class, postage prepaid, registered or certified mail (return receipt requested), or by overnight delivery service or facsimile transmission addressed as follows:

If to TEPPCO:

TE Products Pipeline Company, Limited Partnership 2929 Allen Parkway, Suite 3200 Houston, Texas 77019 Attention: President Telephone: (713) 759-3636 Facsimile Number: (713) 759-3957

and copy to:

Texas Eastern Products Pipeline Company 2929 Allen Parkway, Suite 3200 Houston, Texas 77019 Attention: General Counsel Telephone: (713) 759-3968 Facsimile Number: (713) 759-3645

If to Dreyfus:

Louis Dreyfus Plastics Corporation 10 Westport Road Wilton, CT 06897 Attention: President Telephone: 203/761-8000 Facsimile Number: 203/761-8119

Louis Dreyfus Plastics Corporation 10 Westport Road Wilton, CT 06897 Attention: General Counsel Telephone: 203/761-8000 Facsimile Number: 203/761-8321

Any Party may change the address to which the communications are to be directed to it by giving notice to the other in the manner provided in this Section 15.2. Notice by mail shall be deemed to have been given and received on the third calendar day after posting. Notice by overnight delivery service, facsimile transmission or personal delivery shall be deemed given on the date of actual delivery.

15.3 GOVERNING LAW. This Agreement and the performance of the transactions contemplated hereby shall be governed by and construed and enforced in accordance with the laws of the State of Delaware, without regard to any conflict-of-laws provision thereof that would otherwise require the application of the law of any other jurisdiction.

15.4 ENTIRE AGREEMENT. This Agreement and the Exhibits hereto, together with the certificates, documents, instruments and writings that are delivered pursuant hereto, set forth the entire agreement and understanding of the Parties with respect to the transactions contemplated hereby and supersede all prior agreements, arrangements and understandings relating to the subject matter hereof. No representation, promise, inducement or statement of intention with respect to the subject matter of this Agreement has been made by any Party that is not embodied in this Agreement and Exhibits hereto, the Organizational Agreements and the Conveyance Document, the certificates, documents, instruments and writings that are delivered pursuant hereto, and none of the Parties shall be bound by or liable for any alleged representation, promise, inducement or statement of intention not so set forth.

15.5 ASSIGNMENT. No Party to this Agreement may sell, transfer, assign, pledge or hypothecate its or his rights, interests or obligations under this Agreement without the written consent of the other Party except to a transferee pursuant to Section 8.1(b) or to a transferee of the TEPPCO Assets pursuant to Section 8.6.

15.6 SUCCESSORS. This Agreement shall inure to the benefit of, be binding upon, and be enforceable by the parties hereto and their respective successors and permitted assigns.

15.7 AMENDMENTS; WAIVER. This Agreement may be amended, superseded or canceled, and any of the terms hereof may be waived, only by a written instrument specifically stating that it amends, supersedes or cancels this Agreement or waives any of the terms herein, executed by all Parties intended to be bound thereby or, in the case

of a waiver, by the Party waiving compliance. The failure of any Party at any time to require performance of any provision herein shall in no manner affect the right at a later time to enforce the same. No waiver by any Party of any condition, or of any breach of any term, covenant, representation or warranty, shall be deemed or constitute a waiver of any other condition, or breach of any other term, covenant, representation or warranty, nor shall the waiver constitute a continuing waiver unless otherwise expressly provided.

15.8 COUNTERPARTS. This Agreement may be executed in two or more counterparts, each of which shall be deemed an original, but all of which shall constitute one and the same instrument.

15.9 SEVERABILITY. Any provision hereof that is prohibited or unenforceable in any jurisdiction shall, as to such jurisdiction, be ineffective to the extent of such prohibition or unenforceability without invalidating the remaining provisions hereof, and any such prohibition or unenforceability in any jurisdiction shall not invalidate or render unenforceable such provision in any other jurisdiction.

15.10 NO THIRD PARTY BENEFICIARIES. Except for Newco or to the extent another third Person is expressly given rights herein, any agreement contained, expressed or implied in this Agreement shall be only for the benefit of the Parties hereto and their respective legal representatives, successors and assigns, and such agreements shall not inure to the benefit of the obligees of any indebtedness of any Party hereto, it being the intention of the Parties hereto that no Person or entity shall be deemed a third party beneficiary of this Agreement, except to the extent a third party is expressly given rights herein.

15.11 INDEPENDENT CONTRACTOR; NO PARTNERSHIP. In providing the services called for by this Agreement, each Party and its Affiliates shall be deemed independent contractors with respect to the other Party and its Affiliates. Nothing in this Agreement or the transactions contemplated hereby, except as and when the Partnership Agreement is entered into by the Parties, is intended, nor should it be construed, to form a partnership, joint venture or other association.

15.12 NEGOTIATED TRANSACTION. The provisions of this Agreement were negotiated by the Parties hereto, and this Agreement shall be deemed to have been drafted by all of the Parties hereto.

15.13 BUSINESS DAYS. All references to days contained in this Agreement shall be to calendar days unless otherwise specified.

[Remainder of this page left intentionally blank.]

IN WITNESS WHEREOF, the Parties have duly executed this Agreement as of the date first set forth above.

TE PRODUCTS PIPELINE COMPANY, LIMITED PARTNERSHIP By: Texas Eastern Products Pipeline Company, its General Partner

By: /s/ DAVID LANGLEY

Name: David	Langley
Title: Vice	President

LOUIS DREYFUS PLASTICS CORPORATION

By: /s/ WILLIAM RIPPE Name: William Rippe Title: Executive Vice President

2	۵	
J	э	

EXHIBIT E OF JOINT DEVELOPMENT AGREEMENT

LIMITED LIABILITY COMPANY AGREEMENT

0F

MONT BELVIEU VENTURE, LLC

BETWEEN

TE PRODUCTS PIPELINE COMPANY, LIMITED PARTNERSHIP

AND

LOUIS DREYFUS ENERGY SERVICES L.P.

AS THE MEMBERS

DATED AS OF JANUARY 21, 2003

DEFINITIONS		1
ARTICLE I Section 1.1 Section 1.2	FORMATION OF LIMITED LIABILITY COMPANY Formation Partnership for Tax Purposes Only	6
ARTICLE II	NAME	6
ARTICLE III	PURPOSE	6
ARTICLE IV	NAMES AND ADDRESSES OF MEMBERS	7
ARTICLE V	REGISTERED AGENT; REGISTERED OFFICE; PRINCIPAL OFFICE; ADDITIONAL OFFICES	7
ARTICLE VI	TERM	7
ARTICLE VII	ADDITIONAL MEMBERS	7
ARTICLE VIII Section 8.1 Section 8.2 Section 8.3 Section 8.4	CAPITAL CONTRIBUTIONS. Initial Contributions. Additional Contributions. Capital Accounts. Adjustment of Capital Accounts.	8 10 10
ARTICLE IX Section 9.1 Section 9.2 Section 9.3 Section 9.4	DISTRIBUTIONS. Amount; Timing. Failure to Withdraw. Transferor/Transferee Allocations. Member Advances.	11 11 11
ARTICLE X Section 10.1 Section 10.2 Section 10.3 Section 10.4 Section 10.5	ALLOCATIONS OF INCOME, GAIN, LOSS, DEDUCTION AND CREDIT General Allocations for Tax Purposes Transferor/Transferee Allocations Reliance on Advisors Tax Matters Partner	11 12 13 13
ARTICLE XI Section 11.1	BOOKS OF ACCOUNT, RECORDS AND TAX INFORMATION Maintenance of Books and Records	

-i-

	Section 11.2	Reports	14
ARTICLE	XII	FISCAL YEAR	14
ARTICLE	XIII	COMPANY FUNDS	14
ARTICLE	XIV	STATUS OF MEMBERS	14
ARTICLE	XV Section 15.1 Section 15.2 Section 15.3 Section 15.4 Section 15.5 Section 15.6 Section 15.7 Section 15.8	MANAGEMENT AND OPERATION OF BUSINESS. Member Management. Management Committee. Compensation and Reimbursement. Managing Member. Marketing Activities. Exculpation. Indemnification. Other Activities; Noncompetition.	15 15 17 17 19 20 20 21
ARTICLE	XVI Section 16.1	TRANSFER OF MEMBERSHIP INTERESTS Restrictions on Transfer	21 21
ARTICLE	XVII	DISSOLUTION OF THE COMPANY	22
ARTICLE	XVIII Section 18.1 Section 18.2 Section 18.3 Section 18.4 Section 18.5 Section 18.6	WINDING UP AND TERMINATION OF THE COMPANY Liquidator Reserves Sale of Assets; Distribution of Proceeds Final Accounting Recourse Limited to Company Assets Termination.	22 22 23 23 23 23 23 24
ARTICLE	XIX	NOTICES	24
ARTICLE	XX Section 20.1 Section 20.2	AMENDMENTS AND MEETINGS Amendments Meetings of the Members	24 24 24
ARTICLE	XXI	REPRESENTATIONS, WARRANTIES AND COVENANTS	25
ARTICLE	XXII	DISPUTES	25
ARTICLE	XXIII	MISCELLANEOUS	25
	Section 23.1 Section 23.2	No Right of Partition Entire Agreement; Supersedure	

-ii-

Section 23.3	Governing Law	
Section 23.4	Binding Effect	26
Section 23.5	Construction of Agreement	26
Section 23.6	Captions	
Section 23.7	Effect of Invalid Provision	26
Section 23.8	Counterparts	
Section 23.9	Waiver	26

EXHIBIT A

MB Assets

-iii-

LIMITED LIABILITY COMPANY AGREEMENT OF MONT BELVIEU VENTURE, LLC

THIS AGREEMENT is made as of the 21st day of January, 2003, between and among TE Products Pipeline Company, Limited Partnership, a Delaware limited partnership, and Louis Dreyfus Energy Services L.P., a Delaware limited partnership.

WHEREAS, the parties hereto desire to form a limited liability company under and pursuant to Delaware law.

NOW, THEREFORE, in consideration of the premises and the mutual undertakings contained herein, the parties hereto hereby agree as follows:

DEFINITIONS

The following definitions shall be applicable to the terms set forth below as used in this Agreement:

"ACT" means the Delaware Limited Liability Company Act, 6 Del. Code Section 18-101 et seq., as it may be amended from time to time, and any successor to said Act.

"AFFILIATE" means, when used with respect to a specified person, any other person directly or indirectly controlling or controlled by or under direct or indirect common control with the specified person, provided that the Company shall not be deemed to be an Affiliate of the Members or the Partnership or any of their respective subsidiaries or Affiliates. For purposes of this definition, "control", when used with respect to any specified person, means the power to direct the management and policies of the person, directly or indirectly, whether through the ownership of voting securities, by contract, by family relationship or otherwise; and the terms "controlling" and "controlled" have the meanings correlative to the foregoing. Notwithstanding the foregoing, for purposes of this Agreement, Louis Dreyfus Natural Gas Corporation, an Oklahoma corporation, shall not be deemed an Affiliate of the LD Member.

"AGREED VALUE" means, in the case of any contributions or distributions of property, the fair market value of such property net of any indebtedness or other liability either assumed or to which such property is subject, as such fair market value is determined by the Members using such reasonable method of valuation as they may adopt.

"ALTERNATE REPRESENTATIVES" has the meaning given such term in Section 15.2(a).

"APPROVED BUSINESS PLAN" means a Business Plan and any amendments and revisions thereto that has been approved by the Management Committee pursuant to Section 15.2(b). In the event the Management Committee fails for any reason to approve a Business Plan for any fiscal year, the Approved Business Plan for the immediately preceding fiscal year shall continue in effect

-1-

but shall be increased by a percentage that is equal to the annualized percentage change in the most recent GDP Price Deflator over the immediately preceding GDP Price Deflator until a new Business Plan is approved. In such event, the Managing Member shall be authorized to adjust the amounts spent on each budget item so long as the aggregate budget amount is not exceeded. Should Business Plans not be approved for successive fiscal years, the above-described adjustment shall be made for each such year.

"BANKRUPT MEMBER" means any Member (i) that makes a general assignment for the benefit of creditors; (ii) is declared insolvent in any state insolvency proceedings; (iii) becomes the subject of an order for relief under Chapter 7 of the United States Bankruptcy Code, 11 U.S.C. Section 101 et seq., or successor statute (the "BANKRUPTCY CODE"); (iv) becomes a voluntary debtor in a case under Chapter 11 of the Bankruptcy Code and fails to achieve confirmation of a plan of reorganization within 180 days; (v) becomes an involuntary debtor in a case under either Chapter 7 or 11 of the Bankruptcy Code and fails to achieve a dismissal of the case within 90 days, or, with respect to a Chapter 11 case in which an order for relief is entered prior to the expiration of 90 days, fails to achieve confirmation of a plan of reorganization within 180 days of the commencement of the involuntary case; or (vi) consents to or is subjected to the appointment of a trustee, receiver or liquidator with respect to all or substantially all of its properties, and, where such appointment was contested, there has been a failure to vacate such appointment within 90 days of appointment.

"BUILT-IN GAIN" with respect to any Company property means (i) the excess of the Agreed Value of any Contributed Property over its adjusted basis for federal income tax purposes as of the time of contribution and (ii) in the case of any adjustment to the Carrying Value of any Company property subject to depreciation, cost recovery or amortization pursuant to Section 8.4 as a result of a contribution of cash for a Membership Interest, the Unrealized Gain with respect to such property.

"BUILT-IN LOSS" with respect to any Company property means (i) the excess of its adjusted basis for federal income tax purposes of any Contributed Property over its Agreed Value as of the time of contribution and (ii) in the case of any adjustment to the Carrying Value of any Company property subject to depreciation, cost recovery or amortization pursuant to Section 8.4 as a result of a contribution of cash for a Membership Interest, the Unrealized Loss with respect to such property.

"BUSINESS DAY" means any day other than a Saturday, Sunday or bank holiday in Houston, Texas.

"BUSINESS PLAN" shall mean the projected operating and capital spending budgets of the Partnership for each fiscal year (or in the case of the initial Business Plan the applicable portion thereof) and shall set forth income and expense projections and planned contributions and financing.

"CAPITAL ACCOUNT" means the account established for each Member pursuant to Section 8.3.

"CAPITAL CONTRIBUTIONS" means the Agreed Value of any property and the amount of cash contributed by a Member to the Company.

"CARRYING VALUE" with respect to any Capital Contribution means the Agreed Value of such property reduced as of the time of determination by all depreciation, cost recovery and amortization

-2-

deductions charged to the Capital Accounts with respect to such property and an appropriate amount to reflect any sales, retirements or other dispositions of assets included in such property and, with respect to any other Company property, the adjusted basis of such property for federal income tax purposes as of the time of determination. The Carrying Values shall be further adjusted as provided in Section 8.4.

"CERTIFICATE" means the Certificate of Formation of the Company to be filed in the Office of the Secretary of State of the State of Delaware pursuant to the Act, and any and all amendments thereto and restatements thereof.

"CHANGE OF MEMBER CONTROL" means either a TE Change of Member Control or a LD Change of Member Control.

"CODE" means the Internal Revenue Code of 1986, as amended and in effect on the effective date hereof and, to the extent applicable, as subsequently amended.

"COMPANY" means Mont Belvieu Venture, LLC, the limited liability company heretofore formed and hereby continued hereunder.

"CONTRIBUTED PROPERTY" means any Capital Contribution of property other than cash.

"DEFAULT INTEREST RATE" means a rate per annum equal to the lesser of (a) __% per annum plus the Prime Rate or (b) the maximum rate permitted by law.

"DELINQUENT MEMBER" has the meaning given such term in Section 8.1(c).

"DISCRETIONARY CAPITAL EXPENDITURES" means any expenditure related to the Partnership's assets that is classified under GAAP as a capital expenditure and (i) is not a Mandatory Capital Expenditure, or (ii) if classified as a Mandatory Capital Expenditure, is associated or is in conjunction with, or is the result of, a prior or current Discretionary Capital Expenditure, in which case the portion of such Mandatory Capital Expenditure that is allocable to such prior or current discretionary Capital Expenditure shall be classified as a Discretionary Capital Expenditure.

"DISPOSE, DISPOSING OR DISPOSITION" means, with respect to Membership Interest or any portion thereof, a sale, assignment, transfer, conveyance, gift, exchange or other disposition of such asset, whether such disposition be voluntary, involuntary or by operation of law, including but not limited to the following: (a) in the case of a Membership Interest owned by an entity, a Change of Member Control and (b) a disposition in connection with, or in lieu of, a foreclosure of an Encumbrance; but such terms shall not include the creation of an Encumbrance.

"DISPUTES" has the meaning assigned to such term in Article XXII.

"DISTRIBUTABLE CASH" means, at the time of determination, all Company cash derived from the conduct of the Company's business, other than (i) Capital Contributions, together with interest earned thereon pending utilization thereof, (ii) financing proceeds, (iii) amounts required for

-3-

working capital and (iv) other amounts that the Members reasonably determine to be necessary for the proper operation of the Company's business and its winding up and liquidation.

"EBITDA" means the Partnership's earnings for each fiscal year on a GAAP basis before all charges for such period, for depreciation and amortization, interest and income taxes.

"GAAP" means Generally Accepted Accounting Principles in the United States of America consistently applied for the time periods involved.

"INDEMNITEE" has the meaning ascribed thereto in Section 15.7(a).

"LD MEMBER" means Louis Dreyfus Energy Services L.P., a Delaware limited partnership, or its successors or assigns.

"LD MEMBER CHANGE OF CONTROL" means an event that causes S. A. Louis Dreyfus et Cie., a French societe anonyme, together with its Affiliates to cease to own, directly or indirectly, at least a 50% voting interest in (i) the LD Member or any Affiliate of the LD Member owning a partnership interest in the Partnership or (ii) in the event of the transfer of the LD Member's Membership Interest to an Affiliate, in such Affiliate.

"LENDING MEMBER" has the meaning given such term in Section 8.1(c).

"LIQUIDATOR" has the meaning ascribed thereto in Section 18.1.

"MANAGEMENT COMMITTEE" has the meaning given such term in Section 15.2.

"MANAGING MEMBER" has the meaning given such term in Section 15.4.

"MANDATORY CAPITAL EXPENDITURES" means all expenditures related to the Partnership's assets that are classified under GAAP as a capital expenditure, the primary function of which is for regulatory compliance, safety or operational integrity of the Partnership's assets or is otherwise the result of actions permitted in emergency situations as provided in the Operating Procedures.

"MB ASSETS" means the Mont Belvieu salt cavern storage and Houston area shuttle system more fully described on Exhibit A hereto and all improvements, expansions and additions thereto.

"MEMBERS" means TE Products Pipeline Company, Limited Partnership, a Delaware limited partnership and Louis Dreyfus Energy Services L.P., a Delaware limited partnership, or their respective successors or permitted assigns and "MEMBER" means any one of them.

"MEMBERSHIP INTEREST" as to any Member means the entire ownership interest and rights of that Member in the Company, including, without limitation, its right to a distributive share of the profits and losses of the Company, its right to a distributive share of the assets of the Company in accordance with the provisions hereof, and its right to participate in the management of the affairs of the Company.

-4-

"OPERATING PROCEDURES" means the Operating Procedures in the form attached as Exhibit C to the Partnership Agreement.

"PARENT" means, with respect to TE Member, TEPPCO Partners, L.P., a Delaware limited partnership, and with respect to LD Member, S. A. Dreyfus et Cie., a French societe anonyme.

"PARTNERSHIP" means Mont Belvieu Storage Partners, L.P., a Delaware limited partnership in which the Company is the general partner.

"PARTNERSHIP AGREEMENT" means the Agreement of Limited Partnership of the Partnership as the same may be amended, modified or restated from time to time.

"PARTNERSHIP BUSINESS" means the operation, maintenance and marketing of underground storage facilities for Products and the operation of the pipelines included within the MB Assets to and from their specific points of delivery and receipt.

"PRIME RATE" has the meaning ascribed thereto in Section 9.4.

"PRODUCTS" means propane, normal butane, isobutane, natural gasoline and other commodities as agreed by the Management Committee.

"REPRESENTATIVE" has the meaning given such term in Section 15.2(a).

"RESTRICTED AFFILIATE" means with respect to each Member, any person in which such person's Parent owns, directly or indirectly, at least a 51% voting or equity interest, or which it otherwise controls. For purposes of this definition, "control," has the meaning assigned to such term in the definition of Affiliate and Louis Dreyfus Natural Gas Corporation shall not be deemed a Restricted Affiliate of the LD Member.

"SHARING RATIO" means the percentages described in Section 8.1.

"TE MEMBER" means TE Products Pipeline Company, Limited Partnership or its successors or assigns.

"TE MEMBER CHANGE OF CONTROL" means a Disposition of any entity of which over 75% of such entity's EBITDA for the preceding 12-month period is derived from the Membership Interest unless such disposition is to an Affiliate of the TE Member.

"TMP" has the meaning ascribed thereto in Section 10.5.

"UNREALIZED GAIN" attributable to a Company property means, as of the date of determination, the excess of the fair market value of such property as of such date of determination over the Carrying Value of such property as of such date of determination.

-5-

"UNREALIZED LOSS" attributable to a Company property means, as of the date of such determination, the excess of the Carrying Value of such property as of such date of determination over the fair market value of such property as of such date of determination.

ARTICLE I

FORMATION OF LIMITED LIABILITY COMPANY

SECTION 1.1 FORMATION. The Members hereby form a limited liability company pursuant to the Act. The rights and liabilities of the Members shall be as provided in the Act, except as herein otherwise expressly provided. The Membership Interests of any Member shall be personal property for all purposes. Each Member shall execute, acknowledge, swear to, and deliver all certificates and other instruments conforming with this Agreement that are necessary to qualify, continue, or terminate the Company as a limited liability company under the laws of the State of Delaware and to qualify the Company to do business in such other states or other jurisdictions where such qualification is necessary or desirable.

SECTION 1.2 PARTNERSHIP FOR TAX PURPOSES ONLY. The Members intend that the Company not be a partnership (including a limited partnership) or joint venture, and that no Member be a partner or joint venturer of any other Member, for any purposes other than U.S. federal and state tax purposes, and this Agreement may not be construed to suggest otherwise.

ARTICLE II

NAME

The name of the Company shall be, and the business of the Company shall be conducted under the name of, Mont Belvieu Venture, LLC, or such other name or names that comply with applicable law as the Members may designate from time to time. The Managing Member shall take any action that it determines is required to comply with the Act, assumed name act, fictitious name act, or similar statute in effect in each jurisdiction or political subdivision in which the Company proposes to do business.

ARTICLE III

PURPOSE

The purpose of the Company is to serve as the general partner of the Partnership and to engage in any and all activities necessary, incidental or ancillary to the foregoing.

-6-

ARTICLE IV

NAMES AND ADDRESSES OF MEMBERS

The names and mailing addresses of the Members are as set forth on the signature pages hereof.

ARTICLE V

REGISTERED AGENT; REGISTERED OFFICE; PRINCIPAL OFFICE; ADDITIONAL OFFICES

The name and address of the registered office of the Company in the State of Delaware is c/o The Corporation Trust Company, Corporation Trust Center, 1209 Orange Street, Wilmington, New Castle County, Delaware 19801. The principal office of the Company shall be 2929 Allen Parkway, Suite 3200, Houston, Texas 77019. The name and address of the registered agent for service of process on the Company in the State of Delaware is The Corporation Trust Company, Corporation Trust Center, 1209 Orange Street, Wilmington, New Castle County, Delaware 19801. The Members may change the registered agent or the registered office of the Company and may establish such additional offices of the Company as they may, from time to time determine.

ARTICLE VI

TERM

The term of the Company shall be perpetual from the date of the filing of the Certificate in the Office of the Secretary of State of the State of Delaware unless extended or sooner liquidated or dissolved in accordance with this Agreement.

ARTICLE VII

ADDITIONAL MEMBERS

In the event the Management Committee determines that funds in addition to those acquired pursuant to Section 8.1 are necessary to carry out the purposes of the Company and the Members do not agree to contribute such funds in their respective Sharing Ratios, the Management Committee is authorized to offer and sell additional Membership Interests and admit any purchasers thereof (which may include existing Members or their Affiliates) as additional Members of the Company. In such event, the dilution of the then Members shall be pro rata in accordance with their respective Sharing Ratios prior to the issuance of such additional Membership Interests and this Agreement shall be amended to reflect the admissions of such persons as Members of the Company, the revised ownership and Sharing Ratio of each Member and otherwise to reflect the relative rights and obligations of the parties.

-7-

ARTICLE VIII

CAPITAL CONTRIBUTIONS

SECTION 8.1 INITIAL CONTRIBUTIONS.

(a) The Members shall contribute the amounts of cash set forth below.

Capital Sharing Member Contribution Ratio ------ ---------- TE Member \$250,000 50% LD Member \$250,000 50% ------ --- Total \$500,000 100% =======

===

(b) Without creating any rights in favor of any third party, each Member shall contribute to the Company, in cash, on or before the date specified as described in this Section 8.1(b), that Member's pro rata share of all monies that in the judgment of the Management Committee are necessary to enable the Company to cause the assets of the Company to be properly operated and maintained and to discharge its costs, expenses, obligations, and liabilities (including any capital expenditures for new assets, businesses or opportunities); provided, however, that subject to the nonwaivable requirements of law, a Member is not obligated to contribute a total amount during any fiscal year that exceeds its pro rata share of the contributions contemplated in the Approved Business Plan in effect for such year. The Managing Member shall notify each Member of the need for Capital Contributions pursuant to this Section 8.1(b), which notice must include a date (which date may be no earlier than the second Business Day following each Member's receipt of its notice) before which the Capital Contributions must be made. Notices for Capital Contributions must be made to all Members in accordance with their Sharing Ratios.

(c) If a Member does not contribute by the time required all or any portion of a Capital Contribution that Member is required to make as provided in this Agreement, the Company may exercise, on notice to that Member (the "DELINQUENT MEMBER"), one or more of the following remedies:

> (i) taking such action (including court proceedings) as the Management Committee may deem appropriate to obtain payment by the Delinquent Member of the portion of the Delinquent Member's Capital Contribution that is in default, together with interest on that amount at the Default Interest Rate from the date that the Capital Contribution was due until the date that it is made, all at the cost and expense of the Delinquent Member;

> > -8-

(ii) permitting the other Member (the "LENDING MEMBER") to advance the portion of the Delinquent Member's Capital Contribution that is in default, with the following results:

> 1. the sum advanced shall constitute a loan from the Lending Member to the Delinquent Member and a Capital Contribution of that sum to the Company by the Delinquent Member under the applicable provisions of this Agreement,

> 2. the principal balance of the loan and all accrued unpaid interest is due and payable on the tenth day after written demand by the Lending Member to the Delinquent Member,

3. the amount loaned shall bear interest at the Default Interest Rate from the date that the advance is deemed made until the date that the loan, together with all interest accrued on it, is repaid to the Lending Member,

4. all distributions from the Company that otherwise would be made to the Delinquent Member (whether before or after dissolution of the Company) instead shall be paid to the Lending Member until the loan and all interest accrued on it have been paid in full to the Lending Member (with payments being applied first to accrued and unpaid interest and then to principal), but all such payments to the Lending Member shall be treated for all purposes of this Agreement as a distribution by the Company to the Delinquent Member and a payment by the Delinquent Member to the Lending Member,

5. the payment of the loan and interest accrued on it is secured by a security interest in the Delinquent Member's Membership Interest, as more fully set forth in Section 8.1(d), and

6. the Lending Member has the right, in addition to the other rights and remedies granted to it under this Agreement or at law or in equity, to take any action (including court proceedings) that the Lending Member may deem appropriate to obtain payment by the Delinquent Member of the loan and all accrued and unpaid interest on it, at the cost and expense of the Delinquent Member;

(iii) exercising the rights of a secured party under the Uniform Commercial Code of the State of Delaware, as more fully set forth in Section 8.1(d); or

(iv) exercising any other rights and remedies available at law or in equity.

-9-

(d) Each Member grants to the Company, and to each Lending Member with respect to any loans made by the Lending Member to that Member as a Delinquent Member as described in Section 8.1(c), as security, equally and ratably, for the payment of all Capital Contributions that Member has agreed to make and the payment of all loans and interest accrued on them made by Lending Members to that Member as a Delinquent Member as described in Section 8.1(c), a security interest in and a general lien on its Membership Interest and the proceeds of that Membership Interest, all under the Uniform Commercial Code of the State of Delaware. On any default in the payment of a Capital Contribution or in the payment of such a loan or interest accrued on it, the Company or the Lending Member, as applicable, is entitled to all the rights and remedies of a secured party under the Uniform Commercial Code of the State of Delaware with respect to the security interest granted in this Section 8.1(d). Each Member shall execute and deliver to the Company and the other Member all financing statements and other instruments that the Management Committee or the Lending Member, as applicable, may request to effectuate and carry out the preceding provisions of this Section 8.1(d). At the option of a Lending Member, this Agreement or a photographic or other copy of this Agreement may serve as a financing statement.

SECTION 8.2 ADDITIONAL CONTRIBUTIONS. The liability of each Member to the Company shall be limited to the amount of its Capital Contribution made and to be made pursuant to Section 8.1, and no Member shall have any further personal liability to contribute money to, or in respect of, the liabilities or the obligations of the Company unless it agrees in writing to make additional Capital Contributions to the Company, nor shall any Member be personally liable for any obligations of the Company, except as may be provided in the Act. No Member shall be entitled to the return of any part of its Capital Contribution or to be paid interest in respect of either its Capital Account or any Capital Contribution made by such Member. No unrepaid Capital Contribution shall be deemed or considered to be a liability of the Company or any Member. No Member shall be required to contribute or lend any cash or property to the Company to enable the Company to return any Member's Capital Contribution to the Member.

SECTION 8.3 CAPITAL ACCOUNTS. A capital account ("CAPITAL ACCOUNT") shall be established for each Member and shall be maintained in such a manner as to correspond with the requirements of Treasury Regulations promulgated from time to time under section 704(b) of the Code. The respective Capital Accounts of the Members shall not bear interest.

SECTION 8.4 ADJUSTMENT OF CAPITAL ACCOUNTS. If any Company property is to be distributed in liquidation of the Company or a Membership Interest, the Capital Accounts of the Members (and the amounts at which all Company properties are carried on its books and records) shall, immediately prior to such issuance or distribution, as the case may be, be adjusted (consistent with the provisions of section 704(b) of the Code and the Treasury Regulations promulgated thereunder) upward or downward to reflect any Unrealized Gain or Unrealized Loss attributable to all Company properties (as if such Unrealized Gain or Unrealized Loss had been recognized upon actual sale of such properties upon a liquidation of the Company is properly reflected on the books of the Company at a value that differs from the adjusted tax basis of such property, this Section 8.4 shall be applied with reference to such value.

-10-

ARTICLE IX

DISTRIBUTIONS

SECTION 9.1 AMOUNT; TIMING. Except as otherwise provided herein or by the Act, Distributable Cash shall be distributed among the Members pro rata in accordance with their Sharing Ratios in such aggregate amounts and at such times as shall be determined by the Management Committee.

SECTION 9.2 FAILURE TO WITHDRAW. If any Member does not withdraw the whole or any part of its share of any cash distribution made pursuant to Section 9.1, such Member shall not be entitled to receive any interest thereon without the express written consent of the other Member.

SECTION 9.3 TRANSFEROR/TRANSFEREE ALLOCATIONS. Unless otherwise agreed in writing by a transferor and transferee of a Membership Interest herein, Distributable Cash distributable with respect to any Membership Interest which may have been transferred during any year shall be distributed to the holder of such Membership Interest who was recognized as the owner on the date of such distribution, without regard to the results of Company operations during the year.

SECTION 9.4 MEMBER ADVANCES. Notwithstanding the foregoing, if any Member advances any funds or makes any other payment to or on behalf of the Company, not required pursuant to the provisions hereof, to cover operating or capital expenses of the Company which cannot be paid out of the Company's operating revenues, such advance or payment shall be deemed a loan to the Company by such Member, bearing interest from the date such advance or payment was made until such loan is repaid at a floating rate per annum equal to the lesser of (i) two percent (2%) over the interest rate publicly quoted by The Wall Street Journal from time to time as the prime rate, with adjustments in such varying rate to be made on the same date as any change in the aforesaid rate (herein called "PRIME RATE") or (ii) the maximum rate permitted under applicable law. Notwithstanding Section 9.1 above, all distributions of Distributable Cash shall first be distributed to the Members making such loans until all such loans have been repaid to such Members, together with interest thereon as above provided, and, thereafter, the balance of such distributions, if any, shall be made in accordance with the terms of Section 9.1 above. If distributions are insufficient to repay and return all such loans as provided above, the funds available from time to time shall first be applied to repay and retire the oldest loan first and, if any funds thereafter remain available, such funds shall be applied in a similar manner to remaining loans in accordance with the order of the dates on which they were made; however, as to loans made on the same date, each such loan shall be repaid pro rata in the proportion that such loan bears to the total loans made on said date.

ARTICLE X

ALLOCATIONS OF INCOME, GAIN, LOSS, DEDUCTION AND CREDIT

SECTION 10.1 GENERAL. Except as otherwise provided herein or unless another allocation is required by Treasury Regulations issued under section 704(b) of the Code (including, but not limited to, provisions pertaining to qualified income offsets and minimum gain chargebacks) for

-11-

purposes of maintaining the Capital Accounts, all items of Company income, gain, loss, deduction and credit shall be allocated among the Members pro rata in accordance with their Sharing Ratios in effect for the period during which such items accrue. For purposes of computing the amount of each item of income, gain, deduction or loss to be charged or credited to the Capital Accounts, the determination, recognition and classification of such item shall be the same as its determination, recognition and classification for federal income tax purposes, provided that:

> (a) Any deductions for depreciation, cost recovery, or amortization attributable to any Company property shall be determined as if the adjusted basis of such property were equal to the Carrying Value of such property. Upon an adjustment to the Carrying Value of any Company property subject to depreciation, cost recovery, or amortization pursuant to Section 8.4, any further deductions for such depreciation, cost recovery, or amortization attributable to such property shall be determined as if the adjusted basis of such property were equal to the Carrying Value of such property immediately following such adjustment.

> (b) Any income, gain or loss attributable to the taxable disposition of any Company property shall be determined by the Company as if the adjusted basis of such property as of such date of disposition were equal in amount to the Carrying Value of such property as of such date.

> (c) All fees and other expenses incurred by the Company to promote the sale of a Membership Interest that can neither be deducted nor amortized under section 709 of the Code shall be treated as an item of deduction.

> (d) Computation of all items of income, gain, loss and deduction shall be made without regard to any election under section 754 of the Code which may be made by the Company and, as to those items described in section 705(a)(1)(B) or section 705(a)(2)(B) of the Code, without regard to the fact that such items are not includable in gross income or are neither currently deductible nor capitalizable for federal income tax purposes.

SECTION 10.2 ALLOCATIONS FOR TAX PURPOSES.

(a) The Company shall, except to the extent such item is subject to allocation pursuant to subsection (b) below, allocate each item of income, gain, loss, deduction and credit, as determined for federal and other income tax purposes, in the same manner as such item was allocated for book purposes; and

(b) The Company, for federal and other income tax purposes shall, in the case of Contributed Properties, allocate items of income, gain, loss, depreciation and cost recovery deductions attributable to those properties with a Built-In Gain or a Built-In Loss pursuant to section 704(c) of the Code. Similar allocations shall be made in the event that the Carrying Value of Company properties subject to depreciation, cost recovery or amortization are adjusted pursuant to Section 8.4 upon the issuance of Membership Interests for cash. If an existing Member acquires additional Membership Interests, such allocations shall apply only to the extent of his or its additional Membership Interests. No allocation under section 704(c) of the Code shall be charged or credited to a Member's Capital Account.

-12-

SECTION 10.3 TRANSFEROR/TRANSFEREE ALLOCATIONS. Income, gain, loss, deduction or credit attributable to any Membership Interest which has been transferred shall be allocated between the transferor and the transferee allocated equally among the days of the Company's fiscal year without regard to Company operations during such days.

SECTION 10.4 RELIANCE ON ADVISORS. The Managing Member may rely upon, and shall have no liability to the Members or the Company if it does rely upon, the written opinion of tax counsel or accountants retained by the Company from time to time with respect to all matters (including disputes with respect thereto) relating to computations and determinations required to be made under this Article X or other provisions of this Agreement.

SECTION 10.5 TAX MATTERS PARTNER.

(a) The Managing Member is designated tax matters partner ("TMP") as defined in section 6231(a)(7) of the Code. The TMP and the other Member shall use their best efforts to comply with responsibilities outlined in this Section 10.5 and in sections 6222 through 6232 of the Code (including any Treasury Regulations promulgated thereunder) and in doing so shall incur no liability to any other Member.

(b) If any Member intends to file a notice of inconsistent treatment under section 6222(b) of the Code, such Member shall, prior to the filing of such notice, notify the TMP of such intent and the manner in which the Member's intended treatment of a Company item is (or may be) inconsistent with the treatment of that item by the Company.

(c) No Member other than the TMP shall file a request pursuant to section 6227 of the Code for an administrative adjustment of partnership items for any Company taxable year.

(d) No Member other than the TMP shall file a petition under Code sections 6226, 6228 or other Code sections with respect to any Company item, or other tax matters involving the Company. In the case where the TMP files such petition, he shall determine the forum in which such petition will be filed.

ARTICLE XI

BOOKS OF ACCOUNT, RECORDS AND TAX INFORMATION

SECTION 11.1 MAINTENANCE OF BOOKS AND RECORDS. Proper and complete records and books of account (including those required by the Act) shall be kept by the Company in which shall be entered all transactions and other matters relative to the Company's business as are usually entered into records and books of account maintained by persons engaged in businesses of like character. The Company books and records necessary for purposes of maintaining the Capital Accounts and giving effect to the allocations provided in Articles VIII and X shall be maintained in accordance with the accounting principles described therein, and shall be kept on the accrual basis and the other books and records shall be kept on the accrual basis in accordance with GAAP for

-13-

financial reporting purposes and for purposes of making all other calculations contemplated herein. The books and records shall at all times be made available and shall be open to the reasonable inspection and examination of the Members or their duly authorized representatives during the business hours of the Company for any purpose reasonably related to the interest of such Member as a member of the Company.

SECTION 11.2 REPORTS.

(a) As soon as reasonably practicable after the end of each fiscal year, the Company shall send each person who was a holder of a Membership Interest at any time during the fiscal year then ended all Company tax information as shall be necessary for the preparation by such holder of its federal and any applicable state and local income tax returns.

ARTICLE XII

FISCAL YEAR

The fiscal year of the Company shall end on the thirty-first (31st) day of December in each year.

ARTICLE XIII

COMPANY FUNDS

The funds of the Company shall be deposited in such bank account or accounts, or invested in such interest-bearing or non-interest-bearing accounts, as shall be designated by the Managing Member. All withdrawals from any such bank accounts shall be made by Company employees or agents duly authorized by the Managing Member.

ARTICLE XIV

STATUS OF MEMBERS

No Member shall have any personal liability whatever, whether to the Company, to any of the Members or to the creditors of the Company, for the debts of the Company or any of its losses beyond the amount agreed to be contributed by it to the capital of the Company as set forth in Section 8.1 except to the extent required by the Act. No Member shall be obligated to restore any deficit in its Capital Account upon the liquidation of the Company or its Membership Interest.

-14-

ARTICLE XV

MANAGEMENT AND OPERATION OF BUSINESS

SECTION 15.1 MEMBER MANAGEMENT. Except as otherwise provided herein, the management of the Company is fully vested in the Members, acting exclusively in their membership capacities. To facilitate the orderly and efficient management of the Company, the Members shall act (i) collectively as a "committee of the whole" (named the Management Committee) pursuant to Section 15.2, and (ii) through the delegation of certain responsibility and authority to the Managing Member pursuant to Section 15.4 and the LD Member pursuant to Section 15.5. The Company will not have "managers," as that term is used in the Act, it being understood that the Representatives and Alternate Representatives do not constitute "managers."

SECTION 15.2 MANAGEMENT COMMITTEE. The Members shall act collectively through meetings as a "committee of the whole," which is hereby named the "MANAGEMENT COMMITTEE." Decisions or actions taken by the Management Committee in accordance with the provisions of this Agreement shall constitute decisions or actions by the Company and shall be binding on each Member of the Company. The Management Committee shall conduct its affairs in accordance with the following provisions and the other provisions of this Agreement:

(a) Organization of Management Committee:

(i) The Management Committee shall be composed of five members.

(ii) Each Member shall appoint two individuals to represent it on the Management Committee (individually, such Member's "REPRESENTATIVE" and collectively, such Member's "REPRESENTATIVES"). No individual may serve as the Representative of more than one Member. Each Member shall also appoint one or more individuals ("ALTERNATE REPRESENTATIVES") with the power of substitution and authority to act in place of its Representatives in case of the unavailability thereof. Each Representative and Alternate Representative shall be an officer or agent of the Member appointing him or her and shall be duly authorized to act on behalf of and to bind the appointing Member. Each Member reserves the right to remove any one or more of its Representatives or Alternate Representatives, as the case may be, and to appoint successors and substitutes therefor, from time to time, and any such change shall be effective upon such Member's delivering a written notice of such change to the other Member.

(iii) The individual serving as the president/chief executive officer of the Company shall be the fifth member of the Management Committee; however, that individual shall be a nonvoting member.

(iv) Notwithstanding the number of Representatives and Alternate Representatives, each Member shall have one vote on all matters to be decided by the Management Committee and the Representatives or Alternate Representatives

-15-

of any Member shall vote as a single bloc on all such matters. Voting may occur by voice vote at a meeting of the committee or by written consent.

(v) Any action of the Management Committee shall require the affirmative vote of both Members. Notwithstanding the above, if any Member is a Delinquent Member as described in Section 8.1(c) (or it or its Affiliate is a Delinquent Partner in the Partnership), its Representatives and Alternate Representative(s) may not participate in the Management Committee and all decisions shall be made by the Representatives or Alternate Representative(s) of the other Member.

(b) Matters Requiring Management Committee Consent: Notwithstanding any other provision of this Agreement, the following actions require the unanimous consent of the Management Committee:

(i) the annual Business Plan and any amendments to the Approved Business Plan;

(ii) the incurrence of, or commitment to incur, any capital cost for any Partnership project that exceeds \$250,000 (to the extent not covered by the Approved Business Plan) or that would cause the Partnership to exceed its annual capital budget as reflected on the Approved Business Plan;

(iii) the sale or divestiture of any asset of the Partnership having a fair market value in excess of \$250,000, to the extent not covered in the Approved Business Plan;

(iv) borrowings or issuances of debt securities (excluding trade credits and advances under working capital funding facilities) of the Company or the Partnership, to the extent these matters are not covered in the Approved Business Plan;

 (ν) issuances of any new interests in the Company or the Partnership;

(vi) entering into or amending contracts between the Company or the Partnership and any Member or its Affiliates; and

(vii) the filing or settlement of any lawsuit by the Company or the Partnership involving a claim or settlement payment of more than \$500,000.

(c) Meetings of the Management Committee:

(i) Regular meetings of the Management Committee shall be held periodically, but no less frequently than quarterly, on such dates, at such times and at such locations as the members of such committee shall from time to time determine, taking into account the convenience of all parties. The individual then

-16-

serving as the president/chief executive officer of the Company or any Representative or Alternate Representative may call a special meeting of the Management Committee. Notice of any special meeting shall include a statement of the matters proposed to be considered at such meeting and shall be given to all participants by the Person calling the meeting, under normal circumstances at least ten Business Days prior to the meeting, although shorter notice of a meeting (but not less than 24 hours) may be given if the circumstances of urgency so require. All notices of Management Committee meetings shall be given either in writing, or by telephone if immediately followed by written confirmation, and no Management Committee meeting shall be held unless the president/chief executive officer of the Company and a Representative or Alternate Representative of each Member participates in such meeting. Each Member agrees to use reasonable efforts to cause at least one of its Representatives or an Alternate Representative to participate, in the manner provided for herein, in all Management Committee meetings.

(ii) Representatives and Alternate Representatives may participate in any Management Committee meeting by means of telephone conference call or similar communications equipment so long as all Persons participating in the meeting can hear each other simultaneously. If required, the Management Committee may act without a meeting by written consent.

(iii) The individual serving as the president/chief executive officer of the Company shall preside at all Management Committee meetings.

SECTION 15.3 COMPENSATION AND REIMBURSEMENT. The Management Committee is not entitled to compensation for its services. Each Representative and Alternate Representative shall be reimbursed by their respective Member for out-of-pocket costs and expenses incurred in serving on the Management Committee.

SECTION 15.4 MANAGING MEMBER. The Members hereby delegate the authority described in Section 15.4(b) to the Member that is designated as the "MANAGING MEMBER" in accordance with this Section 15.4. Decisions or actions taken by the Managing Member in accordance with the provisions of this Agreement and the Operating Procedures shall constitute decisions or actions by the Company and shall be binding on each Member. The Managing Member shall be designated, and shall exercise such delegated authority, in accordance with the following provisions and the other provisions of this Agreement:

(a) Designation and Removal. The initial Managing Member is the TE Member. It, and any successor Managing Member that is designated in accordance with this Section 15.4(a), shall cease to be the Managing Member upon the earlier to occur of the following events: (i) it shall Dispose of all of its Membership Interest or (ii) it shall become a Bankrupt Member. Upon the occurrence of either of the events described in the immediately-preceding sentence, the Management Committee (excluding, in the case where the Managing Member is a Bankrupt Member, the votes of the Representatives appointed by that Managing Member) shall designate another Member (that consents to serve as such) as a successor Managing Member.

-17-

(b) Delegation of Authority. Except for situations in which the approval of the Management Committee is expressly required by this Agreement, the Partnership Agreement, or the Operating Procedures, the Managing Member shall have full, complete, and exclusive authority to manage and control the business, affairs, and properties of the Company and the Partnership, to make all decisions regarding those matters, and to perform any and all other acts or activities customary or incident to the management of the Company's business, including the following:

> (i) entering into, making, and performing contracts, agreements, and other undertakings binding the Company and the Partnership that may be necessary, appropriate, or advisable in furtherance of the purposes of the Company and the Partnership and making all decisions and waivers under the foregoing;

(ii) opening and maintaining bank accounts and arrangements for the Company and the Partnership, drawing checks and other orders for the payment of money, and designating individuals with authority to sign or give instructions with respect to those accounts and arrangements;

(iii) maintaining the assets of the Company and the Partnership in good order;

(iv) collecting sums due the Company and the Partnership;

(v) to the extent that funds of the Company and the Partnership are available, paying debts and obligations of the Company and the Partnership;

(vi) acquiring, utilizing for Company and Partnership purposes, and disposing of any asset of the Company and Partnership;

(vii) borrowing money or otherwise committing the credit of the Company and the Partnership for Company and Partnership activities and voluntarily prepaying or extending debt;

(viii) selecting, removing, and changing the authority and responsibility of lawyers, accountants, and other advisers and consultants; and

(ix) obtaining insurance for the Company and the Partnership.

(c) To expedite the handling of Company business, it is understood and agreed that any document executed by the Managing Member shall, as to any third parties, be deemed to be the action of the Company. Further, any person dealing with the Company or the Partnership may rely upon a certificate signed by the Managing Member as to:

> (i) the identity of the Members, their Representatives and Alternate Representatives;

> > -18-

(ii) the existence or nonexistence of any fact or facts that constitute conditions precedent to acts by the Company or the Partnership or are in any other manner related to the affairs of the Company or the Partnership;

(iii) the persons who are authorized to execute and deliver any instrument or document of the Company or the Partnership;

(iv) any act or failure to act by the Company or the Management Committee; or

 (ν) any other matter whatsoever involving the Company or the Partnership.

SECTION 15.5 MARKETING ACTIVITIES.

(a) The LD Member shall have the right (i) to carry out the marketing of the Partnership Business and manage the personnel involved in those marketing activities, and (ii) to make proposals to the Managing Member for Discretionary Capital Expenditures and the development and use of the Partnership's assets, in each case with the purpose of increasing the EBITDA of the Partnership Business. The LD Member will use commercially reasonable efforts in carrying out such activities. The LD Member's activities shall include soliciting new users of the Partnership's assets or increased use by existing customers and investigating and proposing capital projects for the Partnership Business. At least 120 days prior to the beginning of each Fiscal Year, the LD Member will provide the Managing Member with a forecast of revenue for such Fiscal Year. On or before January 15, April 15, July 15 and October 15 of each Fiscal Year, the LD Member will provide a report to the Managing Member on the LD Member's activities for the three month period ending on the last day of the immediately preceding month including, among other things, the status of contracts entered into during such period, potential contracts being explored, performance for that period and revisions, if any, to the revenue forecast for the Fiscal Year in which such three-month period falls. The Managing Member will keep the LD Member advised of any storage contracts or renewals thereof entered into by The Managing Member and will seek the LD Member's advice with respect thereto. The LD Member will commit the equivalent of at least one full-time person to the marketing activities to be carried out under this Section 15.5, and the Managing Member will commit the equivalent of at least one part-time person to aid the LD Member in those marketing activities under the LD Member's direction. The cost of the compensation and benefits of all such employees shall not be included in the expenses of the Partnership but shall be borne by the LD Member or the Managing Member respectively as the employer of that person. The Managing Member and the LD Member will agree on whether the cost of the compensation and benefits of any additional employees of either party who may be engaged in the marketing activities shall be included in the expenses of the Partnership.

(b) The Members understand and agree all storage, service and other agreements relating to the Partnership's assets and the Partnership's Business shall be subject to the

-19-

approval of, and signed by the Managing Member or one of its Affiliates. Neither Member shall have the authority, nor shall it attempt, to bind the other Member in any regard. The LD Member will keep the Managing Member informed of its marketing activities and provide to the Managing Member such information as it may reasonably request with respect to customers and potential customers of the Partnership. The Managing Member will provide all reasonable assistance in the LD Member's marketing and sales efforts and shall make available appropriate authorized representatives to execute any storage agreements developed by the LD Member. All such contracts shall become the property of the Partnership.

SECTION 15.6 EXCULPATION. NEITHER THE MANAGING MEMBER, THE MANAGEMENT COMMITTEE, THE OTHER MEMBER, THEIR RESPECTIVE AFFILIATES, NOR ANY OWNER, OFFICER, DIRECTOR, PARTNER, EMPLOYEE OR AGENT OF THE MEMBERS OR THEIR RESPECTIVE AFFILIATES, SHALL BE LIABLE, RESPONSIBLE OR ACCOUNTABLE IN DAMAGES OR OTHERWISE TO THE COMPANY OR ANY MEMBER FOR ANY ACTION TAKEN OR FAILURE TO ACT (EVEN IF SUCH ACTION OR FAILURE TO ACT CONSTITUTED THE NEGLIGENCE OF SUCH PERSON) ON BEHALF OF THE COMPANY OR THE PARTNERSHIP WITHIN THE SCOPE OF THE AUTHORITY CONFERRED ON SUCH PERSON PURSUANT TO THIS AGREEMENT OR THE PARTNERSHIP AGREEMENT UNLESS SUCH ACT OR OMISSION WAS PERFORMED OR OMITTED FRAUDULENTLY OR CONSTITUTED GROSS NEGLIGENCE OR WILLFUL MISCONDUCT. TO THE EXTENT THAT, AT LAW OR IN EQUITY, THE MANAGEMENT COMMITTEE, THE MEMBERS, THEIR RESPECTIVE AFFILIATES, OR ANY OWNER, OFFICER, DIRECTOR, PARTNER, EMPLOYEE OR AGENT THEREOF HAVE DUTIES (INCLUDING FIDUCIARY DUTIES) AND LIABILITIES RELATING TO THE COMPANY OR TO ANOTHER MEMBER, THE MANAGEMENT COMMITTEE, THE MEMBERS, THEIR RESPECTIVE AFFILIATES, OR ANY OWNER, OFFICER, DIRECTOR, PARTNER, EMPLOYEE OR AGENT THEREOF ACTING UNDER THE AGREEMENT, THE PARTNERSHIP AGREEMENT OR THE OPERATING PROCEDURES SHALL NOT BE LIABLE TO THE COMPANY OR TO ANY OTHER MEMBER OR ITS AFFILIATES FOR THEIR RELIANCE ON THE PROVISIONS OF THIS AGREEMENT. THE PROVISIONS OF THIS AGREEMENT, TO THE EXTENT THAT THEY EXPAND OR RESTRICT THE DUTIES AND LIABILITIES OF THE MANAGEMENT COMMITTEE, THE MEMBERS, THEIR RESPECTIVE AFFILIATES, OR ANY OWNER, OFFICER, DIRECTOR, PARTNER, EMPLOYEE OR AGENT THEREOF OTHERWISE EXISTING AT LAW OR IN EQUITY, ARE AGREED BY THE MEMBERS TO REPLACE SUCH OTHER DUTIES AND LIABILITIES OF THE MANAGEMENT COMMITTEE, THE MEMBERS, THEIR RESPECTIVE AFFILIATES, OR ANY OWNER, OFFICER, DIRECTOR, PARTNER, EMPLOYEE OR AGENT THEREOF.

SECTION 15.7 INDEMNIFICATION.

(a) TO THE FULLEST EXTENT PERMITTED BY LAW, THE MANAGING MEMBER, THE MANAGEMENT COMMITTEE, THE OTHER MEMBER, THEIR RESPECTIVE AFFILIATES AND THEIR RESPECTIVE OFFICERS, DIRECTORS, PARTNERS, EMPLOYEES AND AGENTS OR ANY PERSON PERFORMING A SIMILAR FUNCTION (INDIVIDUALLY, AN "INDEMNITEE") SHALL BE INDEMNIFIED AND HELD HARMLESS BY THE COMPANY FROM AND AGAINST ANY AND ALL LOSSES, CLAIMS, DAMAGES, JUDGMENTS, LIABILITIES, OBLIGATIONS, PENALTIES, SETTLEMENTS AND REASONABLE EXPENSES (INCLUDING LEGAL FEES) ARISING FROM ANY AND ALL CLAIMS, DEMANDS, ACTIONS, SUITS OR PROCEEDINGS, CIVIL, CRIMINAL, ADMINISTRATIVE OR INVESTIGATIVE, IN WHICH THE INDEMNITEE MAY BE INVOLVED, OR THREATENED TO BE INVOLVED, AS A PARTY OR OTHERWISE, BY REASON OF ITS STATUS AS (x) A MEMBER OF THE

-20-

MANAGEMENT COMMITTEE, A MEMBER OR AN AFFILIATE THEREOF, OR (y) AN OFFICER, DIRECTOR, PARTNER, EMPLOYEE OR AGENT OF A MEMBER OR AN AFFILIATE THEREOF, REGARDLESS OF WHETHER THE INDEMNITEE CONTINUES TO BE A MEMBER OF THE MANAGEMENT COMMITTEE, A MEMBER OR AN AFFILIATE THEREOF OR AN OFFICER, DIRECTOR, EMPLOYEE OR AGENT OF A MEMBER OR AN AFFILIATE THEREOF AT THE TIME ANY SUCH LIABILITY OR EXPENSE IS PAID OR INCURRED, UNLESS THE ACT OR FAILURE TO ACT GIVING RISE TO INDEMNITY HEREUNDER WAS PERFORMED OR OMITTED FRAUDULENTLY OR CONSTITUTED GROSS NEGLIGENCE OR WILLFUL MISCONDUCT.

(b) The Company may purchase and maintain insurance on behalf of the Members, the Management Committee and such other persons as the Management Committee shall determine against any liability that may be asserted against or expense that may be incurred by such Person in connection with the Company's activities, regardless of whether the Company would have the power to indemnify such person against such liability under the provisions of this Agreement.

(c) Expenses incurred by any Indemnitee in defending any claim with respect to which such Indemnitee may be entitled to indemnification by the Company hereunder (including without limitation reasonable attorneys' fees and disbursements) shall, to the maximum extent permitted by law, be advanced by the Company prior to the final disposition of such claim, upon receipt of a written undertaking by or on behalf of such Indemnitee to repay the advanced amount of such expenses unless it is determined ultimately that the Indemnitee is entitled to indemnification by the Company under Section 15.7(a).

(d) The indemnification provided in this Section 15.7 is for the benefit of the Indemnitees and shall not be deemed to create any right to indemnification for any other Persons.

SECTION 15.8 OTHER ACTIVITIES; NONCOMPETITION.

Each Member agrees that the terms of Section 15.8 of the Partnership Agreement shall also be applicable to the Members during the term of this Agreement with respect to the Partnership Business.

ARTICLE XVI

TRANSFER OF MEMBERSHIP INTERESTS

SECTION 16.1 RESTRICTIONS ON TRANSFER. No Member shall have the right to Dispose of its interest in the Company, or any portion thereof, without (i) simultaneously Disposing of its and any of its Affiliates' entire interest in the Partnership pursuant to the provisions of Article XVI of the Partnership Agreement, and (ii) Disposing of its entire Membership Interest to a single purchaser or transferee. Accordingly, any Disposition of the Member's interest in the Company shall be in compliance with Article XVI of the Partnership Agreement, and such Article XVI shall be binding on the parties to this Agreement as if fully set out herein with application to the Membership Interests in the Company rather than Partnership Interests in the Partnership.

-21-

ARTICLE XVII

DISSOLUTION OF THE COMPANY

The happening of any one of the following events shall work an immediate dissolution of the Company:

(a) The resignation, expulsion, bankruptcy or dissolution of a Member unless the business of the Company is continued by the written consent of the remaining Member within 90 days following the occurrence of any such event;

(b) The receipt by the Company of the final payment due on the sales price of all or substantially all the assets of the Company or the Company's business following the Company's sale thereof;

(c) The agreement by all of the Members to dissolve; and

(d) The entry of a decree of judicial dissolution under Section 18-802 of the Act.

Each Member covenants and agrees that it will not retire, resign or voluntarily dissolve, without the prior written consent of the other Member.

ARTICLE XVIII

WINDING UP AND TERMINATION OF THE COMPANY

SECTION 18.1 LIQUIDATOR. If the Company is dissolved for any reason, a liquidator (the "LIQUIDATOR") shall commence to wind up the affairs of the Company and to liquidate and sell its assets. The Members shall serve as the Liquidator unless they appoint another person to serve as the Liquidator. The Liquidator shall have full right and discretion to determine the time, manner and terms of sale or sales of Company property pursuant to such liquidation having due regard to the activity and condition of the relevant market and general financial and economic conditions. The Liquidator appointed in the manner provided herein shall have and may exercise, without further authorization or consent of any of the parties hereto or their legal representatives or successors in interest, all of the powers conferred upon the Members under the terms of this Agreement (but subject to all of the applicable limitations, contractual and otherwise, upon the exercise of such powers) to the extent necessary or desirable in the good faith judgment of the Liquidator to carry out the duties and functions of the Liquidator hereunder for and during such period of time, not to exceed two (2) years after the date of dissolution of the Company, as shall be reasonably required in the good faith judgment of the Liquidator to complete the liquidation and dissolution of the Company as provided for herein, including, without limitation, the following specific powers:

> (a) The power to continue to manage and operate any business of the Company during the period of such liquidation or dissolution proceedings, excluding, however, the power to make and enter into contracts which may extend beyond the period of liquidation.

> > -22-

(b) The power to make sales and incident thereto to execute deeds, bills of sale, assignments and transfers of assets and properties of the Company; provided, that the Liquidator may not impose personal liability upon any of the Members under any such instrument.

(c) The power to borrow funds as may, in the good faith judgment of the Liquidator, be reasonably required to pay debts and obligations of the Company or operating expenses, and to execute and/or grant deeds of trust, mortgages, security agreements, pledges and collateral assignments upon and encumbering any of the Company properties as security for repayment of such loans or as security for payment of any other indebtedness of the Company; provided, that the Liquidator shall not have the power to create any personal obligation on any of the Members to repay such loans or indebtedness other than out of available proceeds of foreclosure or sale of the properties or assets as to which a lien or liens are granted as security for payment thereof.

(d) The power to settle, release, compromise or adjust any claims asserted to be owing by or to the Company, and the right to file, prosecute or defend lawsuits and legal proceedings in connection with any such matters.

SECTION 18.2 RESERVES. In making payment or provision for payment of all debts and liabilities of the Company and all expenses of liquidation, the Liquidator may set up such cash reserves as the Liquidator may deem reasonably necessary for any contingent liabilities or obligations of the Company. Upon the satisfaction or other discharge of such contingency, the amount of the reserves not retired, if any, will be distributed in accordance with this Article.

SECTION 18.3 SALE OF ASSETS; DISTRIBUTION OF PROCEEDS. Upon the winding up and termination of the business and affairs of the Company, its assets (other than cash) shall be sold, its liabilities and obligations to creditors and all expenses incurred in its liquidation shall be paid (either by payment or the making of reasonable provision for payment). All items of Company income, gain, loss or deduction shall be credited or charged to the Capital Accounts of the Members pro rata in accordance with their Sharing Ratios. Thereafter, the net proceeds from such sales (after deducting all selling costs and expenses in connection therewith), together with the reserve account referred to in Section 18.2 above, shall be distributed among the Members in accordance with their respective positive balances in their Capital Accounts.

SECTION 18.4 FINAL ACCOUNTING. Within a reasonable time following the completion of the liquidation of the Company's properties, the Liquidator shall supply to each of the Members a statement prepared by the Company's accountants which shall set forth the assets and the liabilities of the Company as of the date of complete liquidation, each Member's pro rata portion of distributions pursuant to Section 18.3, and the amount retained as reserves by the Liquidator pursuant to Section 18.2.

SECTION 18.5 RECOURSE LIMITED TO COMPANY ASSETS. Each holder of an interest in the Company shall look solely to the assets of the Company for all distributions with respect to the Company and his Capital Contribution thereto (including the return thereof) and share of profits or

-23-

losses thereof, and shall have no recourse therefor (upon dissolution or otherwise) against the Company, the Members or the Liquidator. No holder of an interest in the Company shall have any right to demand or receive property other than cash upon dissolution and termination of the Company.

SECTION 18.6 TERMINATION. Upon the completion of the liquidation of the Company and the distribution of all Company funds, the existence of the Company shall terminate and the Liquidator shall (and is hereby given the power and authority to) execute, acknowledge, swear to and record all documents required to effectuate the dissolution and termination of the Company. No Member shall be required to restore any deficit balance existing in its Capital Account upon the liquidation and termination of the Company.

ARTICLE XIX

NOTICES

To be effective, all notices and demands under this Agreement must be in writing and must be given (i) by depositing same in the United States mail, postage prepaid, certified or registered, return receipt requested, (ii) by telecopier with receipt confirmed by return telecopy, or (iii) by delivering same in person and receiving a signed receipt therefor. For purposes of notice, the address of the Company shall be the address of its then principal office and the addresses of the Members or their respective assigns shall be as set forth on the signature pages hereof. Notices made in accordance with the foregoing shall be deemed to have been given and made upon receipt. Any Member or his assignee may designate a different address or telecopier number to which notices or demands shall thereafter be directed by written notice given in the manner hereinabove required and directed to the Company.

ARTICLE XX

AMENDMENTS AND MEETINGS

SECTION 20.1 AMENDMENTS. This Agreement may be modified or amended from time to time by the written agreement of the Members.

SECTION 20.2 MEETINGS OF THE MEMBERS.

(a) Meetings of the Members may be called at any time by a Member. Notice of any meeting shall be given to all Members not less than two (2) Business Days nor more than thirty (30) Business Days prior to the date of such meeting. Each Member may authorize any person to act for it by proxy on all matters in which a Member is entitled to participate, including waiving notice of any meeting, or voting or participating at meeting. Every proxy must be signed by the Member or its attorney-in-fact.

(b) Each meeting of Members shall be conducted by the person that the Member calling the meeting shall designate. The Members shall establish all other provisions relating to meetings of Members, including notice of the time, place or purpose of any meeting at

-24-

which any matter is to be voted on by any Members, waiver of any such notice, action by consent without a meeting, the establishment of a record date, quorum requirements, voting in person or by proxy or any other matter with respect to the exercise of any such right to vote.

ARTICLE XXI

REPRESENTATIONS, WARRANTIES AND COVENANTS

The Members each hereby respectively represent and warrant to the others that (i) it is duly organized, validly existing and in good standing under the jurisdiction of its organization, with full power and authority to enter into and perform its obligations under this Agreement; (ii) it has validly executed this Agreement, and upon delivery, this Agreement shall be a binding obligation of such party, enforceable against such party in accordance with its terms; and (iii) its entry into this Agreement and the performance of its obligations hereunder will not require the approval of any governmental body or regulatory authority and will not violate, conflict with or cause a default under, any of its organizational documents, any contractual covenant or restriction by which such party is bound, or any applicable law, regulation, rule, ordinance, order, judgment or decree.

ARTICLE XXII

DISPUTES

Except for the injunctive remedies provided by Section 15.8, in the event of any claims, counterclaims, demands, causes of action, disputes, controversies, and other matters in question arising out of or relating to this Agreement, any provision hereof, the alleged breach thereof, or in any way relating to the subject matter of this Agreement or the relationship between the Members created by this Agreement or the Partnership Agreement, involving the Members and/or their respective representatives and/or Affiliates (all of which are referred to herein as "DISPUTES"), the Members shall promptly seek to resolve any such Dispute in accordance with the terms and provisions of Article XXIII of the Partnership Agreement.

ARTICLE XXIII

MISCELLANEOUS

SECTION 23.1 NO RIGHT OF PARTITION. The Members agree that the Company properties are not and will not be suitable for partition. Accordingly, each of the Members hereby irrevocably waives any and all rights that he may have to maintain any action for partition of any of the Company property.

SECTION 23.2 ENTIRE AGREEMENT; SUPERSEDURE. This Agreement and the additional documents and agreements referred to herein constitute the entire agreement among the parties. It supersedes any prior agreement or understandings among them with regard to the subject matter hereof, and it may not be modified or amended in any manner other than as set forth herein.

-25-

SECTION 23.3 GOVERNING LAW. This Agreement and the rights of the parties hereunder shall be governed by and interpreted in accordance with the laws of the State of Delaware.

SECTION 23.4 BINDING EFFECT. Except as herein otherwise specifically provided, this Agreement shall be binding upon and inure to the benefit of the parties and their legal representatives, heirs, administrators, executors, successors and assigns.

SECTION 23.5 CONSTRUCTION OF AGREEMENT. Wherever from the context it appears appropriate, each term stated in either the singular or the plural shall include the singular and the plural, and pronouns stated in the masculine, the feminine or the neuter gender shall include the masculine, feminine and neuter. The term "person" means any individual, corporation, partnership, trust or other entity.

SECTION 23.6 CAPTIONS. Captions contained in this Agreement are inserted only as a matter of convenience and in no way define, limit or extend the scope or intent of this Agreement or any provision hereof.

SECTION 23.7 EFFECT OF INVALID PROVISION. If any provision of this Agreement, or the application of such provision to any person or circumstance, shall be held invalid, the remainder of this Agreement, or the application of such provision to persons or circumstances other than those to which it is held invalid, shall not be affected thereby.

SECTION 23.8 COUNTERPARTS. This Agreement may be executed in several counterparts, each of which shall be deemed an original but all of which shall constitute one and the same instrument. It shall not be necessary for all Members to execute the same counterpart hereof.

SECTION 23.9 WAIVER. No failure by any party to insist upon the strict performance of any covenant, duty, agreement or condition of this Agreement or to exercise any right or remedy consequent upon a breach thereof shall constitute waiver of any such breach or any other covenant, duty, agreement or condition.

IN WITNESS WHEREOF, the parties hereto have executed this Agreement as of the day and in the year first above written.

Address:	TE Products Pipeline Company, Limited Partnership, by TEPPCO GP, Inc.,
2929 Allen Parkway, Suite 3200 Houston, Texas 77019	its general partner
Telecopy: 713/759-3645	By:
	Its:

-26-

Louis Dreyfus Energy Services L.P.

10 Westport Road	D. (
Wilton, Connecticut 06897	By:
Telecopy: 203/761-8119	Its:

Its:																											
	-	_	-	-	_	-	-	_	-	-	-	-	-	_	-	-	-		-			-	_	-			

-27-

EXHIBIT G OF JOINT DEVELOPMENT AGREEMENT

AGREEMENT OF LIMITED PARTNERSHIP

0F

MONT BELVIEU STORAGE PARTNERS, L.P.

DATED AS OF JANUARY 21, 2003

DEFINITI	ONS		1
ARTICLE	I	FORMATION OF LIMITED PARTNERSHIP	7
ARTICLE	II	NAME	7
ARTICLE	III	PURPOSE	8
ARTICLE	IV	NAMES AND ADDRESSES OF PARTNERS AND PRINCIPAL OFFICE OF PARTNERSHIP	8
ARTICLE	V	REGISTERED AGENT; REGISTERED OFFICE; ADDITIONAL OFFICES	8
ARTICLE	VI	TERM	8
ARTICLE	VII	ADDITIONAL LIMITED PARTNERS	9
ARTICLE	VIII Section 8.1 Section 8.2 Section 8.3 Section 8.4	CAPITAL CONTRIBUTIONS. Capital Contributions. Additional Contributions. Capital Accounts. Adjustment of Capital Accounts.	
ARTICLE	IX Section 9.1 Section 9.2 Section 9.3 Section 9.4	DISTRIBUTIONS Frequency Failure to Withdraw Transferor/Transferee Allocations Partner Advances	13 13 13 13 13
ARTICLE	X Section 10.1 Section 10.2 Section 10.3 Section 10.4 Section 10.5	ALLOCATIONS OF INCOME, GAIN, LOSS, DEDUCTION AND CREDIT General Allocations for Tax Purposes Transferor/Transferee Allocations Reliance on Advisors Tax Matters Partner	14 14 15 15 16 16
ARTICLE	XI Section 11.1 Section 11.2	BOOKS OF ACCOUNT, RECORDS AND TAX INFORMATION Maintenance of Books and Records Reports	16 16 17

-i-

ARTICLE	XII	FISCAL YEAR	17
ARTICLE	XIII	PARTNERSHIP FUNDS	18
ARTICLE	XIV Section 14.1 Section 14.2	STATUS OF LIMITED PARTNERS No Personal Liability Dissolution or Bankruptcy of Limited Partner	18 18 18
ARTICLE	XV Section 15.1 Section 15.2 Section 15.3 Section 15.4 Section 15.5 Section 15.6 Section 15.7 Section 15.8	MANAGEMENT AND OPERATION OF BUSINESS.Management by General Partner.No Limited Partner Participation.Actions Requiring Limited Partner Consent.Third Party Reliance.Compensation and Reimbursement.Exculpation.Indemnification.Other Activities; Noncompetition.	18 19 19 20 20 20 21
ARTICLE	XVI Section 16.1 Section 16.2 Section 16.3 Section 16.4 Section 16.5 Section 16.6 Section 16.7	TRANSFER OF INTERESTS BY PARTNERS.Restrictions on Transfer.Transfers to Affiliates.Right of First Refusal.Change of Partner Control.Procedures.Purchase and Sale in the Event of Maximum Loss Being Exceeded.Compliance with Securities Laws.	24 24 24 25 25 26 26
ARTICLE	XVII	DISSOLUTION OF THE PARTNERSHIP	27
ARTICLE	XVIII Section 18.1 Section 18.2 Section 18.3 Section 18.4 Section 18.5 Section 18.6	WINDING UP AND TERMINATION OF THE PARTNERSHIP. Liquidator. Reserves. Sale of Assets; Distribution of Proceeds. Final Accounting. Recourse Limited to Assets of the Partnership. Termination.	27 27 28 29 29 29
ARTICLE	XIX	NOTICES	29
ARTICLE	ХХ	AMENDMENT OF LIMITED PARTNERSHIP AGREEMENT	30
ARTICLE	XXI	POWER OF ATTORNEY	30

-ii-

ARTICLE	XXII	REPRESENTATIONS, WARRANTIES AND	
		COVENANTS	30
ARTICLE	XXIII	DISPUTES	31
	Section 23.1	Negotiation	31
	Section 23.2	Failure to Resolve	31
	Section 23.3	Arbitration	32
	Section 23.4	Recovery of Costs and Attorneys' Fees	33
	Section 23.5	Choice of Forum	33
	Section 23.6	Jury Waivers	33
	Section 23.7	Special Buy-Out Options	34
	Section 23.8	Dispute Under Both this Agreement and LLC Agreement	35
ARTICLE	XXIV	MISCELLANEOUS	35
	Section 24.1	No Right of Partition	35
	Section 24.2	Entire Agreement; Supersedure	35
	Section 24.3	Governing Law	35
	Section 24.4	Binding Effect	35
	Section 24.5	Construction of Agreement	35
	Section 24.6	Captions	35
	Section 24.7	Effect of Invalid Provision	36
	Section 24.8	Counterparts	36
	Section 24.9	Waiver	36
EXHIBIT	A	Dreyfus Storage Agreement	
EXHIBIT	В	MB Assets	

- EXHIBIT C Operating Procedures
- EXHIBIT D TE Storage Agreement

-iii-

AGREEMENT OF LIMITED PARTNERSHIP OF MONT BELVIEU STORAGE PARTNERS, L.P.

THIS AGREEMENT is made as of the 21st day of January, 2003, by and among Mont Belvieu Venture, LLC, a Delaware limited liability company, as the General Partner, and TE Products Pipeline Company, Limited Partnership, a Delaware limited partnership, and Louis Dreyfus Energy Services L.P., a Delaware limited partnership, as the Limited Partners.

WHEREAS, the parties hereto desire to form a limited partnership under and pursuant to Delaware law.

NOW, THEREFORE, in consideration of the premises and the mutual undertakings contained herein, the parties hereto hereby agree as follows:

DEFINITIONS

The following definitions shall be applicable to the terms set forth below as used in this Agreement:

"ACCEPTANCE" has the meaning assigned to such term in Section 16.3.

"ACCOUNTANTS" has the meaning given such term in Section 11.2(a)(i).

"ACQUISITION PROPOSAL" has the meaning assigned to such term in Section 16.3.

"AFFILIATE" means, when used with respect to a specified person, any other person directly or indirectly controlling or controlled by or under direct or indirect common control with the specified person, provided that the Partnership shall not be deemed to be an Affiliate of the Partners or any of their respective subsidiaries or Affiliates. For purposes of this definition, "control", when used with respect to any specified person, means the power to direct the management and policies of the person, directly or indirectly, whether through the ownership of voting securities, by contract, by family relationship or otherwise; and the terms "controlling" and "controlled" have the meanings correlative to the foregoing. Notwithstanding the foregoing, for purposes of this Agreement, Louis Dreyfus Natural Gas Corporation, an Oklahoma corporation, shall not be deemed an Affiliate of the LD Partner.

"AGREED VALUE" means, in the case of any contributions or distributions of property other than the contributions described in Section 8.1(a), the fair market value of such property net of any indebtedness or other liability either assumed or to which such property is subject, as such fair market value is determined by the General Partner using such reasonable method of valuation as it may adopt.

-1-

"AGREEMENT" means the Agreement of Limited Partnership of Mont Belvieu Storage Partners, L.P. as the same may be amended, modified or restated from time to time in accordance with Article XX hereof.

"BUILT-IN GAIN" with respect to any Partnership property means (i) the excess of the Agreed Value of any Contributed Property over its adjusted basis for federal income tax purposes as of the time of contribution and (ii) in the case of any adjustment to the Carrying Value of any Partnership property subject to depreciation, cost recovery or amortization pursuant to Section 8.4 as a result of a contribution of cash for a Partnership Interest, the Unrealized Gain with respect to such property.

"BUILT-IN LOSS" with respect to any Partnership property means (i) the excess of its adjusted basis for federal income tax purposes of any Contributed Property over its Agreed Value as of the time of contribution and (ii) in the case of any adjustment to the Carrying Value of any Partnership property subject to depreciation, cost recovery or amortization pursuant to Section 8.4 as a result of a contribution of cash for a Partnership Interest, the Unrealized Loss with respect to such property.

"BUSINESS DAY" means any day other than a Saturday, Sunday or bank holiday in Houston, Texas.

"CAPITAL ACCOUNT" means the account established for each Partner pursuant to Section 8.3.

"CAPITAL CONTRIBUTIONS" means the Agreed Value of any property and the amount of cash contributed to the Partnership.

"CARRYING VALUE" with respect to any Capital Contribution means the Agreed Value of such property reduced as of the time of determination by all depreciation, cost recovery and amortization deductions charged to the Capital Accounts with respect to such property and an appropriate amount to reflect any sales, retirements or other dispositions of assets included in such property and, with respect to any other Partnership property, the adjusted basis of such property for federal income tax purposes as of the time of determination. The Carrying Values shall be further adjusted as provided in Section 8.4.

"CERTIFICATE OF LIMITED PARTNERSHIP" means the Certificate of Limited Partnership required to be filed in the Office of the Secretary of State of the State of Delaware pursuant to the Partnership Act, as the same may be amended, modified or restated from time to time.

"CHANGE OF PARTNER CONTROL" means either a TE Change of Partner Control or a LD Change of Partner Control.

"CODE" means the Internal Revenue Code of 1986, as amended and in effect on the effective date hereof and, to the extent applicable, as subsequently amended.

"COMPETING BUSINESS TRANSACTION" has the meaning assigned to such term in Section 15.8(c).

-2-

"CONTRIBUTED PROPERTY" means any Capital Contribution of property other than cash.

"DEFAULT INTEREST RATE" means a rate per annum equal to the lesser of (a) 3% per annum plus the Prime Rate or (b) the maximum rate permitted by law.

"DELINQUENT PARTNER" has the meaning assigned to such term in Section 8.1(d).

"DISCRETIONARY CAPITAL EXPENDITURES" means any expenditure related to the Partnership Business that is classified under GAAP as a capital expenditure and (i) is not a Mandatory Capital Expenditure, or (ii) if classified as a Mandatory Capital Expenditure, is associated or is in conjunction with, or is the result of, a prior or current Discretionary Capital Expenditure, in which case the portion of such Mandatory Capital Expenditure that is allocable to such prior or current Discretionary Capital Expenditure shall be classified as a Discretionary Capital Expenditure.

"DISPOSE, DISPOSING OR DISPOSITION" means, with respect to Partnership Interest or any portion thereof, a sale, assignment, transfer, conveyance, gift, exchange or other disposition of such asset, whether such disposition be voluntary, involuntary or by operation of law, including but not limited to the following: (a) in the case of a Partnership Interest owned by an entity, a Change of Partner Control and (b) a disposition in connection with, or in lieu of, a foreclosure of an Encumbrance; but such terms shall not include the creation of an Encumbrance.

"DISPUTES" has the meaning assigned to such term in Section 23.1.

"DISTRIBUTABLE CASH" means, at the time of determination, all Partnership cash derived from the conduct of the Partnership's business, other than (i) Capital Contributions, together with interest earned thereon pending utilization thereof, (ii) financing proceeds, (iii) amounts required for working capital and Discretionary and Mandatory Capital Expenditures and (iv) other amounts that the General Partner reasonably determines to be necessary for the proper operation of the Partnership's business and its winding up and liquidation.

"DREYFUS STORAGE AGREEMENT" shall mean the Storage Agreement in the form attached as Exhibit A hereto.

"EBITDA" means for any entity, such entity's earnings for any period on a GAAP basis before all charges for such period, for depreciation and amortization, interest (except as provided below) and income taxes. Notwithstanding the foregoing, all interest incurred in connection with borrowings by such entity the proceeds of which are used for Discretionary Capital Expenditures relating to the MB Assets shall be deducted as an expense in calculating EBITDA.

"ENCUMBER, ENCUMBERING, OR ENCUMBRANCE" means the creation of a security interest, lien, pledge, mortgage or other encumbrance, whether such encumbrance be voluntary, involuntary or by operation of law.

"EXERCISE PERIOD" has the meaning assigned to such term in Section 23.7(a).

-3-

"FIRST PARTY" has the meaning assigned to such term in Section 15.8(c).

"GAAP" means Generally Accepted Accounting Principles in the United States of America consistently applied for the time periods involved.

"GENERAL PARTNER" means Mont Belvieu Venture, LLC, a Delaware limited liability company or its successors or assigns.

"INDEMNITEE" has the meaning assigned to such term in Section 15.7.

"INTEREST" has the meaning assigned to such term in Section 16.3.

"JOINT DEVELOPMENT AGREEMENT" means that certain Joint Development Agreement between the LD Partner and the TE Partner dated effective January 1, 2000, as the same may be modified, amended or restated from time to time.

"LD COMBINED CAPITAL ACCOUNT BALANCE" has the meaning assigned to such term in Section 16.4(b).

"LD PARTNER CHANGE OF CONTROL" means an event that causes S. A. Louis Dreyfus et Cie., a French societe anonyme, together with its Affiliates to cease to own, directly or indirectly, at least a 50% voting interest in (i) the LD Limited Partner or any Affiliate of the LD Limited Partner owning a membership interest in the General Partner or (ii) in the event of the transfer of the LD Partner's Partnership Interest to an Affiliate, in such Affiliate.

"LD PARTNER" means Louis Dreyfus Energy Services L.P., a Delaware limited partnership, or its successors or assigns.

"LENDING PARTNER" has the meaning assigned to such term in Section 8.1(d).

"LIMITED PARTNERS" means TE Products Pipeline Company, Limited Partnership, a Delaware limited partnership and Louis Dreyfus Energy Services L.P., a Delaware limited partnership, or their respective successors or permitted assigns, and "LIMITED PARTNER" means either of them.

"LIQUIDATOR" has the meaning assigned to such term in Section 18.1.

"LLC AGREEMENT" means the Limited Liability Company Agreement of the General Partner as the same may be modified, amended or restated from time to time.

"LOSS" shall mean on any date the amount, if any, by which the sum of the Minimum Cash Balance plus any other net requirements for cash to fund the operations of the Partnership reasonably expected to arise over the following month exceeds the Partnership's cash balance on that date exclusive of segregated cash balances.

-4-

"MANDATORY CAPITAL EXPENDITURES" means all expenditures related to the Partnership's assets that are classified under GAAP as a capital expenditure, the primary function of which is for regulatory compliance, safety or operational integrity of the Partnership's assets or is otherwise the result of actions permitted in emergency situations as provided in the Operating Procedures.

"MAXIMUM LOSS" shall mean \$1 million.

"MB ASSETS" means the Mont Belvieu salt cavern storage and Houston area shuttle system more fully described on Exhibit B hereto and all improvements, expansions and additions thereto.

"MINIMUM CASH BALANCE" shall mean \$100,000 or such larger number as the General Partner determines is reasonable and necessary to carry out the business of the Partnership.

"NET EBITDA" means the amount of EBITDA, if any, in excess of \$7.3 million during any period.

"NOTICE MONTH" has the meaning assigned to such term in Section 16.4(b).

"OFFER" has the meaning assigned to such term in Section 16.3.

"OFFER PERIOD" has the meaning assigned to such term in Section 16.3.

"OFFEREE" has the meaning assigned to such term in Section 16.3.

"OFFEROR" has the meaning assigned to such term in Section 16.3.

"OPERATING PROCEDURES" means the Operating Procedures in the form attached as Exhibit C hereto.

16.4.

"OPTION INTEREST" has the meaning assigned to such term in Section

"OPTION PARTNER" has the meaning assigned to such term in Section 16.4.

"PARENT" means, with respect to TE Partner, TEPPCO Partners, L.P., a Delaware limited partnership, and with respect to LD Partner, S. A. Louis Dreyfus et Cie., a French societe anonyme.

"PARTNERS" means the General Partner and the Limited Partners.

"PARTNERSHIP" means Mont Belvieu Storage Partners, L.P., the limited partnership entered into and formed hereunder pursuant to the Partnership Act.

"PARTNERSHIP ACT" means the Delaware Revised Uniform Limited Partnership Act, 6 Del. Code Sections 17-101 et seq., as it may be amended from time to time, and any successor to said Partnership Act.

-5-

"PARTNERSHIP BUSINESS" means the operation, maintenance and marketing of underground storage facilities for Products and the operation of the pipelines included within the MB Assets to and from their specific points of delivery and receipt.

"PARTNERSHIP INTEREST" as to any Partner means the entire ownership interest and rights of that Partner in the Partnership, including, without limitation, its right to a distributive share of the profits and losses of the Partnership, its right to a distributive share of the assets of the Partnership in accordance with the provisions hereof, and in the case of a General Partner, its right to participate in the management of the affairs of the Partnership.

"PRICING PERIOD" means the trailing 12-month period ending on a date or month as specified in this Agreement.

"PRIME RATE" means the interest rate as defined in Section 9.4 hereof.

"PRODUCTS" means propane, normal butane, isobutane, natural gasoline and any other commodities as determined by the General Partner.

"PUT/CALL RIGHT" has the meaning assigned to such term in Section 23.7(a).

"RESTRICTED AFFILIATE" means with respect to each Limited Partner, any person in which such Limited Partner's Parent owns, directly or indirectly, at least a 51% voting or equity interest, or which it otherwise controls. For purposes of this definition, "control," has the meaning assigned to such term in the definition of Affiliate and Louis Dreyfus Natural Gas Corporation shall not be deemed a Restricted Affiliate of the LD Partner.

"SECOND PARTY" has the meaning assigned to such term in Section 15.8(c).

"SHARING RATIO" means 1% in the case of the General Partner and 49.5% in the case of each Limited Partner.

"STORAGE AGREEMENTS" means the Dreyfus Storage Agreement and the TE Storage Agreement.

"TE PARTNER CHANGE OF CONTROL" means a Disposition of any entity of which over 75% of such entity's EBITDA for the preceding 12-month period is derived from the Partnership unless such disposition is to an Affiliate of the TE Partner.

"TE PARTNER" means TE Products Pipeline Company, Limited Partnership or its successors or assigns.

"TE STORAGE AGREEMENT" means the Storage Agreement in the form attached hereto as Exhibit D.

"TERRITORY" has the meaning assigned to such term in Section 15.8(a).

-6-

"TMP" has the meaning assigned to such term in Section 10.5.

"UNREALIZED GAIN" attributable to a Partnership property means, as of the date of determination, the excess of the fair market value of such property as of such date of determination over the Carrying Value of such property as of such date of determination.

"UNREALIZED LOSS" attributable to a Partnership property means, as of the date of such determination, the excess of the Carrying Value of such property as of such date of determination over the fair market value of such property as of such date of determination.

"UNRECOVERED LOSS BALANCE" shall mean for either Limited Partner the result of (a) the sum of all Capital Contributions under Sections 8.1(b)(iii) and 8.1(c), minus (b) the sum of all distributions under Section 9.1(a).

"VALUATION EXPERT" has the meaning assigned to that term in Section 23.7(b).

"VALUATION MULTIPLE" means the result of dividing the fair market value of businesses comparable or similar to the business of the Partnership by the same businesses' EBITDA for the twelve-month period immediately preceding the date such fair market value was measured.

ARTICLE I

FORMATION OF LIMITED PARTNERSHIP

The parties hereto hereby form a limited partnership pursuant to the Partnership Act. The rights and liabilities of the Partners shall be as provided in the Partnership Act, except as herein otherwise expressly provided. The Partnership Interests of any Partner shall be personal property for all purposes. On the request of the General Partner, the Limited Partners shall execute, acknowledge, swear to, and deliver all certificates and other instruments conforming with this Agreement that are necessary to qualify, continue, or terminate the Partnership as a limited partnership under the laws of the State of Delaware and to qualify the Partnership to do business in such other states where such qualification is necessary or desirable.

ARTICLE II

NAME

The name of the Partnership shall be, and the business of the Partnership shall be conducted under the name of, Mont Belvieu Storage Partners, L.P. or such other name or names that comply with applicable law as the General Partner may designate from time to time. The General Partner shall take any action that it determines is required to comply with the Partnership Act, assumed name act, fictitious name act, or similar statute in effect in each jurisdiction or political subdivision in which the Partnership proposes to do business and the Limited Partners agree to execute any documents requested by the General Partner in connection with any such action.

-7-

ARTICLE III

PURPOSE

The purpose of the Partnership is to acquire, own, operate, maintain, improve and otherwise deal with the MB Assets, and by means thereof, provide storage, truck and railcar loading and unloading and pipeline transportation services to the Partners, their Affiliates and third parties, and any other business activities necessary, incidental or ancillary to any of the foregoing.

ARTICLE IV

NAMES AND ADDRESSES OF PARTNERS AND PRINCIPAL OFFICE OF PARTNERSHIP

The names and mailing addresses of the Partners are as set forth on the signature pages hereof. The location of the principal office of the Partnership where the books and records of the Partnership shall be kept shall be 2929 Allen Parkway, Suite 3200, Houston, Texas 77019. The General Partner may change the location of the principal office of the Partnership.

ARTICLE V

REGISTERED AGENT; REGISTERED OFFICE; ADDITIONAL OFFICES

The name and address of the registered office of the Partnership in the State of Delaware is c/o The Corporation Trust Company, Corporation Trust Center, 1209 Orange Street, Wilmington, New Castle County, Delaware 19801. The name and address of the registered agent for service of process on the Partnership in the State of Delaware is The Corporation Trust Company, Corporation Trust Center, 1209 Orange Street, Wilmington, New Castle County, Delaware 19801. The General Partner may change the registered agent or the registered office of the Partnership and may establish such additional offices of the Partnership as the General Partner may from time to time determine. The General Partner shall provide the Limited Partners with written notice of any change in the Partnership's principal office, registered agent or registered office.

ARTICLE VI

TERM

The term of the Partnership shall be perpetual from the date of the filing of the Certificate of Limited Partnership in the Office of the Secretary of State of the State of Delaware unless sooner liquidated or dissolved in accordance with this Agreement.

-8-

ARTICLE VII

ADDITIONAL LIMITED PARTNERS

In the event the General Partner determines that funds in addition to those acquired pursuant to Section 8.1 are necessary to carry out the purposes of the Partnership and the General Partner and the Limited Partners do not agree to contribute such funds in their respective Sharing Ratios, the General Partner is authorized to offer and sell additional Partnership Interests and admit any purchasers thereof (which may include existing Partners or their Affiliates) as additional limited partners of the Partnership. In such event, the dilution of the then Partners shall be pro rata in accordance with their respective Sharing Ratios prior to the issuance of such additional Partnership Interests and this Partnership Agreement shall be amended to reflect the admissions of such persons as limited partners of the Partnership, the revised ownership and Sharing Ratio of each Partner and otherwise to reflect the relative rights and obligations of the parties.

ARTICLE VIII

CAPITAL CONTRIBUTIONS

SECTION 8.1 CAPITAL CONTRIBUTIONS.

(a) Each Limited Partner shall make the initial Capital Contributions provided in the Joint Development Agreement and the General Partner shall contribute \$500,000 in cash to the Partnership. The Agreed Value of any property included in such Capital Contributions shall be its net book value reduced by liabilities assumed by the Partnership as reflected on the books of the contributor at the date of contribution computed in accordance with GAAP.

(b) Without creating any rights in favor of any third party, (i) each Partner shall contribute to the Partnership, in cash, on or before the date specified as described in this Section 8.1(b), that Partner's Sharing Ratio of Discretionary Capital Expenditures shown in the current capital expenditure budget adopted for the Partnership by the General Partner as being funded by Capital Contributions, (ii) the TE Partner shall contribute to the Partnership each quarter an amount equal to the actual Mandatory Capital Expenditure for such quarter (excluding Mandatory Capital Expenditures attributable to prior Discretionary Capital Expenditures), in cash, on the earlier of (x) the date needed by the Partnership to pay such expenses or (y) the date of the quarterly distribution under Section 9.1(b) for such quarter, and (iii) the TE Partner shall contribute to the Partnership, in cash, within three days of such balance being known, an amount equal to any Loss of the Partnership, except that in the event the TE Partner's Unrecovered Loss Balance is equal to or greater than the Maximum Loss, then Section 8.1(c) shall apply. Unless otherwise stated herein, the General Partner shall notify each Partner of the need for such Capital Contributions pursuant to this Section 8.1(b), which notice must include a date (which date may be no earlier than the second Business Day following each Partner's receipt of its notice) before which the Capital Contributions must be made. All Capital Contributions under Section 8.1(b)(i) shall be kept

-9-

in a segregated account, shall not be commingled with other funds of the Partnership, shall be used only for Discretionary Capital Expenditures and shall not be included in Distributable Cash.

(C)

(i) If the TE Partner's Unrecovered Loss Balance is, or with the then current Loss added, will be, equal to or greater than the Maximum Loss then the TE Partner shall have the option to (x) contribute, in cash, to the Partnership an amount equal to the current Loss, or (y) cease to contribute any cash, in which case the LD Partner shall have the option to contribute, in cash, to the Partnership an amount equal to the current Loss.

(ii) If, at any time the TE Partner's Unrecovered Loss Balance is equal to or greater than the Maximum Loss and neither Partner chooses to contribute, in cash, to the Partnership an amount equal to the current Loss then the provisions of Section 16.6 shall apply.

(d) If a Partner does not contribute by the time required all or any portion of a Capital Contribution that Partner is required to make as provided in this Agreement, the Partnership may exercise, on notice to that Partner (the "DELINQUENT PARTNER"), one or more of the following remedies:

> (i) taking such action (including court proceedings) as the other Partners may deem appropriate to obtain payment by the Delinquent Partner of the portion of the Delinquent Partner's Capital Contribution that is in default, together with interest on that amount at the Default Interest Rate from the date that the Capital Contribution was due until the date that it is made, all at the cost and expense of the Delinquent Partner;

(ii) permitting one or more other Partners (the "LENDING PARTNER"), to advance the portion of the Delinquent Partner's Capital Contribution that is in default, with the following results:

1. the sum advanced shall constitute a loan from the Lending Partner to the Delinquent Partner and a Capital Contribution of that sum to the Partnership by the Delinquent Partner under the applicable provisions of this Agreement,

2. the principal balance of the loan and all accrued unpaid interest is due and payable on the tenth day after written demand by the Lending Partner to the Delinquent Partner,

-10-

3. the amount loaned shall bear interest at the Default Interest Rate from the date that the advance is deemed made until the date that the loan, together with all interest accrued on it, is repaid to the Lending Partner,

4. all distributions from the Partnership that otherwise would be made to the Delinquent Partner (whether before or after dissolution of the Partnership) instead shall be paid to the Lending Partner until the loan and all interest accrued on it have been paid in full to the Lending Partner (with payments being applied first to accrued and unpaid interest and then to principal), but all such payments to the Lending Partner shall be treated for all purposes of this Agreement as a distribution by the Partnership to the Delinquent Partner and a payment by the Delinquent Partner to the Lending Partner,

5. the payment of the loan and interest accrued on it is secured by a security interest in the Delinquent Partner's Partnership Interest, as more fully set forth in Section 8.1(e), and

6. the Lending Partner has the right, in addition to the other rights and remedies granted to it under this Agreement or at law or in equity, to take any action (including court proceedings) that the Lending Partner may deem appropriate to obtain payment by the Delinquent Partner of the loan and all accrued and unpaid interest on it, at the cost and expense of the Delinquent Partner;

(iii) exercising the rights of a secured party under the Uniform Commercial Code of the State of Delaware, as more fully set forth in Section 8.1(e); or

(iv) exercising any other rights and remedies available at law or in equity.

(e) Each Partner grants to the Partnership, and to each Lending Partner with respect to any loans made by the Lending Partner to that Partner as a Delinquent Partner as described in Section 8.1(d), as security, equally and ratably, for the payment of all Capital Contributions that Partner has agreed to make and the payment of all loans and interest accrued on them made by Lending Partners to that Partner as a Delinquent Partner as described in Section 8.1(d), a security interest in and a general lien on its Partnership Interest and the proceeds of that Partnership Interest, all under the Uniform Commercial Code of the State of Delaware. On any default in the payment of a Capital Contribution or in the payment of such a loan or interest accrued on it, the Partnership or the Lending Partner, as applicable, is entitled to all the rights and remedies of a secured party under the Uniform Commercial Code of the State of Delaware with respect to the security interest granted in this Section 8.1(d). Each Partner shall execute and deliver to the Partnership and the other Partners all financing statements and other instruments that the other Partners or the Lending Partner, as applicable, may request to effectuate and carry out the preceding provisions of this

-11-

Section 8.1(e). At the option of a Lending Partner, this Agreement or a photographic or other copy of this Agreement may serve as a financing statement.

SECTION 8.2 ADDITIONAL CONTRIBUTIONS.

(a) The liability of each Limited Partner to the Partnership shall be limited to the amount of its Capital Contributions to be made pursuant to Section 8.1 and no Limited Partner shall have any further personal liability to contribute money to, or in respect of, the liabilities or the obligations of the Partnership unless it agrees in writing to make additional capital contributions to the Partnership, nor shall any Limited Partner be personally liable for any obligations of the Partnership, except as may be provided in the Partnership Act.

(b) Notwithstanding the foregoing provisions of Section 8.2(a), in the event the MB Assets contributed to the Partnership by the TE Partner are Encumbered as collateral for indebtedness that is not assumed by the Partnership, the TE Partner shall (i) continue to assume responsibility for (A) the payment of such indebtedness, and (B) upon payment of such indebtedness, the removal of all Encumbrances, and (ii) shall indemnify and hold harmless the Partnership for any cost, loss or expense incurred by the Partnership as a result of such Encumbrance.

(c) No Partner shall be entitled to the return of any part of its Capital Contribution or to be paid interest in respect of either its Capital Account or any Capital Contribution made by such Partner. No unrepaid Capital Contribution shall be deemed or considered to be a liability of the Partnership or any Partner. No Partner shall be required to contribute or lend any cash or property to the Partnership to enable the Partnership to return any Partner's Capital Contributions to the Partner.

SECTION 8.3 CAPITAL ACCOUNTS. A capital account ("CAPITAL ACCOUNT") shall be established for each Partner and shall be maintained in such a manner as to correspond with the requirements of Treasury Regulations promulgated from time to time under section 704(b) of the Code.

SECTION 8.4 ADJUSTMENT OF CAPITAL ACCOUNTS. If any additional Partnership Interests are to be issued in consideration for a contribution of property or cash or if any Partnership property is to be distributed in liquidation of the Partnership or a Partnership Interest, the Capital Accounts of the Partners (and the amounts at which all Partnership properties are carried on its books and records utilized to maintain the Capital Accounts) shall, immediately prior to such issuance or distribution, as the case may be, be adjusted (consistent with the provisions of section 704(b) of the Code and the Treasury Regulations promulgated thereunder) upward or downward to reflect any Unrealized Gain or Unrealized Loss attributable to all Partnership properties (as if such Unrealized Gain or Unrealized Loss had been recognized upon actual sale of such properties upon a liquidation of the Partnership immediately prior to such issuance). If the Agreed Value of any property of the Partnership is properly reflected on the books of the Partnership at a value that differs from the adjusted tax basis of such property, this Section 8.4 shall be applied with reference to such value.

-12-

ARTICLE IX

DISTRIBUTIONS

SECTION 9.1 FREQUENCY. Except as otherwise provided herein, Distributable Cash attributable to each fiscal year shall be distributed as follows:

(a) First, quarterly all amounts to the Limited Partner or Partners who funded Losses under Sections 8.1(b)(iii) or (c) pro rata in proportion to their respective Unrecovered Loss Balances until such time that such Partners' Unrecovered Loss Balances, if any, are reduced to zero.

(b) Second, quarterly to the TE Partner in an amount up to the result of (i) \$1.825 million, minus (ii) Mandatory Capital Expenditures for such quarter, minus (iii) any amounts payable in such quarter to a third party operator of the Partnership's assets if the General Partner, acting by and through its Managing Member, ceases to operate such assets, plus (iv) any Capital Contributions made by the TE Partner under Section 8.1(b)(ii) in such quarter.

(c) Third, for the full fiscal year, to the TE Partner in an amount equal to the positive difference, if any, of the result of (i) \$7.3 million, minus (ii) Mandatory Capital Expenditures for such fiscal year, minus (iii) any amounts payable in such fiscal year to a third party operator of the Partnership's assets if the General Partner, acting by and through its Managing Member, ceases to operate such assets, minus (iv) all distributions to the TE Partner pursuant to Section 9.1(b) for quarters included in such fiscal year plus (v) any Capital Contributions made by the TE Partner under Sections 8.1(b)(ii) in such fiscal year.

(d) Fourth, for the full fiscal year, the balance, if any, among the Partners pro rata in accordance with their Sharing Ratios.

SECTION 9.2 FAILURE TO WITHDRAW. If any Partner does not withdraw the whole or any part of his share of any cash distribution made pursuant to Section 9.1, such Partner shall not be entitled to receive any interest thereon without the express written consent of the other Partner.

SECTION 9.3 TRANSFEROR/TRANSFEREE ALLOCATIONS. Unless otherwise agreed in writing by a transferor and transferee of a Partnership Interest herein, Distributable Cash distributable with respect to any Partnership Interest which may have been transferred during any year shall be distributed to the holder of such Partnership Interest who was recognized as the owner on the date of such distribution, without regard to the results of Partnership operations during the year.

SECTION 9.4 PARTNER ADVANCES. Notwithstanding the foregoing, if any Partner advances any funds or makes any other payment to or on behalf of the Partnership, not required pursuant to the provisions hereof, to cover operating or capital expenses of the Partnership which cannot be paid out of the Partnership's operating revenues, such advance or payment shall be deemed a loan to the Partnership by such Partner, bearing interest from the date such advance or payment was made until

-13-

such loan is repaid at a floating rate per annum equal to the lesser of (i) two percent (2%) over the interest rate publicly quoted by The Wall Street Journal from time to time as the prime rate, with adjustments in such varying rate to be made on the same date as any change in the aforesaid rate (herein called "PRIME RATE") or (ii) the maximum rate permitted under applicable law. Notwithstanding Section 9.1 above, all distributions of Distributable Cash shall first be distributed to the Partners making such loans until all such loans have been repaid to such Partners, together with interest thereon as above provided, and, thereafter, the balance of such distributions, if any, shall be made in accordance with the terms of Section 9.1 above. If distributions are insufficient to repay and return all such loans as provided above, the funds available from time to time shall first be applied to repay and retire the oldest loan first and, if any funds thereafter remain available, such funds shall be applied in a similar manner to remaining loans in accordance with the same date, each such loan shall be repaid pro rata in the proportion that such loan bears to the total loans made on said date.

ARTICLE X

ALLOCATIONS OF INCOME, GAIN, LOSS, DEDUCTION AND CREDIT

SECTION 10.1 GENERAL.

(a) Except as otherwise provided herein or unless another allocation is required by Treasury Regulations issued under section 704(b) of the Code (including, but not limited to, provisions pertaining to qualified income offsets and minimum gain chargebacks), for purposes of maintaining the Capital Accounts, all items of Partnership income, gain, loss, deduction and credit shall be allocated among the Partners pro rata in accordance with their Sharing Ratios in effect for the period during which such items accrue. For purposes of computing the amount of each item of income, gain, deduction or loss to be charged or credited to the Capital Accounts, the determination, recognition and classification of such item shall be the same as its determination, recognition and classification for federal income tax purposes, provided that:

> (i) Any deductions for depreciation, cost recovery, or amortization attributable to any Partnership property shall be determined as if the adjusted basis of such property were equal to the Carrying Value of such property. Upon an adjustment to the Carrying Value of any Partnership property subject to depreciation, cost recovery, or amortization pursuant to Section 8.4, any further deductions for such depreciation, cost recovery, or amortization attributable to such property shall be determined as if the adjusted basis of such property were equal to the Carrying Value of such property immediately following such adjustment.

> (ii) Any income, gain or loss attributable to the taxable disposition of any Partnership property shall be determined by the Partnership as if the adjusted basis of such property as of such date of disposition were equal in amount to the Carrying Value of such property as of such date.

> > -14-

(iii) All fees and other expenses incurred by the Partnership to promote the sale of a Partnership Interest that can neither be deducted nor amortized under section 709 of the Code shall be treated as an item of deduction.

(iv) Computation of all items of income, gain, loss and deduction shall be made without regard to any election under section 754 of the Code which may be made by the Partnership and, as to those items described in the section 705(a)(1)(B) or section 705(a)(2)(B) of the Code, without regard to the fact that such items are not includable in gross income or are neither currently deductible nor capitalizable for federal income tax purposes.

(b) Notwithstanding the foregoing, (i) each Limited Partner shall be allocated Partnership gross income each fiscal year in an amount equal to the distribution made to it pursuant to Section 9.1(a), 9.1(b) and 9.1(c) with respect to such fiscal year and (ii) each Partner shall be allocated (x) the depreciation and/or amortization deductions attributable to the property, if any, contributed by such Partner to the Partnership pursuant to Section 8.1(a) of this Agreement and (y) an amount of deductions equal to the Capital Contributions made by such Partner to fund Losses pursuant to Sections 8.1(b)(iii) and 8.1(c).

SECTION 10.2 ALLOCATIONS FOR TAX PURPOSES.

(a) The Partnership shall, except to the extent such item is subject to allocation pursuant to subsection (b) below, allocate each item of income, gain, loss, deduction and credit, as determined for federal and other income tax purposes, in the same manner as such item was allocated for Capital Account purposes; and

(b) The Partnership, for federal and other income tax purposes shall, in the case of Contributed Properties, allocate items of income, gain, loss, depreciation and cost recovery deductions attributable to those properties with a Built-In Gain or Built-In Loss pursuant to section 704(c) of the Code and the Treasury Regulations thereunder. Similar allocations shall be made in the event that the Carrying Value of Partnership properties subject to depreciation, cost recovery or amortization are adjusted pursuant to Section 8.4 upon the issuance of Partnership Interests for cash. If an existing Partner acquires additional Partnership Interests, such allocations shall apply only to the extent of his or its additional Interests. No allocation under section 704(c) of the Code shall be charged or credited to a Partner's Capital Account.

SECTION 10.3 TRANSFEROR/TRANSFEREE ALLOCATIONS. Unless another method permitted by the Code is agreed to by the General Partner, the transferor and the transferee, income, gain, loss, deduction or credit attributable to any Partnership Interest which has been transferred shall be allocated between the transferor and the transferee allocated equally among the days of the Partnership's fiscal year without regard to Partnership operations during such days.

-15-

SECTION 10.4 RELIANCE ON ADVISORS. The General Partner may rely upon, and shall have no liability to the Limited Partner or the Partnership if it does rely upon, the written opinion of tax counsel or accountants retained by the Partnership from time to time with respect to all matters (including disputes with respect thereto) relating to computations and determinations required to be made under this Article X or other provisions of this Agreement.

SECTION 10.5 TAX MATTERS PARTNER.

(a) The General Partner is designated tax matters partner ("TMP") as defined in section 6231(a)(7) of the Code. The TMP and the Limited Partners shall use their best efforts to comply with responsibilities outlined in this Section 10.5 and in sections 6222 through 6232 of the Code (including any Treasury regulations promulgated thereunder) and in doing so shall incur no liability to any other Partner.

(b) If a Limited Partner intends to file a notice of inconsistent treatment under section 6222(b) of the Code, such Partner shall, prior to the filing of such notice, notify the TMP of such intent and the manner in which the Limited Partner's intended treatment of a Partnership item is (or may be) inconsistent with the treatment of that item by the Partnership.

(c) No Partner other than the TMP shall file a request pursuant to section 6227 of the Code for an administrative adjustment of partnership items for any Partnership taxable year.

(d) No Partner other than the TMP shall file a petition under Code sections 6226, 6228 or other Code sections with respect to any Partnership item, or other tax matters involving the Partnership. In the case where the TMP files such petition, it shall determine the forum in which such petition will be filed.

ARTICLE XI

BOOKS OF ACCOUNT, RECORDS AND TAX INFORMATION

SECTION 11.1 MAINTENANCE OF BOOKS AND RECORDS. Proper and complete records and books of account (including those required by the Partnership Act) shall be kept by the Partnership in which shall be entered all transactions and other matters relative to the Partnership's business as are usually entered into records and books of account maintained by persons engaged in businesses of like character. The Partnership books and records necessary for purposes of maintaining the Capital Accounts and giving effect to the allocations provided in Articles VIII and X shall be maintained in accordance with the accounting principles described therein, and shall be kept on the accrual basis and the other books and records shall be kept on the accrual basis in accordance with GAAP for financial reporting purposes and for purposes of making all other calculations contemplated herein. The books and records shall at all times be made available at the principal office of the Partnership and shall be open to the reasonable inspection and examination of the

-16-

Partners or their duly authorized representatives during the business hours of the General Partner for any purpose reasonably related to the interest of such Partner as a partner in the Partnership.

SECTION 11.2 REPORTS.

(a) The Partnership shall

(i) appoint a firm of certified public accountants to report on the Partnership's financial statements (the "ACCOUNTANTS"), and

(ii) direct the Accountants to render an audit opinion on the Partnership's financial statements within 90 days after the end of each fiscal year.

(b) As soon as reasonably practicable after the end of each fiscal year and after the Accountants have rendered an opinion on the Partnership's financial statements, the Partnership shall send each person who was a holder of a Partnership Interest at any time during the fiscal year then ended:

> (i) All Partnership tax information as shall be necessary for the preparation by such holder of its federal and any applicable state and local income tax returns; and

(ii) A copy of the Partnership's audited financial statements which shall include an income statement, a balance sheet, statement of cash flow and required notes to the financial statements.

(c) As soon as practicable after the end of each calendar quarter, the Partnership shall send to each Partner:

(i) An unaudited income statement and a statement of cash flows for such quarter; and

(ii) An unaudited balance sheet as of the end of such quarter and a profit and loss statement and statement of cash flows for the portion of the fiscal year then ended.

ARTICLE XII

FISCAL YEAR

The fiscal year of the Partnership shall end on the thirty-first (31st) day of December in each year.

-17-

ARTICLE XIII

PARTNERSHIP FUNDS

The funds of the Partnership shall be deposited in such bank account or accounts, or invested in such interest-bearing or non-interest-bearing accounts, as shall be designated by the General Partner. All withdrawals from any such bank accounts shall be made by the General Partner or other agent or agents duly authorized by the General Partner. Partnership funds shall not be commingled with those of any other person (including the General Partner or any of its Affiliates).

ARTICLE XIV

STATUS OF LIMITED PARTNERS

SECTION 14.1 NO PERSONAL LIABILITY. No Limited Partner shall have any personal liability whatever, whether to the Partnership, to any of the Partners or to the creditors of the Partnership, for the debts of the Partnership or any of its losses beyond the amount agreed to be contributed by him to the capital of the Partnership as set forth in Section 8.1 except to the extent required by the Partnership Act. No Limited Partner shall be obligated to restore any deficit in its Capital Account upon the liquidation of the Partnership or its Partnership Interest.

SECTION 14.2 DISSOLUTION OR BANKRUPTCY OF LIMITED PARTNER. The dissolution or bankruptcy of a Limited Partner shall not cause a dissolution of the Partnership, but the rights of such Limited Partner to share in the profits and losses of the Partnership, and to receive distributions of Partnership funds, shall on the happening of such an event, devolve upon such Limited Partner's legal representatives or successors in interest, as the case may be, subject to the terms and conditions of this Agreement, and the Partnership shall continue as a limited partnership. Each Limited Partner's successor in interest shall be liable for all the obligations of such Limited Partner. In no event, however, shall such estate, legal representative or other successor in interest become a substituted Limited Partner, except in accordance with Article XVI hereof.

ARTICLE XV

MANAGEMENT AND OPERATION OF BUSINESS

SECTION 15.1 MANAGEMENT BY GENERAL PARTNER.

(a) Except as otherwise expressly provided in Section 15.3 of this Agreement, all management powers over the business and affairs of the Partnership shall be exclusively vested in the General Partner, and the Limited Partners shall not have any right of control or management power over the business and affairs of the Partnership. The General Partner may, from time to time, designate one or more individuals as Partnership officers to carry out the day to day operations of the Partnership's activities. Such Partnership officers may include a president, vice-presidents, a secretary, a treasurer and such other officers as the General Partner shall designate.

-18-

(b) Whenever in this Agreement (including the Operating Procedures) the General Partner is permitted or required to make a decision (i) in its "discretion" or under a similar grant of authority or latitude, the General Partner shall be entitled to consider such interests and factors as it desires and may consider its own interests or (ii) in its "good faith" or under another express standard, the General Partner shall act under such express standard and shall not be subject to any other or different standards imposed by this Agreement or by law. The Limited Partners hereby agrees that any standard of care or duty imposed in this Agreement or under the Partnership Act or any other applicable law, rule or regulation shall be modified, waived or limited in each case as required to permit the General Partner to act under this Agreement and to make any decision pursuant to the authority prescribed in this Agreement so long as such action or decision was not performed or omitted with the intent to defraud or deliberately cause injury to the Limited Partners.

SECTION 15.2 NO LIMITED PARTNER PARTICIPATION. No Limited Partner shall participate in the management or operations of or have any control over the Partnership's business nor shall any Limited Partner have the power to represent, act for, sign for or bind the General Partner or the Partnership. The Limited Partners hereby consent to the exercise by the General Partner of the powers conferred on it by this Agreement.

SECTION 15.3 ACTIONS REQUIRING LIMITED PARTNER CONSENT. Notwithstanding any other provision of this Agreement to the contrary, without the consent of the Limited Partners, the General Partner shall not have the authority to:

(a) Admit a person as a General Partner, except as provided in this Agreement;

(b) Admit a person as a Limited Partner, except as provided in this Agreement;

(c) Knowingly perform any act that would subject the Limited Partners to any liability as a general partner; or

(d) Sell or dispose of all or substantially all of the assets of the Partnership.

SECTION 15.4 THIRD PARTY RELIANCE. Notwithstanding any other provision of this Agreement to the contrary, no lender, lessor, lessee, purchaser or any other person dealing with the Partnership, shall be required to verify any representation by the General Partner as to the extent of the interest in the assets of the Partnership that the General Partner is entitled to encumber, sell, lease or otherwise use or Partnership action that the General Partner is empowered to authorize and conduct and such person shall be entitled to rely exclusively on the representations of the General Partner as to its authority to enter into such arrangements and shall be entitled to deal with the General Partner as if it were the sole party in interest therein, both legally and beneficially. Each Limited Partner and each assignee or successor hereby waives any and all defenses or other remedies that may be available against such lender, purchaser or other person to contest, negate or disaffirm any such action of the General Partner. In no event shall any person dealing with the General Partner or its representative with respect to any business or property of the Partnership be obligated to

-19-

ascertain that the terms of this Agreement have been complied with, or to inquire into the necessity or expediency of any act of the General Partner or its representative; and every contract, agreement, deed, assignment, lease, mortgage, security agreement, promissory note or other instrument or document executed by the General Partner or its representative with respect to any business or property of the Partnership shall be conclusive evidence in favor of any and every person relying thereon or claiming thereunder that (i) at the time of the execution and delivery thereof this Agreement was in full force and effect, (ii) such instrument or document was duly executed in accordance with the terms and provisions of this Agreement and is binding upon the Partnership, and (iii) the General Partner or its representative was duly authorized and empowered to execute and deliver any and every such instrument or document for and on behalf of the Partnership.

SECTION 15.5 COMPENSATION AND REIMBURSEMENT. The General Partner shall receive no compensation for its services hereunder but shall be reimbursed for any costs, expenses, fees or other disbursements paid or incurred by it for or on behalf of the Partnership. As more fully provided in the Operating Procedures, the General Partner may pay with funds of the Partnership or incur for and on behalf of the Partnership, costs, expenses, fees and other disbursements as it in its sole discretion deems necessary for the ongoing operations of the Partnership's business, activities and affairs.

SECTION 15.6 EXCULPATION. Neither the General Partner, its Affiliates, nor any owner, member, manager, officer, director, partner, employee or agent of the General Partner or its Affiliates, shall be liable, responsible or accountable in damages or otherwise to the Partnership or any Partner for any action taken or failure to act (EVEN IF SUCH ACTION OR FAILURE TO ACT CONSTITUTED THE NEGLIGENCE OF SUCH PERSON) on behalf of the Partnership within the scope of the authority conferred on the person described in this Agreement and the LLC Agreement or by law unless such act or omission was performed or omitted fraudulently or constituted gross negligence or willful misconduct. To the extent that, at law or in equity, the General Partner, its Affiliates, or any owner, manager, officer, director, partner, employee or agent thereof have duties (including fiduciary duties) and liabilities relating to the Partnership or to another Partner, the General Partner, its Affiliates, or any owner, manager, officer, director, partner, employee or agent thereof acting under the Agreement shall not be liable to the Partnership or to any such other Partner for their reliance on the provisions of this Agreement. The provisions of this Agreement, to the extent that they expand or restrict the duties and liabilities of the General Partner, its Affiliates, or any owner, manager, officer, director, partner, employee or agent thereof otherwise existing at law or in equity, are agreed by the Partners to replace such other duties and liabilities of the General Partner, its Affiliates, or any owner, officer, director, partner, employee or agent thereof.

SECTION 15.7 INDEMNIFICATION.

(a) TO THE FULLEST EXTENT PERMITTED BY LAW, THE GENERAL PARTNER, ITS AFFILIATES, THEIR RESPECTIVE OFFICERS, MANAGERS, DIRECTORS, PARTNERS, MEMBERS, EMPLOYEES AND AGENTS OR ANY PERSON PERFORMING A SIMILAR FUNCTION (INDIVIDUALLY, AN "INDEMNITEE") SHALL BE INDEMNIFIED AND HELD HARMLESS BY THE PARTNERSHIP FROM AND AGAINST ANY AND ALL LOSSES, CLAIMS, DAMAGES, JUDGMENTS, LIABILITIES, OBLIGATIONS, PENALTIES, SETTLEMENTS AND REASONABLE EXPENSES (INCLUDING LEGAL FEES) ARISING FROM ANY AND ALL CLAIMS, DEMANDS, ACTIONS, SUITS OR PROCEEDINGS,

-20-

CIVIL, CRIMINAL, ADMINISTRATIVE OR INVESTIGATIVE, IN WHICH THE INDEMNITEE MAY BE INVOLVED, OR THREATENED TO BE INVOLVED, AS A PARTY OR OTHERWISE, BY REASON OF ITS STATUS AS (X) THE GENERAL PARTNER OR AN AFFILIATE THEREOF, OR (Y) AN OFFICER, MANAGER, DIRECTOR, PARTNER, EMPLOYEE OR AGENT OF THE GENERAL PARTNER OR AN AFFILIATE THEREOF, REGARDLESS OF WHETHER THE INDEMNITEE CONTINUES TO BE THE GENERAL PARTNER OR AN AFFILIATE OR AN OFFICER, MANAGER, DIRECTOR, EMPLOYEE OR AGENT OF THE GENERAL PARTNER OR AN AFFILIATE THEREOF AT SUCH LIABILITY OR EXPENSE IS PAID OR INCURRED, UNLESS THE ACT OR FAILURE TO ACT GIVING RISE TO INDEMNITY HEREUNDER CONSTITUTED FRAUD, GROSS NEGLIGENCE OR WILLFUL MISCONDUCT.

(b) The Partnership through the General Partner may purchase and maintain insurance on behalf of the General Partner and such other persons as the General Partner shall determine against any liability that may be asserted against or expense that may be incurred by such person in connection with the Partnership's activities, regardless of whether the Partnership would have the power to indemnify such person against such liability under the provisions of this Partnership Agreement.

(c) Expenses incurred by any Indemnitee in defending any claim with respect to which such Indemnitee may be entitled to indemnification by the Partnership hereunder (including without limitation reasonable attorneys' fees and disbursements) shall, to the maximum extent permitted by law, be advanced by the Partnership prior to the final disposition of such claim, upon receipt of a written undertaking by or on behalf of such Indemnitee to repay the advanced amount of such expenses unless it is determined ultimately that the Indemnitee is entitled to indemnification by the Partnership under Section 15.7(a).

(d) The indemnification provided in this Section 15.7 is for the benefit of the Indemnitees and shall not be deemed to create any right to indemnification for any other persons.

SECTION 15.8 OTHER ACTIVITIES; NONCOMPETITION.

(a) Each Limited Partner agrees that, for a period beginning on the date of this Agreement and ending on the earlier to occur of (i) the termination of this Agreement for whatever reason, or (ii) the disposition, directly or indirectly, by such Limited Partner, respectively (or, to the extent such interest has been transferred to an Affiliate of such Partner as permitted hereby, by such Affiliate), of its interest in the Partnership to a person that is not a wholly-owned, direct or indirect, subsidiary of such Limited Partner's Parent, no such Limited Partner or any of its Restricted Affiliates will engage or participate in, or carry on, directly or indirectly, either as proprietor, partner, member, director, stockholder, agent, consultant, advisor, trustee, Affiliate, or otherwise, whether or not for compensation, any business that competes with the Partnership Business in Chambers and Harris Counties, Texas (the "TERRITORY"). Each Limited Partner aknowledges and agrees that the foregoing noncompetition agreement is given in partial consideration for the other party's entering into this Agreement.

-21-

(b) Each of the Limited Partners or its Affiliates have, pursuant to Storage Agreements, leased or obtained the right to use storage capacity of the Partnership. Without limiting the generality of the provisions of Section 15.8(a) and in furtherance thereof, each of the Limited Partners agrees that it and its Affiliates will use such storage capacity solely for its own internal business operations and will not lease or sublease such storage capacity to third parties, and will direct to the Partnership any such third party seeking access to the Partnership's storage facility so that such additional business shall be for the benefit of the Partnership. The foregoing shall not be interpreted to prohibit or restrict the current practices of the TE Partner with respect to such storage capacity. Additionally, except as otherwise herein specifically provided, the Limited Partners agree to act in good faith with respect to each other and the Partnership Business to maximize the returns from such business throughout the term of the Partnership.

(c) In the event either Limited Partner (the "FIRST PARTY") or any of its Restricted Affiliates desires to pursue a business opportunity that will compete with the Partnership Business in the Territory as prohibited by Section 15.8(a) ("COMPETING BUSINESS TRANSACTION"), such Limited Partner shall be obligated to promptly bring such Competing Business Transaction to the attention of the other Limited Partner (the "SECOND PARTY"), to fully disclose all details regarding the proposed opportunity (subject to the execution of appropriate confidentiality undertakings), and to offer such business opportunity to become part of the Partnership Business. The Limited Partners shall negotiate in good faith for a period of 45 days (unless the business opportunity dictates a shorter period of time, in which event, for such shorter time period as will avoid the loss of such opportunity) to bring such Competing Business Transaction into the Partnership Business, either by contribution or purchase. Should the Limited Partners be unable to agree within such 45-day period on the terms and provisions for incorporating the Competing Business Transaction into the Partnership Business and the First Party or its Restricted Affiliate, as may be the case, wishes to pursue the Competing Business Transaction rather than continue with the Partnership Business and the transactions contemplated by this Agreement, subject to the provisions of Section 15.8(e) below, either Limited Partner may invoke the special buy-out provisions of Section 23.7 without first complying with Section 23.1. The First Party agrees to give written notice within one Business Day of the consummation of such transaction.

(d) In the event the First Party or any of its Restricted Affiliates desires to pursue a Competing Business Transaction which cannot become a part of the Partnership Business for any reason other than the failure of the Second Party to consent to such transaction, and the First Party or its Restricted Affiliate, as may be the case, elects to pursue such Competing Business Transaction rather than continue with the Partnership Business, the First Party shall give written notice of such election to the Second Party and either Party may invoke the special buy-out provisions of Section 23.7 without first complying with Section 23.1. The First Party agrees to give written notice within one Business Day of the consummation of such transaction.

-22-

(e) Notwithstanding any other provision of this Agreement, the Second Party may waive the non-competition provision of this Section 15.8 as it applies to any Competing Business Transaction offered by the First Party under Section 15.8(c). In such event, (i) the First Party may participate in such Competing Business Transaction without being in violation of this Agreement; (ii) the Second Party thereby waives any right to participate in, or to benefit from, such transaction; and (iii) neither Limited Partner shall be entitled to invoke the special buy-out provisions of Section 23.7 as a result of such transaction. Any such waiver shall apply only to the specific Competing Business Transaction being presented and shall not be construed as applicable to any other Competing Business Transaction.

(f) Notwithstanding the foregoing or any other provision of this $\ensuremath{\mathsf{Agreement}}$:

(i) Neither Limited Partner nor any of its Affiliates shall be prohibited from conducting business activities or operations outside the Territory of a similar nature as are conducted by such Limited Partner or Affiliate as of the date hereof; or

(ii) Neither Limited Partner nor any of its Affiliates shall be prohibited from constructing and operating other pipelines to and from or through the Territory so long as they do not compete with the pipelines that comprise a part of the Partnership's assets as of the date hereof,

(iii) Neither Limited Partner nor any of its Affiliates shall be prohibited from operating pipelines to, from or through the Territory which are currently operated by such persons, including the servicing of persons who are customers of the Partnership, or from expanding, improving or otherwise modifying such pipelines, or

(iv) Neither Limited Partner, nor any of its Affiliates, has any obligation or duty to first offer to the Partnership new business ideas that are outside the scope of the Partnership Business.

(g) Each Limited Partner acknowledges that the covenant provided in this Section 15.8 is manifestly reasonable on its face and is no more restrictive than is required for the protection of the Partnership, including the interest of the other Limited Partner and the Partnership. In the event that the provisions of this Section 15.8 should ever be deemed to exceed the time and geographic limitations permitted by applicable law, then such provisions shall be reformed to the maximum time or geographic limitations permitted by applicable law.

(h) It is specifically understood and agreed that any breach of the provisions of this Section 15.8 hereof is likely to result in irreparable harm to the Partnership and that an action at law for damages alone will be an inadequate remedy for such breach, and that the Partnership would suffer irreparable harm in the event either Limited Partner or its Restricted Affiliates fail to comply with its obligations hereunder. Therefore, in addition to any other remedy that may be available to it, the other Limited Partner or the Partnership shall be

-23-

entitled to enforce the specific performance of Section 15.8 on the breaching Limited Partner and its Restricted Affiliates and to seek both temporary and permanent injunctive relief (to the extent permitted by law) without the necessity of proving actual damages, and such other relief as the court may allow.

ARTICLE XVI

TRANSFER OF INTERESTS BY PARTNERS

SECTION 16.1 RESTRICTIONS ON TRANSFER. No Partner shall have the right to Dispose of its interest in the Partnership, or any portion thereof without (a) complying with the provisions of this Article XVI, (b) Disposing of its entire Partnership Interest to a single purchaser or transferee, and (c) simultaneously Disposing of its and any Affiliate's entire membership interest in the General Partner. Any attempted transfer or assignment of any interest in the Partnership in violation of the provisions of this Article XVI shall be void and of no force and effect.

SECTION 16.2 TRANSFERS TO AFFILIATES. A Limited Partner may Dispose of all of its Partnership Interest to an Affiliate of such Limited Partner provided that such purchaser or transferee, prior to such Disposition, becomes a party to this Agreement and agrees to be bound by all applicable terms and conditions including this Article XVI.

SECTION 16.3 RIGHT OF FIRST REFUSAL. If a Limited Partner ("OFFEROR") desires to Dispose of its Partnership Interest (other than a sale or other transfer of its Partnership Interest to an Affiliate or a Change of Partner Control), such Disposition may be made only by a sale of all of its entire Partnership Interest and its and any Affiliate's entire membership interest in the General Partner (collectively, the "INTEREST") for cash (or cash equivalent) and only if the Offeror receives with respect thereto a bona fide binding written proposal for the acquisition of such Interest (an "ACQUISITION PROPOSAL"), and then only in compliance with the following procedures:

> (a) Upon receipt of an Acquisition Proposal, the Offeror shall offer, by written notice (the "OFFER") to the other Limited Partner (the "OFFEREE"), to sell the Interest to the Offeree or Affiliates thereof on the terms (including price) specified in the Acquisition Proposal pursuant to the terms of this Section 16.3. Such Offer shall contain a description of and a copy of the Acquisition Proposal subject to appropriate confidentiality provisions. In addition, the Offeror shall provide to the Offeree all other information with respect to the Acquisition Proposal and the proposed transferee reasonably requested by the Offeree in order for it to evaluate the Acquisition Proposal, to verify the bona fide nature thereof and to evaluate the effect of having the person making such Acquisition Proposal as a Partner in the Partnership.

(b) The Offeree shall have the right, to be exercised by notice (the "ACCEPTANCE") from the Offeree to the Offeror on or before the 30th day following receipt of the Offer (the "OFFER PERIOD"), to elect to purchase all (but not less than all) of the Interest pursuant to the terms of the Offer.

-24-

(c) If the Offeree accepts the Offer, the closing of the acquisition of the Interest shall be consummated on or before the 60th day after the Offeror receives the Acceptance but effective at the end of the calendar month occurring on or immediately prior to such closing. The acquisition shall be consummated at a closing held at the principal offices of the Partnership (unless otherwise agreed by the purchaser of the Partnership Interest and the Offeror), at which time the purchaser shall deliver to the Offeror the purchase price (in the form of immediately available funds), and the Offeror shall deliver to the purchaser as shall be required to evidence the transfer of such Interest free and clear of all liens and encumbrances, except those created under this Agreement.

(d) If the Offeree does not accept the Offer, the Offeror shall be permitted for a period of 180 days after expiration of the Offer Period to sell all (but not less than all) of the Interest on terms not more favorable to such transferee than the terms specified in the Acquisition Proposal and at a price that is not less than the price specified in the Acquisition Proposal.

SECTION 16.4 CHANGE OF PARTNER CONTROL. If a Limited Partner or an Affiliate desires to Dispose of its Partnership Interest by a Change of Partner Control or undergoes an involuntary change of Partner Control, the other Limited Partner ("OPTION PARTNER") or its designated Affiliate shall have an option to acquire the Partnership Interest of the other Limited Partner and it or its Affiliate's membership interest in the General Partner (collectively, the "OPTION INTEREST") in accordance with the following procedures:

(a) If the Partner undergoing the Change of Partner Control is the TE Partner, the terms and provisions of Section 16.3 shall apply.

(b) If the Partner undergoing the Change of Partner Control is the LD Partner, the option price payable shall be the greater of (i) 50% of the Net EBITDA for the 12-month period preceding the Notice Month (as hereinafter defined) times ten (10.0) and (ii) the LD Partner's capital account in the Partnership and in the General Partner (or, if applicable, the capital account of its Affiliate in the General Partner), in both cases according to GAAP (collectively, the "LD COMBINED CAPITAL ACCOUNT BALANCE"). "Notice Month" means the month during which the LD Change of Partner Control occurs.

(c) The provisions of Sections 16.3(a), (b) and (c), with respect to notice, the offer period for the Option Interests, acceptance of the offer by the Option Partner and the closing of the acquisition shall apply.

SECTION 16.5 PROCEDURES. Any Partner who desires to transfer its Partnership Interest shall arrange for any permitted transferee to be bound by the provisions of this Agreement, as it may then be amended, by having such transferee execute two counterparts of an instrument of assignment satisfactory in form and substance to the General Partner and by delivering the same to the General Partner together with any such other information that may be required by counsel to the Partnership to determine whether the proposed transfer violates applicable federal or state securities or other laws

-25-

or regulations. If and when the consent of the other Partners to such assignment and the substitution of such transferee hereof is secured and the other requirements of this Article XVI are satisfied, the transferee shall become a substituted Partner as to the Partnership Interest thus transferred effective as of the first day of the calendar month during which the General Partner actually receives the aforesaid instrument of assignment executed by both the transferor and transferee. The transferee shall be required to pay any and all reasonable filing and recording fees, legal fees, accounting fees, and other charges and fees incurred by the Partnership and its counsel as a result of such transfer.

SECTION 16.6 PURCHASE AND SALE IN THE EVENT OF MAXIMUM LOSS BEING EXCEEDED.

(a) In the event that the Maximum Loss is exceeded and the provisions of Section 8.1(c)(ii) apply, then the TE Partner shall acquire the Partnership Interest of the LD Partner and the membership interest held by the LD Partner or its Affiliate in the General Partner for an amount equal to the LD Combined Capital Account Balance, less the product of (i) the sum of the LD Partner's Unrecovered Loss Balance and TE Partner's Unrecovered Loss Balance, times (ii) the LD Combined Capital Account Balance divided by the total of all GAAP capital account balances in the Partnership and the General Partner.

(b) The closing of the purchase and sale contemplated hereunder shall be consummated within 60 days after the provisions of Section 8.1(c)(ii) first become applicable at the principal office of the Partnership (unless otherwise mutually agreed by the Limited Partners), at which time the purchase price (in the form of immediately available funds) shall be delivered to the sellers and the sellers shall deliver to the buyer or its designee such transfer documentation reasonably acceptable to the buyer and the sellers as shall be required to evidence the transfer of such Partnership Interest and membership interest in the General Partner, free and clear of all liens and encumbrances, except those created under this Agreement.

SECTION 16.7 COMPLIANCE WITH SECURITIES LAWS. All Partners acknowledge that the Partnership interests have not been registered under (i) the Securities Act of 1933, as amended (the "1933 ACT"), in reliance on the exemptions afforded by Section 4(2) of the 1933 Act, or (ii) applicable state securities laws in reliance on exemptions under such laws. Therefore, to preserve said exemptions and notwithstanding anything contained herein to the contrary, the Partners hereby agree that interests of the Partners shall be nontransferable and nonassignable, except in compliance with the registration provisions of the 1933 Act and the Partnership Act, or an exemption or exemptions therefrom, and any attempted or purported transfer or assignment in violation of the foregoing shall be void and of no effect. Accordingly, as an additional condition precedent to any assignment or other transfer of any interest in the Partnership, the General Partner may require an opinion of counsel satisfactory to the General Partner that such assignment or transfer will be made in compliance with the registration provisions of the 1933 Act and applicable state securities laws or exemption(s) therefrom, and such transfer or assignor shall be responsible for paying said counsel's fee for the opinion.

-26-

ARTICLE XVII

DISSOLUTION OF THE PARTNERSHIP

The happening of any one of the following events shall work an immediate dissolution of the Partnership:

(a) An event of withdrawal of the General Partner as defined in Section 17-402 of the Partnership Act except that an event described in Subsections (a)(4) and (a)(5) of Section 17-402 shall not be an event of withdrawal unless within 90 days of such event of withdrawal, the remaining Partners agree in writing to continue the business of the Partnership and appoint, effective as of the date of withdrawal, a successor General Partner of the Partnership who acquires, or enters into an agreement to acquire, the Partnership Interest of the prior General Partner during such 90-day period;

(b) The receipt by the Partnership of the final payment due on the sales price of all or substantially all the assets of the Partnership or the Partnership's business following the Partnership's sale thereof;

- (c) The agreement by all of the Partners to dissolve;
- (d) At the time there are no Limited Partners; and

(e) Entry of a decree of judicial dissolution under Section 17-802 of the Partnership Act.

Any withdrawal of the General Partner under Section 17-602 of the Partnership Act shall be effective on the date specified in a written notice of withdrawal given by the General Partner to the Limited Partners; provided, however, that the effective date shall be not less than 60 days following the date of delivery of such notice.

ARTICLE XVIII

WINDING UP AND TERMINATION OF THE PARTNERSHIP

SECTION 18.1 LIQUIDATOR. If the Partnership is dissolved for any reason, a liquidator (the "LIQUIDATOR") shall commence to wind up the affairs of the Partnership and to liquidate and sell its assets. The General Partner shall serve as the Liquidator unless the dissolution occurred as a result of an event of withdrawal of the General Partner, in which case the person designated by the Limited Partners shall serve as the Liquidator. The Liquidator shall have full right and discretion to determine the time, manner and terms of sale or sales of Partnership property pursuant to such liquidation having due regard to the activity and condition of the relevant market and general financial and economic conditions. The Liquidator appointed in the manner provided herein shall have and may exercise, without further authorization or consent of any of the parties hereto or their legal representatives or successors in interest, all of the powers conferred upon the General Partner

-27-

under the terms of this Agreement (but subject to all of the applicable limitations, contractual and otherwise, upon the exercise of such powers) to the extent necessary or desirable in the good faith judgment of the Liquidator to carry out the duties and functions of the Liquidator hereunder for and during such period of time, not to exceed two (2) years after the date of dissolution of the Partnership, as shall be reasonably required in the good faith judgment of the Liquidator to complete the liquidation and dissolution of the Partnership as provided for herein, including, without limitation, the following specific powers:

> (a) The power to continue to manage and operate any business of the Partnership during the period of such liquidation or dissolution proceedings, excluding, however, the power to make and enter into contracts which may extend beyond the period of liquidation.

(b) The power to make sales and incident thereto to make deeds, bills of sale, assignments and transfers of assets and properties of the Partnership; provided, that the Liquidator may not impose personal liability upon any of the Partners under any such instrument.

(c) The power to borrow funds as may, in the good faith judgment of the Liquidator, be reasonably required to pay debts and obligations of the Partnership or operating expenses, and to execute and/or grant deeds of trust, mortgages, security agreements, pledges and collateral assignments upon and encumbering any of the Partnership properties as security for repayment of such loans or as security for payment of any other indebtedness of the Partnership; provided, that the Liquidator shall not have the power to create any personal obligation on any of the Partners to repay such loans or indebtedness other than out of available proceeds of foreclosure or sale of the properties or assets as to which a lien or liens are granted as security for payment thereof.

(d) The power to settle, release, compromise or adjust any claims asserted to be owing by or to the Partnership, and the right to file, prosecute or defend lawsuits and legal proceedings in connection with any such matters.

SECTION 18.2 RESERVES. After making payment or provision for payment of all debts and liabilities of the Partnership and all expenses of liquidation, the Liquidator may set up, for a period not to exceed the aforesaid two (2) years, such cash reserves as the Liquidator may deem reasonably necessary for any contingent liabilities or obligations of the Partnership. Upon the satisfaction or other discharge of such contingency, the amount of the reserves not retired, if any, will be distributed in accordance with this Article XVIII.

SECTION 18.3 SALE OF ASSETS; DISTRIBUTION OF PROCEEDS. Upon the winding up and termination of the business and affairs of the Partnership, its assets (other than cash) shall be sold, its liabilities and obligations to creditors and all expenses incurred in its liquidation shall be paid (either by payment or the making of reasonable provision for payment). All items of Partnership income, gain, loss or deduction shall be credited or charged to the Capital Accounts of the Partners pro rata in accordance with their Sharing Ratios. Thereafter, the net proceeds from such sales (after deducting all selling costs and expenses in connection therewith), together with (at the expiration

-28-

of the two (2) year period referred to therein) the balance and reserve account referred to in Section 18.2 above, shall be distributed among the Partners in accordance with their respective positive balances in their Capital Accounts.

SECTION 18.4 FINAL ACCOUNTING. Within a reasonable time following the completion of the liquidation of the Partnership's properties, the Liquidator shall supply to each of the Partners a statement prepared by the Partnership's accountants which shall set forth the assets and the liabilities of the Partnership as of the date of complete liquidation, each Partner's pro rata portion of distributions pursuant to Section 18.3, and the amount retained as reserves by the Liquidator pursuant to Section 18.2.

SECTION 18.5 RECOURSE LIMITED TO ASSETS OF THE PARTNERSHIP. Each holder of an interest in the Partnership shall look solely to the assets of the Partnership for all distributions with respect to the Partnership and its Capital Contribution thereto (including the return thereof) and share of profits or losses thereof, and shall have no recourse therefor (upon dissolution or otherwise) against the Partnership, the General Partner or the Liquidator. No holder of an interest in the Partnership shall have any right to demand or receive property other than cash upon dissolution and termination of the Partnership.

SECTION 18.6 TERMINATION. Upon the completion of the liquidation of the Partnership and the distribution of all Partnership funds, the Partnership shall terminate and the Liquidator shall (and is hereby given the power and authority to) execute, acknowledge, swear to and record all documents required to effectuate the dissolution and termination of the Partnership. No Partner shall be required to restore any deficit balance existing in its Capital Account upon the liquidation and termination of the Partnership.

ARTICLE XIX

NOTICES

To be effective, all notices and demands under this Agreement must be in writing and must be given (i) by depositing same in the United States mail, postage prepaid, certified or registered, return receipt requested, (ii) by telecopy with receipt confirmed by return telecopy or (iii) by delivering same in person and receiving a signed receipt therefor. For purposes of notice, the addresses of the Partners or their respective assigns shall be as set forth on the signature pages hereof. Notices mailed in accordance with the foregoing shall be deemed to have been given and made upon receipt. Any Partner or his assignee may designate a different address to which notices or demands shall thereafter be directed by written notice given in the manner hereinabove required and directed to the Partnership at its principal office as hereinabove set forth.

-29-

ARTICLE XX

AMENDMENT OF LIMITED PARTNERSHIP AGREEMENT

This Agreement may be modified or amended from time to time by the written agreement of the General Partner and each of the Limited Partners.

ARTICLE XXI

POWER OF ATTORNEY

Each Limited Partner and any permitted assignee or transferee of its interest hereunder, does hereby irrevocably constitute and appoint the General Partner, its true and lawful attorney in fact and agent, to execute, acknowledge, verify, swear to, deliver, record and file, in such Limited Partner's or assignee's name, place and stead, all instruments, documents and certificates which may from time to time be required to effectuate and implement the terms and provisions of this Agreement including, without limitation, the power and authority to execute, verify, swear to, acknowledge, deliver, record and file (i) all certificates and other instruments which the General Partner deems appropriate to form, qualify or continue the Partnership as a limited partnership in the State of Delaware and in such other states as the General Partner deems appropriate, (ii) all instruments which the General Partner deems appropriate to reflect any amendment to this Agreement, or modification of the Partnership, made in accordance with the terms of this Agreement, (iii) all conveyances and other instruments which the General Partner deems appropriate to reflect the dissolution and termination of the Partnership pursuant to the terms of this Agreement, including any writing required by the Partnership Act, (iv) all instruments relating to the admission of any additional or substituted Limited Partner, (v) a certificate of assumed name and such other certificates and instruments as may be necessary under any fictitious or assumed name statutes from time to time in effect in jurisdictions in which the Partnership conducts or plans to conduct business and (vi) all other instruments that may be required or permitted by law to be filed on behalf of the Partnership and that are not inconsistent with this Agreement. The power of attorney granted herein shall be deemed to be coupled with an interest, shall be irrevocable, shall survive and not be affected by the subsequent death, dissolution, bankruptcy, incompetency or legal disability of a Limited Partner and shall extend to such Limited Partner's successors and assigns; and may be exercised by executing such instrument acting as attorney in fact for the Limited Partner or in such other manner as said agent and attorney in fact may deem appropriate. Each Limited Partner hereby agrees to be bound by any action taken by the General Partner acting pursuant to such power of attorney, and each Limited Partner and assigns hereby waives any and all defenses which may be available to contest, negate or disaffirm any action of the General Partner taken under such power of attorney.

ARTICLE XXII

REPRESENTATIONS, WARRANTIES AND COVENANTS

The General Partner and each Limited Partner each hereby respectively represent and warrant to the others that (i) it is duly organized, validly existing and in good standing under the jurisdiction

-30-

of its organization, with full power and authority to enter into and perform its obligations under this Agreement; (ii) it has validly executed this Agreement, and upon delivery, this Agreement shall be a binding obligation of such party, enforceable against such party in accordance with its terms; and (iii) its entry into this Agreement and the performance of its obligations hereunder will not require the approval of any governmental body or regulatory authority and will not violate, conflict with or cause a default under, any of its organizational documents, any contractual covenant or restriction by which such party is bound, or any applicable law, regulation, rule, ordinance, order, judgment or decree.

ARTICLE XXIII

DISPUTES

SECTION 23.1 NEGOTIATION. Except for the injunctive remedies provided by Section 15.8, in the event of any claims, counterclaims, demands, causes of action, disputes, controversies, and other matters in question arising out of or relating to this Agreement, any provision hereof, the alleged breach thereof, or in any way relating to the subject matter of this Agreement or the relationship between the Partners created by this Agreement, involving the Limited Partners and/or their respective representatives and/or Affiliates (all of which are referred to herein as "DISPUTES"), the Limited Partners shall promptly seek to resolve any such Dispute by negotiations between senior executives of the Limited Partners who have authority to settle the Dispute. When a Limited Partner believes there is a Dispute under this Agreement, that Limited Partner will give the other written notice of the Dispute. Within thirty (30) days after receipt of such notice, the receiving Limited Partner shall submit to the other a written response. Both the notice and response shall include (i) a statement of each Limited Partner's position and a summary of the evidence and arguments supporting its position, and (ii) the name, title, fax number, and telephone number of the executive who will represent that Limited Partner. In the event the Dispute involves a claim arising out of the actions of any person or entity not a signatory to this Agreement, the receiving Limited Partner shall have such additional time as necessary, not to exceed an additional thirty (30) days, to investigate the Dispute before submitting a written response. The executives shall meet at a mutually acceptable time and place within fifteen (15) days after the date of the response and thereafter as often as they reasonably deem necessary to exchange relevant information and to attempt to resolve the Dispute. If one of the executives is an attorney or intends to be accompanied at a meeting by an attorney, the other executive shall be given at least five (5) days' notice of such intention and may also be accompanied by an attorney. All negotiations and communications pursuant to this Article XXIII shall be treated and maintained by the Limited Partners as confidential information and shall be treated as compromise and settlement negotiations for the purposes of federal and state rules of evidence.

SECTION 23.2 FAILURE TO RESOLVE. If the Dispute has not been resolved within sixty (60) days after the date of the response given pursuant to Section 23.1 above, or such additional time, if any, that the Limited Partners mutually agree to in writing, or if the Limited Partner receiving such notice denies the applicability of the provisions of Section 23.1 or otherwise refuses to participate under the provisions of Section 23.1, either Limited Partner may initiate arbitration pursuant to the

-31-

provisions of Section 23.3 below; provided, however, that either Partner may reject arbitration and trigger the special Buy-Out Option of Section 23.7, all as provided below.

SECTION 23.3 ARBITRATION. Any Disputes not settled pursuant to the foregoing provisions shall be resolved as follows:

(a) Any Limited Partner desiring to initiate arbitration in connection with any Dispute shall send, via certified mail, written notice of demand of arbitration to the other Limited Partner and the name of the arbitrator appointed by the Limited Partner demanding arbitration together with a statement of the matter in controversy.

(b) Within fifteen (15) days after receipt of such demand, the receiving Limited Partner shall either (i) agree to submit the Dispute to arbitration and name its arbitrator or (ii) refuse to submit the Dispute to arbitration, in which case the provisions of this Section 23.3 shall not be applicable to such Dispute. If the receiving Limited Partner fails or refuses to name its arbitrator within such 15-day period, it shall be deemed to have agreed to submit the Dispute to arbitration and the second arbitrator shall be appointed, upon request of the party demanding arbitration, by the Chief U.S. District Court Judge for the Southern District of Texas or such other person designated by such judge. The two arbitrators so selected shall within fifteen (15) days after their designation select a third arbitrator; provided, however, that if the two arbitrators are not able to agree on a third arbitrator within such 15-day period, either Limited Partner may request the Chief U.S. District Court Judge for the Southern District of Texas or such other person designated by such judge to select the third arbitrator as soon as possible. In the event the Judge declines to appoint an arbitrator, appointment shall be made, upon application of either Limited Partner, pursuant to the Commercial Arbitration Rules of the American Arbitration Association. If any arbitrator refuses or fails to fulfill his or her duties hereunder, such arbitrator shall be replaced by the Limited Partner which selected such arbitrator (or if such arbitrator was selected by another person, through the procedure which such arbitrator was selected) pursuant to the foregoing provisions.

(c) Each arbitrator selected by the Limited Partners shall be a certified public accountant or licensed attorney with at least fifteen (15) years of oil and gas experience as a certified public accountant and/or practicing attorney. The arbitrators selected by the Limited Partners are not required to be neutral, but the third arbitrator shall be neutral and shall be a retired judge.

(d) The Limited Partners hereto hereby request and consent to the three (3) arbitrators conducting a hearing in Houston, Texas no later than sixty (60) days following their selection or thirty (30) days after all prehearing discovery has been completed, whichever is later, at which the Limited Partners shall present such evidence and witnesses as they may choose, with or without counsel.

(e) Arbitration shall be conducted in accordance with the Commercial Arbitration Rules and procedures of the American Arbitration Association.

- 32 -

(f) The Federal Rules of Civil Procedure, as modified or supplemented by the local rules of civil procedure for the U.S. District Court for the Southern District of Texas, shall apply in the arbitration. The Limited Partners shall make their witnesses available in a timely manner for discovery pursuant to such rules. If a Limited Partner fails to comply with this discovery agreement within the time established by the arbitrators, after resolving any discovery disputes, the arbitrators may take such failure to comply into consideration in reaching their decision. All discovery disputes shall be resolved by the arbitrators pursuant to the procedures set forth in the Federal Rules of Civil Procedure.

(g) Adherence to formal rules of evidence shall not be required. The arbitrators shall consider any evidence and testimony that they determine to be relevant.

(h) The Limited Partners hereto hereby request that the arbitrators render their decision within thirty (30) calendar days following conclusion of the hearing.

(i) Any decision by a majority of the arbitration panel shall be final, binding and non-appealable. Any such decision may be filed in any court of competent jurisdiction and may be enforced by either Limited Partner as a final judgment in such court. There shall be no grounds for appeal of any arbitration award hereunder.

(j) The defenses of statute of limitations and laches shall be tolled from and after the date a Party gives the other Limited Partner written notice of a Dispute as provided in Section 23.1 above until such time as the Dispute has been resolved pursuant to Section 23.1, or an arbitration award has been entered pursuant to this Section 23.3.

(k) The Partners hereby waive any claim to, and the arbitrators are especially divested of any power to award, special, consequential, punitive, exemplary, incidental, indirect costs, expenses, damages or claims or statutory damages in excess of compensatory damages, or any form of damages in excess of compensatory damages.

SECTION 23.4 RECOVERY OF COSTS AND ATTORNEYS' FEES. In the event arbitration (or, despite the Limited Partners' agreement to resolve the Disputes through binding arbitration, litigation) arising out of this Agreement is initiated by either Limited Partner, the prevailing Limited Partner, after the entry of a final non-appealable order, shall be entitled to recover from the other party, as a part of said order, all court costs, fees and expenses of such arbitration (or litigation), including, without limitation, reasonable attorneys' fees.

SECTION 23.5 CHOICE OF FORUM. If, despite the Limited Partners' agreement to submit any Disputes to binding arbitration, there are any court proceedings arising out of or relating to this Agreement or the transactions contemplated hereby, such proceedings shall be brought and tried in the federal or state courts situated in the City of Houston, Texas.

SECTION 23.6 JURY WAIVERS. THE LIMITED PARTNERS HEREBY WAIVE ANY AND ALL RIGHTS TO DEMAND A TRIAL BY JURY.

-33-

(a) Upon the failure of the Limited Partners to resolve or submit a Dispute to binding arbitration within seventy-five (75) days after the date of the response given pursuant to Section 23.1, either Limited Partner shall have the right (the "PUT/CALL RIGHT") for a period of 30 days following the expiration of such 75-day period (the "EXERCISE PERIOD"), to be exercised by written notice to the other Limited Partner within such applicable period, to cause a purchase and sale to the TE Partner of the Interest of the LD Partner and its Affiliate. Such purchase and sale shall be for cash at a purchase price equal to the greater of (i) 50% of the Net EBITDA for the Pricing Period ending with the month immediately preceding the month in which the written notice is given exercising the Put/Call Right times the Valuation Multiple and (ii) the LD Combined Capital Account Balance.

(b) If either Limited Partner timely exercises such Put/Call Right, the Limited Partners shall consult for the purpose of determining the Valuation Multiple and, on or before the 10th day after exercise of the Put/Call Right, the non-exercising Limited Partner shall select one of four entities proposed by the exercising Limited Partner, each of which such entities shall have at least ten years experience in valuing similar assets or businesses (the "VALUATION EXPERT") if required as contemplated below. If on or before the 30th day after exercise of the Put/Call Right, such Limited Partners have not reached agreement on the Valuation Multiple, each Limited Partner shall submit a proposed Valuation Multiple to the Valuation Expert, together with any supporting documentation such Limited Partners deems appropriate, on or before the 60th day after the exercise of the Put/Call Right. The Valuation Expert shall determine the Valuation Multiple by selection of one of the proposed Valuation Multiples submitted by the Limited Partners (and shall have no authority beyond selection of one of such proposals) as promptly as possible (and in any event on or before the 30th day after submittal of the competing proposals). The cost of such Valuation Expert shall be paid in equal portions by the Limited Partners.

(c) The purchase price shall be the amount calculated under Section 23.7(a) unless the Put/Call Right has been triggered by either Partner pursuant to Section 15.8(c) or (d), in which case the purchase price to be paid by the TE Partner shall be the amount computed as follows:

(i) If the TE Partner is the First Party, the purchase price shall be 115% of the amount calculated under Section 23.7(a); and

(ii) if the LD Partner is the First Party, the purchase price shall be 85% of the amount calculated under Section 23.7(a).

(d) The closing of the acquisition of the Interests contemplated hereunder shall be consummated on or before the 60th day after the determination of the Valuation Multiple but effective at the end of the calendar month occurring on or immediately prior to such closing. The acquisition shall be consummated at a closing held at the principal offices of

-34-

the Partnership (unless otherwise mutually agreed by the Limited Partners), at which time the purchase price (in the form of immediately available funds) shall be delivered to the sellers and the sellers shall deliver to the buyer such transfer documentation reasonably acceptable to the buyer and the seller as shall be required to evidence the transfer of such Partnership Interest and membership interest in the General Partner, free and clear of all liens and encumbrances, except those created under this Agreement.

SECTION 23.8 DISPUTE UNDER BOTH THIS AGREEMENT AND LLC AGREEMENT. Notwithstanding anything herein to the contrary, it is the intention of the Partners that any Dispute hereunder that is also subject to the terms and provisions of the LLC Agreement shall be handled under one unified proceeding under and pursuant to the terms and provisions of this Agreement and that no Partner shall be entitled to seek a second resolution of the same Dispute under the Terms and Provisions of this Article XXIII.

ARTICLE XXIV

MISCELLANEOUS

SECTION 24.1 NO RIGHT OF PARTITION. The Partners agree that the Partnership properties are not and will not be suitable for partition. Accordingly, each of the Partners hereby irrevocably waives any and all rights that he may have to maintain any action for partition of any of the Partnership property.

SECTION 24.2 ENTIRE AGREEMENT; SUPERSEDURE. This Agreement and the additional documents and agreements referred to herein or contemplated hereunder constitute the entire agreement among the parties. It supersedes any prior agreement or understandings among them, and it may not be modified or amended in any manner other than as set forth herein.

SECTION 24.3 GOVERNING LAW. This Agreement and the rights of the parties hereunder shall be governed by and interpreted in accordance with the laws of the State of Delaware.

SECTION 24.4 BINDING EFFECT. Except as herein otherwise specifically provided, this Agreement shall be binding upon and inure to the benefit of the parties and their legal representatives, heirs, administrators, executors, successors and assigns.

SECTION 24.5 CONSTRUCTION OF AGREEMENT. Wherever from the context it appears appropriate, each term stated in either the singular or the plural shall include the singular and the plural, and pronouns stated in the masculine, the feminine or the neuter gender shall include the masculine, feminine and neuter. The term "person" means any individual, corporation, partnership, trust or other entity.

SECTION 24.6 CAPTIONS. Captions contained in this Agreement are inserted only as a matter of convenience and in no way define, limit or extend the scope or intent of this Agreement or any provision hereof.

-35-

SECTION 24.7 EFFECT OF INVALID PROVISION. If any provision of this Agreement, or the application of such provision to any person or circumstance, shall be held invalid, the remainder of this Agreement, or the application of such provision to persons or circumstances other than those to which it is held invalid, shall not be affected thereby.

SECTION 24.8 COUNTERPARTS. This Agreement may be executed in several counterparts, each of which shall be deemed an original but all of which shall constitute one and the same instrument. It shall not be necessary for all Partners to execute the same counterpart hereof.

SECTION 24.9 WAIVER. No failure by any party to insist upon the strict performance of any covenant, duty, agreement or condition of this Agreement or to exercise any right or remedy consequent upon a breach thereof shall constitute waiver of any such breach or any other covenant, duty, agreement or condition.

IN WITNESS WHEREOF, the parties hereto have executed this Agreement as of the day and in the year first above written.

Address:	GENERAL PARTNER:
2929 Allen Parkway, Suite 3200	Mont Belvieu Venture, LLC
Houston, TX 77019	Dere
Telecopy: 713/759-3645	Ву:
	Its:
	LIMITED PARTNERS:
2929 Allen Parkway, Suite 3200 Houston, TX 77019 Telecopy: 713/759-3645	TE Products Pipeline Company, Limited Partnership, by TEPPCO GP, Inc., its general partner
	By:

Its:

-36-

Louis Dreyfus Energy Services L.P.

10 Westport Road Wilton, CT 06897 Telecopy: 203/761-8119

By: Its:

-37-

EXHIBIT 12.1

•

STATEMENT OF COMPUTATION OF RATIO OF EARNINGS TO FIXED CHARGES

YEARS ENDED DECEMBER 31, THREE MONTHS --------------------------------ENDED 1998 1999 2000 2001 2002 MAR. 31, 2003 ---------- ------ ------------------ (in thousands) EARNINGS Income From Continuing Operations * 53,885 72,856 65,951 92,533 105,882 30,465 Fixed Charges 30,915 34,305 55,621 72,217 77,726 23,207 Distributed Income of Equity Investment -- -- --40,800 30,938 --Capitalized Interest (795) (2, 133)(4, 559)(4,000) (4, 345)(596) --------------- ------ ------ - - - - - - - -Total Earnings 84,005 105,028 117,013 201,550 210,201 53,076 ======= ======= ======= ======= ======= ======= FIXED CHARGES Interest

Expense 29,784 31,563 48,982 66,057 70,537 21,905 Capitalized Interest 795 2,133 4,559 4,000 4,345 596 Rental Interest Factor 336 609 2,080 2,160 2,844 706 ---------- ------ -------------- Total Fixed Charges 30,915 34,305 55,621 72,217 77,726 23,207 _____ ======= ======= ======= ======= ======= RATI0: EARNINGS / FIXED CHARGES 2.72 3.06 2.10 2.79 2.70 2.29 ======= ======= _____ ======= ======= =======

* Excludes minority interest, extraordinary loss and undistributed equity earnings.

CERTIFICATION OF CHIEF EXECUTIVE OFFICER PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

The undersigned, being the Chief Executive Officer of Texas Eastern Products Pipeline Company, LLC, the sole general partner of TEPPCO Partners, L.P. (the "Company"), hereby certifies that, to his knowledge, the Company's Quarterly Report on Form 10-Q for the quarterly period ended March 31, 2003, filed with the United States Securities and Exchange Commission pursuant to Section 13(a) or 15(d) of the Securities Exchange Act of 1934 (15 U.S.C. 78m or 78o(d)), fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934 and that information contained in such Quarterly Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

A signed original of this written statement required by Section 906 has been provided to TEPPCO Partners, L.P. and will be retained by TEPPCO Partners, L.P. and furnished to the Securities and Exchange Commission or its staff upon request.

May 1, 2003 ------Date

CERTIFICATION OF CHIEF FINANCIAL OFFICER PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

The undersigned, being the Chief Financial Officer of Texas Eastern Products Pipeline Company, LLC, the sole general partner of TEPPCO Partners, L.P. (the "Company"), hereby certifies that, to his knowledge, the Company's Quarterly Report on Form 10-Q for the quarterly period ended March 31, 2003, filed with the United States Securities and Exchange Commission pursuant to Section 13(a) or 15(d) of the Securities Exchange Act of 1934 (15 U.S.C. 78m or 78o(d)), fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934 and that information contained in such Quarterly Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

A signed original of this written statement required by Section 906 has been provided to TEPPCO Partners, L.P. and will be retained by TEPPCO Partners, L.P. and furnished to the Securities and Exchange Commission or its staff upon request.

May 1, 2003 ------Date

/s/CHARLES H. LEONARD

Charles H. Leonard Senior Vice President and Chief Financial Officer Texas Eastern Products Pipeline Company, LLC, General Partner